#### **ABSTRACT**

### MALE HEALTH PROMOTION PROFESSIONALS IN HIGHER EDUCATION: WHY THEY ENTER AND REMAIN

Alicia Czachowski, Ed.D.

Department of Counseling, Adult, & Higher Education
Northern Illinois University, 2015

D. Eric Archer, Director

This dissertation examines men in the career of health promotion in higher education.

This study was designed to gain insight from men currently working in health promotion in higher education about why and how they entered their career as well as what is keeping them working in this field. The purpose of this qualitative study was to explore the perceptions and experiences of men working as health promotion professionals in the higher education setting regarding their entry into and persistence in their chosen profession.

Using a basic interpretive qualitative study approach, five participants took part in three semi-structured interviews, in which they were asked questions regarding their motivations for entry into health promotion in higher education and what influences them to remain in the career. Findings indicate that there were many factors impacting career entry and career persistence for the participants. While the findings of this study may be transferable to other men in health promotion in higher education, they are only generalizable to the five participants. The findings add to the current literature on professionals working in health promotion in higher education and student affairs, and to the literature on men in historically nontraditional career.





# NORTHERN ILLINOIS UNIVERSITY DEKALB, ILLINOIS

#### **AUGUST 2015**

# MALE HEALTH PROMOTION PROFESSIONALS IN HIGHER EDUCATION: WHY THEY ENTER AND REMAIN

BY

# ALICIA KATHERINE CZACHOWSKI ©2015 Alicia Katherine Czachowski

# A DISSERTATION SUBMITTED TO THE GRADUATE SCHOOL IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE DOCTOR OF EDUCATION

DEPARTMENT OF COUNSELING, ADULT, & HIGHER EDUCATION

Doctoral Director:
D. Eric Archer



ProQuest Number: 3723848

#### All rights reserved

#### INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



#### ProQuest 3723848

Published by ProQuest LLC (2015). Copyright of the Dissertation is held by the Author.

All rights reserved.

This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 - 1346



#### **ACKNOWLEDGMENTS**

First, thank you to my committee Chair, D. Eric Archer, PhD. You took me on without really knowing me and helped me get through this whole crazy process. I know that at times I was nervous about timelines so I thank you for seeing me through. I also thank my committee members. James Ciesla, while we have still never met in person, I truly valued your feedback, comments, and flexibility with doing so much of this work as a committee member electronically and via video chats. Finally, Michael McNeil, you have managed to have a hand in all three of my degrees, and I could not have asked for a better teacher, mentor, or friend. Thank you for your feedback and support throughout my professional and academic journey.

I also must thank the five men who gave their time during a busy fall semester to spend countless hours with me on the phone sharing your stories, insights, and experiences. This study could not have occurred without you. I truly appreciate the trust you placed in me and the time you gave to me. I hope that my findings will benefit you and the men to come in the future! Stacy Andes, thank you for agreeing to be my peer debriefer and for giving me great ideas and feedback to make this study better. To all of my friends and colleagues, thank you for your support, words of encouragement, notes, and hugs at meetings and conferences. It is amazing to have so many supportive people in my life. To those just starting this process and those still working on your book reports, you can do it!



Finally, to my family, specifically Mom, Dad, and Jessica, your love and support through this whole process made it accomplishable. You listened to laughter, tears, my frustration, and my triumphs. Thank you for all that you have done for me! I love you all.



#### TABLE OF CONTENTS

	Page
LIST OF TABLES.	xi
LIST OF APPENDICES.	xii
Chapter	
1. INTRODUCTION.	1
Problem Statement.	. 4
Purpose	. 7
Research Questions	. 8
Significance of the Study	8
Nature of the Study	. 10
The Researcher	. 11
Overview of the Study	. 13
2. LITERATURE REVIEW	. 14
Organization of This Review	. 16
Men in Nontraditional Careers	. 16
Job Requirements for Men in Nontraditional Careers	. 17
Entering Nontraditional Careers.	. 19
Education	19
Influence From Others	20

Chapter		Page
	Previous Interest in the Career.	. 22
	Personality	. 23
	Perception of the Career	26
	Theories of Career Entry	. 27
	Preparation for and Persistence in Nontraditional Careers	28
Possib	ole Issues and Challenges	. 30
	Pressure to Advance	. 31
	Male Privilege.	. 32
	Gender Discrimination	. 33
	Salary	. 34
	Role Strain.	35
	Stereotypes Regarding Sexual Identity	36
	Perceptions of Men.	. 37
Caree	rs in Higher Education, Student Affairs, and Health Promotion	. 38
Studen	nt Affairs	. 39
	Motivation for Entry	40
	Career Mobility & Persistence	. 42
	Relationships	. 43
	Professional Involvement.	45
	Career Satisfaction	46
	Gender and Student Affairs	46



C	Chapter	Page
	Health Promotion.	. 48
	Men in Health Promotion.	53
	Conclusion	. 55
3.	METHODS	57
	Purpose.	58
	Research Questions.	58
	Methodology	. 58
	Research Procedures	. 60
	Research Participants	. 60
	Data Collection.	61
	Data Analysis	. 65
	Conceptual Framework	. 66
	Trustworthiness	. 68
	Ethical Considerations.	. 69
	Summary and Conclusions	70
4.	FINDINGS	71
	Participants	71
	David	72
	Harry	. 73
	James	75
	Mark	76



Chapter		Page
	Matt	. 77
Facto	ors of Entry Into Health Promotion in Higher Education	. 78
	Attraction to or Interest in Teaching, Helping Others, or Health	. 78
	One of Very Few Men.	80
	Female-Dominated	. 81
	A Lot of Women in Graduate School	81
	It is More Than Just a Public Health or Student Affairs Degree	. 82
	What is College Health?	82
	Coursework Did Not Prepare Me for This	83
	Influence From Others	. 84
	Experiences Growing Up.	. 85
	Experience as a Student Leader	. 86
Facto	ors of Persistence in Health Promotion in Higher Education	. 88
	Perceived Benefits	. 89
	Fulfilling and Enjoyable	89
	Mentorship	. 90
	Students and Newer Professionals	. 90
	Participants' Mentors	. 91
	Professional Relationships.	92
	Support From Friends and Family	93
	The Privilege That Comes From Being a Man	94



Chapter		Page
Perce	ived Challenges	95
	Feeling Unsupported.	95
	Salary	97
	What Do You Actually Do?	98
	Career Aspirations.	99
	Moving Up Means More Administration	. 100
٠	Balancing Being a Man and a Lack of Men in the Profession	. 101
	Gender and the Work	. 102
	A "Different Kind of Man"	. 103
	More Men	. 104
	Sexual Identity	. 106
Gend	er Not Considered	. 107
Summary		. 107
5. DISCUSSION		. 108
Discussion o	f Findings	. 108
Career Entry	······	. 109
Attrac	ction to or Interest in Teaching, Helping Others, or Health	109
One o	of Very Few Men	110
	A Lot of Women in Graduate School.	. 110
	Female-Dominated.	111
It is M	More Than Just a Public Health or Student Affairs Degree	112



Chapter		Page
	What is College Health?	112
	Coursework Did Not Prepare Me for This	113
I	Influence From Others	115
I	Experiences Growing Up	116
I	Experience as a Student Leader.	116
Persister	nce in Health Promotion in Higher Education	119
I	Beneficial Factors.	119
	Fulfilling and Enjoyable.	119
	Mentorship	121
	Students and Newer Professionals	121
	Participants' Mentors.	122
	Professional Relationships.	124
	Support From Friends and Family	125
	The Privilege That Comes From Being a Man	126
(	Challenging Factors.	128
	Feeling Unsupported.	129
	Salary	130
	What Do You Actually Do?	132
	Career Aspirations	133
	Moving Up Means More Administration.	134
	Balancing Being a Man and a Lack of Men in the Profession	135



	Chapter	age
	Gender and the Work	36
	A "Different Kind of Man"	37
	More Men	37
	Sexual Identity	39
	Gender Not Considered.	41
	Implications for Practice.	42
	Implications for the Literature	46
	Recommendations for Future Research.	49
	Limitations. 1:	51
	Conclusion	53
R	REFERENCES15	54
Δ	A PPENDICES 16	61

#### LIST OF TABLES

Table		Page
1.	Participant Demographics.	. 72
2.	Factors of Career Entry.	. 79
3.	Factors of Persistence.	. 88

#### LIST OF APPENDICES

Appendix	
A. RECRUITMENT EMAIL	162
B. INFORMED CONSENT FORM	164
C. INTERVIEW 1 GUIDE	167
D. INTERVIEW 2 GUIDE	170
E. INTERVIEW 3 GUIDE	172

•

#### CHAPTER 1

#### INTRODUCTION

The majority of the research on gender and career choice focuses on women who choose to enter traditionally male careers including careers in technology and the sciences (Williams, 1992). In addition, there is a growing body of literature on men who choose to enter traditionally female careers (Dodson & Borders, 2006; Jackson, Wright, & Perrone-McGovern, 2010). A nontraditional career is one predominately pursued by one gender or the other (Williams, 1993). Some examples of nontraditional careers for men explored in the literature include elementary education, nursing, and social work (Sargent, 2005; Williams, 1995). In this study, I focused on men working in the predominately female field of health promotion in higher education. A health promotion professional, "works to prevent the development of personal and campus population-level health problems, while enhancing individual, group, and institutional health and safety" (American College Health Association, 2012b, p. 2). Health promotion at institutions of higher education is a nontraditional career choice for men because approximately 80% of health promotion professionals working in higher education are female (National Association of Student Personnel Administrators, 2011; A. Ticker, personal communication, November 5, 2012). Both prior qualitative and quantitative studies explore why men enter nontraditional careers, why they remain in their chosen career, and the benefits and challenges they experience when working in these careers; however, the studies are not about



health promotion in higher education (Williams, 1993). In this study, I utilized a qualitative approach to explore why men enter and remain in the field of health promotion in higher education.

Institutions of higher education (IHEs) vary across the country and include private institutions, public institutions, research institutions, and technical colleges among other types of classifications (Berdahl, Altbach, & Gumport, 1999; Merriam & Brockett, 2007). IHEs also include community colleges offering 2 years of college-level education, universities offering undergraduate bachelor's and graduate degrees, and technical schools offering training programs that prepare students for specific careers (National Center for Education Statistics, 2012a).

According to the 1998 Amendments to the Higher Education Act of 1965, an institution of higher education is a state-authorized educational institution that admits students who can show they graduated from high school or hold an equivalent certification such as from passing the General Educational Development (GED) test. For the purposes of this study, IHEs are public and private universities in the United States offering, at minimum, 4-year bachelor's degrees. The institutions may also offer graduate degrees. I did not include technical schools and 2-year community colleges.

For the purposes of this study, the definition I used for health promotion in higher education follows the principles set forth by the American College Health Association's (ACHA) *Standards of Practice for Health Promotion in Higher Education* (2012b): "Specific health promotion initiatives aim to expand protective factors and campus strengths, and reduce personal, campus, and community health risk factors" (American College Health Association, 2012b). The purpose of health promotion in higher education is to support the success of



students by working to create a positive and healthy environment for them to learn and live in including the physical and social aspects of the environment (World Health Organization, 1991).

According to Zimmer (2002), approximately 80% of member institutions of the ACHA stated that they had health education or promotion services on their campuses. Divisions of student affairs house the majority of health promotion departments at most colleges and universities (National Association of Student Personnel Administrators, 2011). Student affairs departments at institutions of higher education provide educational opportunities outside of the classroom for students which focus on student learning and personal development (American College Personnel Association, 1996). The ACHA Health Promotion Needs Assessment 2012 (American College Health Association, 2012a) reports that 89% of participants stated that their department was located within student affairs.

Health promotion professionals working in the setting of higher education focus on college and university students, faculty, and staff. The ACHA's *Standards of Practice for Health Promotion in Higher Education* (American College Health Association, 2012b) states that the purpose of health promotion in higher education is to support student success by practicing prevention and increasing those programs and initiatives that support a healthy environment while decreasing barriers to a healthy campus. IHEs include most health promotion departments within student affairs; therefore, many health promotion professionals may consider themselves student affairs professionals. "Student affairs professionals are educators who share responsibility with faculty, academic administrators, other staff, and students themselves for creating the conditions under which students are likely to expend time and



energy in educationally-purposeful activities" (American College Personnel Association, 1996, p. 9).

According to the ACHA Health Promotion Section Needs Assessment 2012 (American College Health Association, 2012a), the majority of health promotion departments provide services for undergraduate and graduate students. Based on the ACHA National College Health Assessment (American College Health Association, 2014b), topic areas discussed by health promotion departments include, but are not limited to, alcohol and other drugs, stress management, sleep hygiene, interpersonal violence, sexual health, general health education, exercise, and nutrition. Approximately half of health promotion professionals consider themselves generalists rather than specialists in one health topic area (American College Health Association, 2012b). According to the ACHA's Guidelines for Hiring Health Promotion Professionals in Higher Education (American College Health Association, 2014a), "regardless of position level, all qualified health promotion professionals should be competent in conducting needs assessments, selecting priority health issues and measurable behavioral objectives, designing and implementing evidence- and theory-based initiatives, evaluating effectiveness, and reporting outcomes" (p. 3).

#### **Problem Statement**

Although there is literature available on men working in nontraditional careers in general, I found no literature focusing on men working in health promotion in higher education. Several researchers (Calhoun & Taub, 2014; Jackson et al., 2010; Lupton, 2006; Simpson, 2005) highlight the need for more research on the area of men in nontraditional careers, though they do not specifically mention the field of health promotion in higher education. There is also



literature suggesting that it is important to have men involved in health promotion within the college environment (Beshers, 2008; Cupples, Zukoski, & Dierwechter, 2010; Harris & Edwards, 2010). Cupples et al. (2010) suggest that men may connect well with other men on certain health issues. The study focused on male peer health educators, young men trained to educate their peers; however, the findings may be transferrable to men working in health promotion in higher education. Though not specifically related to health promotion, two studies that focused on the experiences of men in higher education found that older male role models may positively influence college-age men (Harris & Edwards, 2010). Finally, researchers who focused on the experiences of entry-level men in student affairs state that there is a need to recruit and retain men in student affairs (Calhoun & Taub, 2014). Based on the findings of these previous studies, a male health promotion professional could have a positive influence on men at institutions of higher education. Therefore, understanding why men enter and remain in this career is important.

This study focused on men who work in health promotion in higher education.

According to the National Center for Education Statistics (2012b), in the 2009-2010 academic year, of those institutions offering degrees in health professions and related programs, women earned 85% of undergraduate degrees and 81% of master's degrees. Degrees in "health professions and related programs" include, but are not limited to,

Health Services/Allied Health/Health Sciences, General; Dentistry; Health and Medical Administrative Services; Allied Health Diagnostic, Intervention, and Treatment Professions; Health/Medical Preparatory Programs; Medicine; Mental and Social Health Services and Allied Professions; Pharmacy, Pharmaceutical Sciences, and Administration; Public Health; Rehabilitation and Therapeutic Professions; Dietetics and Clinical Nutrition Services; and Registered Nursing, Nursing Administration, Nursing Research and Clinical Nursing. (National Center for Education Statistics, 2012b, p. 1)



Focusing specifically on health promotion in higher education, the *ACHA Health Promotion*Needs Assessment 2012 (American College Health Association, 2012a) reports that 72.4% of respondents had a degree in health-related fields including Public Health, Health

Education/Health Promotion, Community Health, or Health Care Administration (p. 24).

A. Ticker, the membership coordinator of the ACHA, stated that in 2011-2012, 16.2% of the 481 members of the ACHA Health Promotion Section were male (personal communication, November 5, 2012). Finally, the *Wellness and Health Promotion Benchmark Survey 2011* (National Association of Student Personnel Administrators, 2011) completed by the National Association of Student Personnel Administrators (NASPA) found that 23.5% of directors of health promotion departments stated they were male (p. 7). While these are two separate studies, the findings indicate few men enter health promotion in higher education; however more may advance into a director role. The findings are similar to information about student affairs in higher education where more women enter the profession, but more men advance (Hamrick & Carlisle, 1990; Engstrom, McIntosh, Ridzi, & Kruger, 2006).

The ACHA's Guidelines for Hiring Health Promotion Professionals in Higher Education (American College Health Association, 2014a) list only two levels of staffing for health promotion departments: the staff-level and the director-level health promotion professional. Therefore, there is little room for advancement in the field. Williams (1992) indicates men may be encouraged to enter into higher-level administrative positions, and the lack of opportunity for advancement could be a reason why some men do not remain in the field.

There is no standard definition of what makes a career traditional or nontraditional for a person to enter based on gender. The World Health Organization (2013) defines gender as a



socially constructed role of what society considers appropriate for a man or woman. Therefore, in this research study, when I use the term gender, it indicates the "roles, behaviors, activities, and attributes" (World Health Organization, 2013) society deems appropriate in the United States for men and women rather than the biological sex of a person. The participants in this study did not need to be biologically male, but must have self-identified as male. Kanter (1977) states that in order for a career to be considered "gendered," 85% of those people working in the field need to be of one gender. Based on the statistics discussed above, health promotion in higher education may fall slightly outside of these numbers. However, Williams (1992) states that nontraditional careers for men, including social workers and librarians, also fall outside of that 85%. The statistics confirm practitioners in health promotion in higher education are predominately female. Therefore, for the purposes of this study, a career in health promotion is nontraditional for men.

There is a need for more information on why men choose to enter health promotion in higher education and the influence their presence has on practice in the field. As prior research suggests, it may be helpful for colleges and universities to have male health promotion professionals on staff to assist with connecting male students on campus with health resources (Beshers, 2008; Cupples et al., 2010; Harris & Edwards, 2010).

#### Purpose

The purpose of this basic interpretive qualitative study was to explore the perceptions of men working as health promotion professionals in the higher education setting regarding their entry into and persistence in their chosen profession.



#### **Research Questions**

In order to conduct this research and address the stated purpose above, the following research questions guided this study:

- 1. What influenced participants in their decision to enter health promotion in the setting of higher education?
- 2. What influences participants to remain in health promotion in the setting of higher education?
  - 2a. What benefits do men perceive working in health promotion in higher education?
  - 2b. What challenges do men perceive working in health promotion in higher education?

#### Significance of the Study

In this study, I aimed to gain a better understanding about why men enter and remain in health promotion in higher education. Previous research indicates men connect well with other men when discussing health issues and older adult males positively influence college-age men (Cupples et al., 2010; Harris & Edwards, 2010). Male health promotion professionals may connect with college-age men and influence them to make positive health decisions. Therefore, having men in a role to discuss health issues with college-age men may be important for health promotion departments and their institutions. The current study sheds light on why some men choose to become health promotion professionals in higher education and why they remain in

their roles. The findings of this study helps those working in the profession to better understand the benefits, barriers, and challenges of men who choose to work in this field.

Along with acquiring a better understanding of men in health promotion in higher education, this study also adds to the body of research on men working in nontraditional careers. The majority of studies I reviewed on men in nontraditional careers focused on three careers: primary/elementary education (Galbraith, 1992; Sargent, 2004; Sargent, 2005; Simpson, 2005; Walsh, 1988; Williams, 1992), social work (Gillingham, 2006; Williams, 1992; Williams, 1995), and nursing (Galbraith, 1992; Simpson, 2005; Walsh, 1988; Williams, 1992; Williams, 1995). I also found one study about entry-level men working within different departments in student affairs, though health promotion was not represented by the participants (Calhoun & Taub, 2014). Jackson et al. (2010) states that it is important to study men in other nontraditional careers to gain an understanding of the impact their experiences have on their life, their career, and on society. Therefore, if health promotion departments want to recruit and retain male health promotion professionals, it is important for them to understand the professionals' experiences and perceptions.

There is a lack of empirical research available on health promotion professionals who choose to work in higher education. The American College Health Association has published the *Standards of Practice for Health Promotion in Higher Education* (2012b), the *Guidelines for Hiring Health Promotion Professionals in Higher Education* (2014), and the *ACHA Health Promotion Needs Assessment 2012* (2012b). These publications include knowledge and skills professionals working in health promotion in higher education should have, and include guidelines for those hiring qualified health promotion professionals. Additionally, the NASPA completed a study focused on directors of health promotion departments in higher education

(National Association of Student Personnel Administrators, 2011). I was able to find one study focusing on the perceived professional development needs of health promotion professionals in higher education (Davidson, 2008). I was not able to locate any other published research focusing specifically on why people enter a career as health promotion professionals in higher education. Health promotion departments and institutions of higher education could utilize the findings of this study to better understand one subset of the professionals working in health promotion.

This study adds to the literature about both men in nontraditional careers and professionals working in health promotion in higher education and student affairs; members of the research community who are interested in men working in nontraditional careers, and professionals working in health promotion in higher education and in student affairs could find the results of this study beneficial in their practice. Health promotion and student affairs professionals may utilize the findings to better understand why men enter and remain in this field and to help create environments that encourage men to remain in the field.

#### Nature of the Study

I utilized a basic interpretive qualitative design for this study (Merriam, 2009). I conducted interviews with men who were currently working in health promotion in higher education. I chose this methodology because the purpose of the research was to better understand the lives, perceptions, and experiences of the participants (Merriam, 2009). I drew on the literature available on career choice and career persistence of men in nontraditional careers and professionals working in higher education when conducting my analysis of the



findings. However, because I focused on this previously unstudied population, I explored factors of career entry and persistence not previously discussed in the literature.

The research participants for this study were five men who worked as health promotion professionals at 4-year, public or private IHEs in the United States. All participants had at least 10 years of experience working in health promotion in higher education, though, the minimum required for participation was 5 years. As I discussed previously, there were a small number of men employed in this field; therefore, I did not want to limit the pool of candidates for this study by requiring more than 5 years of experience in the field. Men no longer working in higher education, retirees, or those who had taken on different roles in higher education were not included in this study because they may not have been able to explain why they remained in the profession. I utilized purposeful sampling in order to recruit and select participants (Merriam, 2009). I collected data using a semistructured interview approach and analyzed the data using the constant comparative method (Merriam, 2009). Member checks and review by a peer debriefer aided in obtaining trustworthy data (Merriam, 2009).

#### The Researcher

When asked to describe my research interest, many people ask why I, a woman, focused my research on men working in a nontraditional career. The following section offers an explanation of why I chose this research area and why I considered it important.

I began working in health promotion in higher education when I was a freshman at the State University of New York at Oswego. I started as a peer health educator while going to a school that taught criminal justice, studying to be a police officer. My experiences working as a peer educator influenced me to change my career path to pursue a master's degree in public



health in order to gain employment as a health promotion professional. During the time of this study, I was in my 8th year of working as a health promotion professional in higher education. My experiences made me interested in why others decide to enter this profession, as there are currently no degree programs for health promotion in higher education (McNeil, 2012).

I heard conversations regarding men working in health promotion in higher education since I began working in this field. Why were there not more male peer educators? How can we recruit more? How can a female health promotion professional connect on health issues better with fraternity men or male athletes? How can a male student feel comfortable talking about personal matters with a female health educator? Health promotion professionals work to create an environment in which students are comfortable asking questions and obtaining information on many health topics including relationship issues and sexual health.

I have experience working at five institutions of higher education and with three male health promotion professionals, one of whom retired and a second who left the profession.

Additionally, I volunteer on regional and national committees for health promotion in higher education and peer health education. Through my experiences, I have observed a limited number of men in this profession and I know little about this subset of professionals working in health promotion in higher education.

For the above reasons, I chose to study men in health promotion in higher education.

This study did not answer all of my questions, nor is it generalizable to all men working in this profession; however, the findings may help the field's professionals gain a better understanding of why and how men choose to enter this profession and why they choose to remain.



#### Overview of the Study

In Chapter 1, I introduced the key points of the current study including the purpose, significance, the need for this research, and the nature of the study. Chapter 2 provides a review of the related literature. Chapter 3 provides a more detailed explanation of the qualitative research methodology guiding this study. Chapter 4 provides an explanation of the findings, and chapter 5 provides a discussion of findings, final conclusions, and suggestions for further research and for practice.



#### **CHAPTER 2**

#### LITERATURE REVIEW

There are assumptions regarding gender and biological sex that play a role in job creation, the writing of job descriptions, the hierarchy of organizations, and the culture of the workplace (Williams, 1995). Conducting research on men and women entering nontraditional careers is important because it sheds light on the barriers to these careers and how to reduce them (Lupton, 2006). The majority of research studies and literature regarding people entering nontraditional careers focuses on women and their struggle to gain entrance into careers traditionally held by men (Jackson et al., 2010; Williams, 1992). However, it is important to better understand how and why men enter nontraditional careers because more men are entering them (Jackson et al., 2010). Therefore, in this chapter, I discuss the literature on men entering nontraditional careers and professionals entering careers in student affairs. Additionally, I discuss the literature on career persistence for men in nontraditional career and student affairs.

There are several careers dominated by women discussed in the literature including elementary teachers, nurses, and social workers (Galbraith, 1992; Gillingham, 2006; Sargent, 2004; Sargent, 2005; Simpson, 2005; Walsh, 1988; Williams, 1992). Another example of a career typically held by females is working as a health promotion professional in institutions of higher education. The membership coordinator of ACHA, A. Ticker, stated that in 2011-2012, 16.2% of the 481 members of the ACHA Health Promotion Section who disclosed their



biological sex were male (personal communication, November 5, 2012). Additionally, the findings of the NASPA *Wellness and Health Promotion Benchmark Survey 2011* (National Association of Student Personnel Administrators, 2011) show that 23.5% of directors of health promotion departments who completed the survey were male. However, I found no literature focused specifically on men working in this profession.

The National Center for Education Statistics (2012b) found that for the 2009-2010 academic year, those institutions that offered degrees in health professions and related studies conferred the majority of them to women. The *ACHA Health Promotion Needs Assessment 2012* (American College Health Association, 2012a) reported that the majority of respondents from the Health Promotion Section had their degree in a health-related field including Public Health, Health Education/Health Promotion, Community Health, and/or Health Care Administration. The fields discussed in the *ACHA Health Promotion Needs Assessment 2012* (American College Health Association, 2012a) are consistent with the degree fields dominated by women.

While there is current literature on men working in nontraditional careers, I could find no literature regarding men working in the field of health promotion or, more specifically, on men working in health promotion in the higher education setting. I completed this study, as I discuss in subsequent chapters, in order to understand why certain men choose to enter this career and why they maintain their employment. By better understanding the men who chose to enter and remain in this field, I hoped to understand how colleges and universities may recruit and retain more male professionals.

I designed the study to gain insight from men who worked in health promotion in higher education about why and how they entered their career as well as what encouraged their



persistence. The following literature review reports information regarding nontraditional careers for men as well as information on student affairs and health promotion in higher education.

#### Organization of This Review

In the first section of this literature review, I discussed the research conducted about men working in nontraditional careers including types of nontraditional careers, factors associated with men entering these careers, and the issues and challenges associated with working in these careers. The second section examined research conducted about careers in student affairs and health promotion in terms of gender and current practices.

#### Men in Nontraditional Careers

Many of the nontraditional careers for men discussed in the literature, regardless of specific profession, involved professionals working with and helping people (Gillingham, 2006). Traditionally, professions that focused on helping and caring for others were considered careers for women because professionals in these positions had close contact and involved building relationships with others (Gillingham, 2006). Employers in nontraditional careers for men may expect employees to be nurturing, caring, empathetic, and sensitive (Cross & Bagilhole, 2002). Additionally, these positions may also involve "mothering" a student, child, client, or patient (Cross & Bagilhole, 2002; Sargent, 2005). These characteristics are not traditionally associated with men (Cross & Bagilhole, 2002). Men in health promotion in higher education need to work closely with students and help them, similar to the professions discussed above.



I found no research that examined men working in health promotion in higher education. However, there are several careers that prior researchers focused on about men choosing nontraditional career pathways. These careers included, but were not limited to, primary/elementary education (Galbraith, 1992; Lupton, 2006; Rice & Goessling, 2005; Sargent, 2004; Sargent, 2005; Simpson, 2005; Walsh, 1988; Williams, 1992), social work (Gillingham, 2006; Williams, 1992), nursing (Galbraith, 1992; Simpson, 2005; Walsh, 1988; Williams, 1992), family and consumer science education (Werhan, 2010), clerical work (Lupton, 2006), and inflight services (Simpson, 2005). I also located one article about men working in student affairs in higher education; however, health promotion was not discussed in that research (Calhoun & Taub, 2014). Rather than focusing on one occupation, other articles focused more broadly on men in nontraditional careers (Cross & Bagilhole, 2002; Heppner & Heppner, 2009; Jome, Surething, & Taylor, 2005). There may be other female-dominated careers covered in the research; however, the majority of the research centered on the careers listed above.

#### Job Requirements for Men in Nontraditional Careers

Male professionals employed in nontraditional positions may have certain characteristics in common and engage in similar activities. However, the literature discussed some requirements relate to specific careers. Male early-childhood educators indicated that some female teachers asked male teachers to be the disciplinarian in their classroom or asked male teachers to play the role of "father figure" for students who did not have a father at home (Sargent, 2004). Additionally, male teachers indicated that they experienced discouragement against hugging or showing affection toward students because they could be viewed as



pedophiles (Sargent, 2005). These two studies may not account for the experiences of all male teachers, because Sargent (2004, 2005) only focused on early-childhood education.

The two male family and consumer science (FCS) educators interviewed by Werhan (2010) stated that male students saw them as role models and that administrators encouraged them to show students that FCS classes provide important education regardless of a person's gender. Library employees may expect male librarians to lift heavy objects, clean vomit in the lobbies, repair machines, and assist with other manual labor (Carmichael, 1992). Men in student affairs reported taking on more manual labor because of their gender (Calhoun & Taub, 2014).

While some demands and characteristics needed to enter a career may be similar regardless of position, there may also be important demands specific to each profession. Therefore, men entering a nontraditional career may find it important to be aware of the demands of the specific career they choose to enter. Men interested in entering the career of health promotion in higher education could review the ACHA's *Guidelines for Hiring Health Promotion Professionals in Higher Education* (2014a) for information on requirements of the field.

Prior to this study, there was no information available that focused on men working in health promotion in higher education and it was unknown what similarities or differences existed between this and other nontraditional careers for men. In completing this study, I spoke directly to practitioners and was able to discuss specific demands that men in health promotion experienced and how these demands impacted their perceptions of their career. By exploring this, I was able to uncover demands that positively or negatively impacted their careers, and I discuss these factors in Chapter 4.



#### **Entering Nontraditional Careers**

#### **Education**

One entry factor mentioned in the literature related to nontraditional careers was educating men about the potential of entering such a career. Galbraith (1992) suggested elementary school educators "focus on informing men who are potential educators of children what other men who are currently employed as elementary educators find rewarding and important in their careers" (p. 252). Similarly, Rice and Goessling (2005) stated that male students in special education preparation programs could connect with male academic advisors and male co-operating teachers. They also suggested the creation of male cohorts. Connecting male students and newer male practitioners with more seasoned men in the field of education may be beneficial. However, a quantitative study conducted by Betz and O'Connell (1992) about engineering and nursing students indicated that connecting with someone working in the field, regardless of the worker's gender, could help motivate a man to enter a nontraditional career. In addition, regardless of the specific career, colleges and universities could set up internship programs that pair a male student with a male professional currently working in a field. Another way to create connections is to have male practitioners speak to classes about working in a nontraditional career.

For the purposes of this study, I asked participants about their interactions with professionals working in health promotion in higher education and whether these interactions had any influence on their decision to enter the field.



#### Influence From Others

A second factor for men choosing a nontraditional career was for them to have a positive influence in their life who did not hold them to the traditional expectations of what was masculine. In the two qualitative studies on male gender identity development conducted by Harris and Edwards (2010), the researchers found that older men influenced college-age men in regard to the norms and expectations of masculinity. "The participants shared stories about teachers, parents, coaches, and historical figures who openly challenged traditional norms and expectations of expressing masculinity while still being respected as men" (p. 53). The structure of this article is unique in that Edwards completed one qualitative study by interviewing 10 male participants, and Harris completed a separate study by interviewing 12 male participants.

Additionally, Harris had 56 men participate in focus groups. Harris and Edwards (2010) discussed the shared findings of both studies in the one article.

Having this type of influence could make young men more comfortable with pursuing a nontraditional career. High schools and institutions of higher education could expose young men still in school to professional men who challenge norms. I explored influencers with this study's participants and whether these influences challenged the norms of what it is to be a man.

Similarly, Tokar and Jome (1998) conducted a study on how gender roles and expectations can influence career interests and career choice. The researchers studied 100 male engineers and 100 male school counselors who were randomly selected for participation. Each participant completed a survey on job satisfaction, the Occupation Choice Dilemma Inventory, a Male Role Norms Scale, and Gender Role Conflict Scale (Tokar & Jome, 1998). The findings suggested that men who received a more traditional orientation of what it means to be masculine were more likely to have an interest in and engage in traditionally male careers and activities.

However, this study examined only engineers and school counselors. Therefore, if the researchers had included other careers, the findings might have been different (Tokar & Jome, 1998).

Betz and O'Connell (1992) focused on male and female nursing and engineering students and the opportunity structure approach. The opportunity structure approach suggests that professionals working in a person's career of interest are influential regardless of the influencers' gender/biological sex (Betz & O'Connell, 1992). "The findings suggest that insider status, not the same sex, is more important in the nontraditional choice of these women. Similarly, male nurses cite female insiders" (Betz & O'Connell, 1992, p. 103). Betz and O'Connell collected the data in 1985-86; therefore, the findings may not be true today. I found no other studies using the opportunity structure approach to research people entering nontraditional careers. Though Betz & O'Connell focused on only two careers, the findings indicated that young men may not necessarily need a male influencer.

Engaging in a mentor relationship may encourage someone to enter a nontraditional career. Koberg, Boss, and Goodman's (1998) findings from a study of health care professionals "indicate that the following influence psychosocial mentoring: protégé characteristics (ethnicity and education), gender and racial composition of the mentoring dyad, and group characteristics (intergroup trust and leader approachability)" (p. 68). In other words, someone was more likely to have a mentor relationship with someone who was similar in race, ethnicity, gender, and educational background. Additional research focusing on gender, insider status, and the mentor relationship may be helpful to determine how these three variables influence one another.

Participants in Werhan's (2010) study found motivation to enter a degree program in FCS through contact with departmental administrators, advisors, and faculty members. Advisors



and faculty members became role models and supported participants to finish their educational program and enter a career in FCS (Werhan, 2010). These findings indicated that having a role model or motivator working in nontraditional degree programs may motivate men to obtain the education needed to enter a nontraditional career. However, this study included only two participants; therefore, more exploration into how degree programs influence male students is needed. I asked my participants about their educational experiences and about their interactions with faculty and staff in their degree programs. This allowed me to learn what role faculty and staff played in influencing participants to enter into health promotion in higher education.

Men currently working in nontraditional careers have the potential to be positive role models for those considering entry into a field (Gillingham, 2006). Williams (1992) completed 99 in-depth interviews with 76 men and 23 women in female-dominated careers from 1985-1991. The researcher included only four careers, and 90% of the participants were White; therefore, the findings may not be transferrable to all men in all nontraditional careers. The researcher found that many men entering into nontraditional careers described experiencing positive interactions with professors, colleagues, and supervisors (Williams, 1992). By learning more about prior influencers and mentors in the lives of the participants I interviewed, I gained more information about who influenced these men to enter into health promotion in higher education.

#### Previous Interest in the Career

Having an interest in the profession was another potential motivator for career entry. Werhan (2010) found that participants' interest in the content covered in the FCS degree program influenced them to obtain that degree and enter into that career. Werhan's study



focused on only two participants; therefore, other male FCS teachers may not have had the same interest. Similarly, Sargent (2004) found that participants in his study of male teachers in elementary schools had a genuine interest in working with and educating children.

Carmichael (1992) completed a study of male librarians by using the *American Library*Association Handbook of Organization and Membership Directory; he surveyed 482 men. The use of a directory was problematic because he was not always able to determine whether a name was male or female; therefore, there is some selection bias. The researcher asked participants questions regarding entry into the career, gender issues, male stereotypes, and demographics.

Carmichael found that more than half of the participants chose to become librarians because of a love of books and almost half indicated that experiences in libraries impacted their decision.

The findings of these studies suggested that an interest in the subject area of a field may play a role in the decision to enter a nontraditional career. During this study's interviews, I asked participants about their career interests. By doing this, I was able to learn more about when and how the men developed an interest in this field.

# **Personality**

Personality was another factor found to influence men who enter a nontraditional career. Galbraith (1992) conducted a study comparing men and women working in nontraditional and traditional careers. The researcher compared men working in nursing and elementary education with men entering the more traditional career of engineering. The results of this study indicated that men working within the nontraditional careers found relationship building with their peers and clientele more important than did men working in the more traditional profession. However, men in both nontraditional and traditional careers indicated that power, money, and prestige



were also important. Galbraith (1992) concluded that while there may be some characteristics that are different between men who chose to enter nontraditional careers and those who did not, there might also be similar personality traits between the two groups.

Galbraith (1992) stated that the *Important Components of a Career Scale (ICC)* used in his study might not have accounted for all of the factors men deem important in their careers. Therefore, there may be additional factors influencing men to enter nontraditional careers. In the current study, I asked participants about what they deemed to be the benefits of their career. These findings are described in subsequent chapters.

Dodson and Borders (2006) completed a study of 100 male engineers and 100 male school counselors that examined gender role attitudes and gender role conflict. The participants completed a job satisfaction survey, the Occupational Choice Dilemma Inventory, and the Male Role Norms Scale. The researchers found that male counselors were more willing than male engineers to make sex-atypical choices in order to have roles with prestige. Additionally, the male engineers had a more masculine ideology than the male counselors. Dodson and Borders (2006) focused on only two careers, almost all participants were White, and all were from the southeastern United States. Therefore, including more varied participants may lead to different findings.

Lemkau (1984) completed a study that compared the personalities of men in sex-typical professions with those in sex-atypical professions. The researcher included men from the Atlanta area who had completed a bachelor's degree, were 50 years of age or younger, and were born in the United States. Each participant completed a biographical questionnaire, a personality inventory, and the Bem Sex Role Inventory. Lemkau (1984) found that men employed in atypical positions had different personality scores and engaged in different behaviors than men



in typical professions. The sample in the study was very homogeneous, with the vast majority being White and educated. Additionally, the study was 30 years old; therefore, the findings may not hold true. I did not have this study's participants complete a personality or sex role inventory.

Holland's (1962) theory of career entry stated that men and women will choose a career that matches their personality type. Holland (1985) explained that there are four assumptions guiding the theory: most people are one of six personality types, there are six types of workplace environments, people will choose an environment that allows them to thrive, and a person's behavior will be determined by the interaction between his/her personality and the environment in which he/she works (pp. 3-4). Holland (1985) stated that a person's career choice is a way for him/her to express his/her personality type, and people working in the same field will have similar personalities. In order to determine a person's personality type, Holland (1962) developed the Vocational Preference Inventory (VPI). "The VPI is a personality inventory closely tied to vocational interests and composed of 160 occupational titles to which the respondent indicates like or dislike" (Walsh, 1988, p. 630). Holland's (1985) theory did not take gender into account.

In a quantitative study, Walsh (1988) examined Holland's (1985) theory by comparing 151 men and women employed in traditionally female careers including nursing, elementary education, and library work. The researcher utilized the VPI to determine the personality types of participants. The researcher found a significant difference in the VPI scores for men and women, regardless of career (Walsh, 1988). However, the findings also indicated that men and women employed in traditionally female careers had similar scores in the "social" personality



area (Walsh, 1988). I did not utilize the VPI in the current study, as the inventory only examines one factor of career entry.

Jome et al. (2005) conducted a study that examined masculinity and career interests. The research studied 166 men in upstate New York who were "employed full-time, not currently a student, and working in their general field or career for at least 2 years" (p. 186). The participants completed the Male Role Norms Scale, the Gender Role Conflict Scale, and the Vocational Preference Inventory. The participants working in nontraditional careers showed an interest in careers in which they worked closely with people. Participants working in traditional careers for men did not have that interest. Based on the findings of the study, men interested in nontraditional careers prefer working in environments involving interaction with other people (Jome et al., 2005). I did not ask participants to complete the above scales; rather, I focused on asking open-ended questions so I could learn more about why they chose their current career.

### Perception of the Career

A final factor influencing career entry was how men perceive their career. Williams (1995) found that several of her study participants stated that they did not perceive their career as traditionally female or female-dominated when they chose to enter. Specifically, participants in the study stated that the idea that nursing, social work, or librarianship was a feminine career was a false stereotype (Williams, 1995). Based on the findings of the study, if the perception is that a career is not nontraditional, a man might be more willing to enter. Before I started this study, I did not know whether men working in health promotion in higher education perceived

the career as nontraditional. By completing this study, I learned that participants perceived health promotion in higher education as a female-dominated profession.

### Theories of Career Entry

The theory of career choice developed by Gottfredson (1996) discusses how people, throughout the course of their lives, develop and implement plans to accomplish their career goals (Gottfredson, 1996). According to Gottfredson (1981), people move through four life stages: a)from age 3-5, they experience orientation to size and power; b) from age 6-8, they experience orientation to sex roles; c) from age 9-13, they experience orientation to social valuation or social class; and d) from age 14 on, they experience their orientation to their own self. During each of the stages, people orient themselves to the career types they can pursue based on their interactions with society and through self-assessment (Gottfredson, 1996). The theory focuses on several major concepts that impact career choice including a person's sense of self, his/her social space or environment, career stereotypes, how a person's self-image impacts compatibility with certain careers, the accessibility of certain careers, and how people go through the process of circumspection and compromise when deciding upon a career (Gottfredson, 1996). Gottfredson's (1996) theory accounts for the sex and gender expectations of a person in how he/she makes his/her choice of career.

Chusmir (1990) developed a model of career choice to study men who make nontraditional career choices and proposed that both personal and external influences impact career choice. Personal influences include one's background, attitudes and values, and intrinsic needs. External influences include the family and society. The factors that influence men impact both their decision to enter a nontraditional career and the positive or negative outcome of the



decision (Chusmir, 1990). Lease (2003) tested Chusmir's model on 354 men and women entering into traditional and nontraditional careers for their gender; however, he did not specify the careers. The researcher obtained the data from the 1986 and 1990 Cooperative Institutional Research Program (CIPR) and reviewed specific data on "involvement in college activities, extracurricular activities, and family and occupational goals" (p. 247). Lease found that some of Chusmir's factors for career choice played a role male participants' career choice. Impactful factors included social ideology, socioeconomic status, and interest in a prestigious career. Factors with no impact included academic ability, interest in raising a family, and mother's career choice.

In the current and subsequent chapters, I discuss similar internal and external factors that played a role in my participants' choice to enter a career in health promotion in higher education. While the theory and the model do not account for all factors, I used this information as a starting point when analyzing the findings from this study. I also examined what additional factors played a role in participants' decision to enter health promotion in higher education.

### Preparation for and Persistence in Nontraditional Careers

In order to prepare men to enter a nontraditional career, it may be important to create an encouraging environment where men can explore their definition of masculinity. There are many factors that influenced how college-age men defined masculinity including "parents, coaches, teachers, media, and sports" (Harris & Edwards, 2010, p. 49). Institutions of higher education could play a role in defining what is considered masculine. The culture of a college or university campus could reinforce traditional gender expectations or help to eliminate these



expectations (Harris & Edwards, 2010). Men may feel more comfortable entering nontraditional educational programs if the culture of the program and institution were accepting and encouraging (Harris & Edwards, 2010). Faculty and staff could intentionally encourage men who chose to enter a degree program leading to a nontraditional career (Werhan, 2010).

Connecting men currently working in a profession with younger men was one factor that appeared to assist men preparing for entry into a nontraditional career. In order to develop these types of connections, Galbraith (1992) suggested that men currently working as elementary teachers could describe the positive aspects of their careers and the importance of having male educators in elementary schools with new male teachers. This example focused specifically on men working in elementary education. However, educators could bring men working in other types of nontraditional careers into classroom settings to speak with students about their career. Similarly, having a number of men within a graduate preparation program may allow them to make connections with their peers and future colleagues (Calhoun & Taub, 2014). Creating these types of relationships could assist men preparing for a future nontraditional career.

Dohner, Loyd, and Stenberg (1990) completed a mixed-methods study of male home economics teachers. Twenty-four men completed a questionnaire about their age, education, salary, and other demographics. Additionally, 11 of the men completed phone interviews. The researchers found that male home economics teachers fell into their positions, meaning that they did not plan their career but ended up there by chance. The researchers suggested that "men in the field should serve as role models in recruiting, and mentoring, other men" (p. 35).

Additionally, the researchers stated that professionals should set up meetings and conferences for male home economics teachers in order to assist with the profession's male recruitment and



retention. The researchers studied participants in Ohio only; therefore, the findings may not be true for male home economics teachers across the country.

Similarly, seeking out a mentor once in a position could also encourage men to persist in a nontraditional career. Successful men working in a nontraditional career can act as positive role models for newer professionals navigating their first position (Gillingham, 2006). As previously discussed, Koberg et al. (1998) found that professionals were more likely to have a mentor who was similar in race, ethnicity, gender, and education. Similarly, Calhoun and Taub (2014) found that entry-level men in student affairs were influenced positively by male mentors. However, other research indicated that having a mentor within the same chosen nontraditional profession, regardless of gender, assists men working in a nontraditional career (Betz & O'Connell, 1992).

I asked participants about their education and preparation for their career. I learned about mentors and influencers and about how these relationships assisted with participants' decision to enter and remain in their career. By learning how the participants prepared for their career entry, I learned about how professionals could assist future male health promotion professionals to prepare to enter the field.

### Possible Issues and Challenges

The literature discusses several challenges and issues for men working in nontraditional careers. The issues and challenges discussed in this section may be factors that influence a man's decision to leave a nontraditional profession.

#### Pressure to Advance

One challenge was the encouragement of men to advance into administrative roles within their professional field. Simpson (2005) completed 40 interviews with men working as primary school teachers, flight attendants, librarians, and nurses. Simpson recruited participants using professional journals, professional websites, and personal contacts and, therefore, these participants "may be more positive about their job than other men who may not have felt motivated to offer themselves for an interview" (p. 378). The majority of study participants indicated that they wanted to remain at the lower-levels of their professional hierarchy and that they did not have an interest in advancing into more administrative roles. However, the men noted that they sometimes felt pushed to advance within their profession (Simpson, 2005). Williams (1992) stated that "some men described being tracked into practice areas within their professions which were considered more legitimate for men" (p. 296). So called legitimate positions included those in upper administration, away from general practice (Williams, 1992). Carmichael (1992) found that male librarians "are expected to be administrators, and are apt to be considered weak, unambitious, or inferior specimens of men if they decline invitations to rise in the hierarchy" (p. 433). Finally, Sargent (2004) highlighted a myth within early-childhood education that men will teach for a short period of time, but they focus on upward advancement within the field of education or are waiting for a more masculine job opportunity. Some participants indicated that others looked down on them for not advancing into more administrative roles even though it was their choice to remain a teacher (Sargent, 2004).

While some may see the encouragement of advancement as a challenge, other research indicated that encouraging men to advance within their profession may be an advantage to upper level positions (Williams, 1992). For example, Lupton (2006) stated in his literature review



about men in nontraditional occupations that some men found being the "token" male made advancement easier into the upper-level positions of their profession. Additionally, Calhoun and Taub (2014) found that some men in student affairs were motivated by "career advancement and having more power, influence, and control" (p. 46).

I explored career trajectory with participants. I asked about their career path and career goals for the future. By examining this information, I learned whether participants planned to remain in the field, whether they planned to advance within higher education, or whether they planned to leave the profession. The ACHA's *Guidelines for Hiring Health Promotion*Professionals in Higher Education (American College Health Association, 2014) indicated that staff and director are the two professional levels in the profession. However, some departments also had assistant-director or senior-administrator positions. Even with those positions added, once a professional reaches the director-level position, there were likely no advancement opportunities within the field of health promotion in higher education.

# Male Privilege

Similar to the discussion of advancement in female-dominated careers, male privilege was seen as both a positive and a negative factor for men working in nontraditional careers. While no participants worked in health promotion in higher education, Calhoun and Taub (2014) found that "being a man within student affairs carries with it an obligation to counter traditional gender stereotypes and combat the negativity that exists about men, power, and privilege" (p. 45).

Giesler (2013) completed a study on how male social work students experienced privilege in both the classroom and in the field of social work. The researcher interviewed 10



male students currently in the social work program, seven male alumni of the same program, and six members of the social work faculty. The researcher also conducted focus groups with current female social work students. The participants all attended one school; therefore, the findings of the study may not be generalizable to all social work students. The researcher found that during their field experiences, male students "experienced firsthand both male privilege and the stress that its expectations placed on them with few opportunities to process that tension" (p. 210). Participants explained that field instructors expected them to be role models for the male clients they were working with and that they were tasked to connect with men in the population they served. Some participants stated that they received more interesting placements simply because they were male. While participants found these factors to be positive, they also experienced challenges. One participant working with victims of domestic violence was worried "that in his female clients' eyes, he represented the oppressor" (p. 211). The male participants also mentioned that faculty did not create a space to discuss gender-based concerns like the one mentioned previously. Therefore, male privilege can be seen as both a benefit for men in nontraditional careers, and a potential challenge. I asked participants in the current study about what benefits they perceived as a man in health promotion in higher education.

### **Gender Discrimination**

Unlike the research previously discussed in which there was encouragement to advance, some researchers found that some men in nontraditional careers experience discrimination.

Williams (1992) studied men entering four nontraditional careers: elementary education, nursing, social work, and librarianship. While the researcher found that some employers preferred to hire men, there were accounts of discriminative practices including barring men



from teaching the youngest grade levels in some school districts and from working as nurses in pediatrics or gynecology (Williams, 1992). However, Williams conducted this study over 20 years ago; therefore, these findings may no longer be true. Additionally, Lupton (2006) stated that some male early-childhood educators face false accusations of being perverts or pedophiles because of their career choice.

Hickey (2006) completed a qualitative study of male librarians and their experiences in and perceptions of their career. The researcher recruited young male librarians, all working in different sections of the same library. The researcher found that the men did not feel they were part of the "in group" and that older, female colleagues would reminisce about the golden age of libraries prior to male employment. The participants indicated that the "in group" made them feel uncomfortable during these conversations. Hickey studied men at only one library; therefore, the findings may not be true of men working in other libraries.

Johnson (2010) suggested that professionals need to reflect on and be aware of their own biases in regard to men in nontraditional careers and look at how their practices reinforce cultural stereotypes of what is women's work and what is men's work. Being aware of this could help decrease discriminatory attitudes and practices that keep men from entering or maintaining their employment in historically female occupations. I had planned to explore any incidences of discrimination described by participants. However, participants did not describe any outright discrimination based on gender.

### Salary

In a study completed by Hogue, DuBois, and Fox-Cardamone (2010), the researchers utilized the gender of participants to explore pay expectations. The participants completed self-



esteem scales and disclosed the field they planned to enter, the position they intended to hold, and their salary expectation. The researchers utilized the U.S. Department of Labor Statistics to gender-type the jobs the participants indicated. The participants were 435 undergraduate students at a midwestern university. The researchers used a convenience sample; therefore, the findings of this study may not be representative of all undergraduate students. The researchers found that "women and those intending to hold female-dominated jobs expect lower pay than men and those intending to hold male-dominated jobs" (p. 222). However, Calhoun and Taub (2014) found that the need for a higher salary could influence entry-level men in student affairs to leave the profession. I did not ask specific questions regarding salary during the interviews; however, the salary of health promotion professionals working in higher education was mentioned by participants.

### Role Strain

Role strain was another challenge that some men experience when they enter nontraditional careers. Simpson (2005) stated that "conflicts can occur between the need to maintain masculine identities and the feminine associations or demands of the job" (p. 371). Some men in the study described feeling embarrassed or judged by others for having a nontraditional career (Simpson, 2005). Experiencing role strain could lead men to question continuing to work in a nontraditional career. Calhoun and Taub (2014) found that fear of femininity was a component of gender role conflict for men in student affairs but one that did not negatively impact most participants. Simpson found that experiencing role strain influenced three study participants to leave their nontraditional careers.



While some men reported leaving their career after experiencing role strain, not all men chose to leave their profession. Some men developed coping mechanisms in order to remain in their chosen career. Simpson's (2005) participants described coping mechanisms such as not sharing information about their career with their family and friends or focusing on the more masculine aspects of their career. Cross and Bagilhole (2002) conducted a qualitative study in which they examined "ways in which masculinities are defined, (re)constructed, and maintained by men working in nontraditional jobs" (p. 210). The researchers found that some men retained their masculine identity by not working closely with female colleagues. Additionally, participants reported discussing with others how men are more successful in these positions than female colleagues. Other participants described reconstructing their definition of masculinity to help cope with role strain (Cross & Bagilhole, 2002). While the findings of Cross and Bagilhole's study may offer possible coping mechanisms for men employed in nontraditional careers, the findings are not generalizable. All 10 of the participants worked in different nontraditional careers and reported differing amounts of work experience. Therefore, it is difficult to determine which coping mechanisms may work for each of the nontraditional careers represented in this study. In summary, though role strain may be a challenge for some men, there are mechanisms that some of them develop to cope. I did not ask participants specifically about role strain; however, several discussed challenges of being male in a female-dominated workplace.

# Stereotypes Regarding Sexual Identity

Another challenge was that some men working in a nontraditional career reported being stereotyped as homosexual. Simpson (2005) noted that one of the stigmas that participants



discussed was the assumption of homosexuality or sexual deviance. Sargent (2004) stated that men working as early-childhood educators, regardless of their sexual orientation, experienced homophobic reactions from others both inside and outside of their profession. Werhan (2010) stated that participants indicated they were perceived as homosexual. Carmichael (1992) found that 81% of participating male librarians stated that a stereotype of male librarians was that they are gay. Giesler (2013) found that some participants made a point to share their heterosexual identity. Calhoun and Taub (2014) stated that men in student affairs indicated being perceived as homosexual was a component of gender role strain but was not something that seemed to negatively impact participants.

While men in nontraditional positions may face misperception as homosexuals, a study conducted by Jome et al. (2005) found that men employed in nontraditional careers, regardless of their sexual identity, were more accepting of homosexuals. These findings suggested that men in nontraditional careers are more accepting of homosexual men than those working in more traditional occupations. I did not ask participants about their sexual identity; however, all five brought it up during our discussions.

# Perceptions of Men

To some people, a man in a nontraditional career was seen as having accepted a "step down" or a demotion (Williams, 1992). Carmichael (1992) found that a stereotype of male librarians was that they lacked power and ambition. According to Williams (1992), "this is a crucial difference from the experience of women in nontraditional professions: 'My daughter, the physician' resonates far more favorably in most peoples' ears than 'My son, the nurse'" (p. 262). As stated previously, Williams conducted her research over 20 years ago, and this



perception may no longer be relevant. However, some men may feel the need to justify why they chose their nontraditional career or they may hide their profession from others.

Based on the prior literature, it is important to consider the issues and challenges that men may face when working in nontraditional careers. They may experience discrimination or false perceptions by people both inside and outside of their profession. I explored the challenges experienced by men working in health promotion in higher education. Practitioners and researchers can utilize the information in this study to help create environments in which men feel more comfortable entering and persisting in nontraditional professions.

Careers in Higher Education, Student Affairs, and Health Promotion

Higher education is one setting in which health promotion professionals work. Higher education

informs public understanding, cultivates public taste, and contributes to the nation's wellbeing as it nurtures and trains each new generation of architects, artists, authors, business leaders, engineers, farmers, lawyers, physicians, poets, scientists, social workers, and teachers as well as a steady succession of advocates, dreamers, doers, dropouts, parents, politicians, preachers, prophets, social reformers, visionaries, and volunteers who leaven, nudge, and shape the course of public life. (Rhodes, 2001, p. xi)

According to the National Center for Education Statistics (2012a), institutions of higher education vary greatly across the United States and provide a wide variety of educational programs including academic, technical, and continuing education programs. The 1998 Amendments to the Higher Education Act of 1965 stated that an institution of higher education can only admit students, who completed high school or an equivalent certificate program, are authorized by a state, are accredited by a recognized agency, and are public or nonprofit. Additionally, the institution must provide no less than a bachelor's degree, or 1 year of technical



training, or no less than 2 years of academic credit that can be applied toward a bachelor's degree program (U.S. Department of Education, 1998).

#### Student Affairs

According to the NASPA (National Association of Student Personnel Administrators, 2011), divisions of student affairs house the majority of health promotion departments in higher education settings. Similarly, the *ACHA Health Promotion Needs Assessment 2012* (2012a) found that divisions of student affairs housed 89% of participating health promotion departments. Therefore, it was important for me to not only understand the health promotion department, but also the student affairs divisions in which the majority of these departments reside. Student affairs professionals work with students when students are outside of the classroom and focus on learning and student/personal development (American College Personnel Association, 1996).

Student affairs and services vary greatly from institution to institution, as can the different departments housed within student affairs (Blimling, 2002). Regardless of the institution, student affairs

works in partnership with the academic faculty to clarify or define and achieve satisfaction of broadly desirable, student-focused educational goals; the success of this work results both from the unique competencies, experiences, and expertise of student affairs and from ways in which student affairs supports the work of learning that is done everywhere in the institution. (Keeling & Dungy, 2004, p. 20)

Student affairs professionals are expected to create leadership development opportunities for students, create an inclusive environment on campus, conduct themselves in an ethical manner, mentor and advise students and student groups, conduct evaluations and assessments, assist with



student learning and development, among many other expectations (American College Personnel Association & National Association of Student Personnel Administrators, 2010).

The Council for the Advancement of Standards in Higher Education (CAS) developed standards and self-assessment guides for 43 functional areas in higher education that include many functional areas within student affairs (2012). Some of these areas included, but are not limited to, admissions, financial aid, registrar, career services, Greek life, judicial affairs, residential life, student activities, orientation, counseling, clinical health, health promotion, multicultural and LGBT resource centers, military student services, disability services, and recreation (Biddix, 2011; Council for the Advancement of Standards in Higher Education, 2012). Therefore, there are many varied opportunities for specialization for professionals, including health promotion.

# **Motivation for Entry**

Most young people do not plan to enter a career in student affairs, and it is not a career option discussed in most secondary schools (Blimling, 2002). While student affairs may not be a planned career path, the literature described factors regarding motivation for entering a career in student affairs.

One factor was the influence of a professional already working in the field. Several of the senior student affairs officers (SSAOs) participating in Blimling's (2002) study discussed how professionals in student affairs supported them as an undergraduate student. Blimling asked SSAOs to reflect on their career path. The number of participants and their gender were not included in Blimling's study; therefore, the information may not be true of all SSAOs.

Additionally, Hunter (1992) completed a study of 93 students enrolled in student affairs



master's degree program from 1985-1989. The researcher asked participants during their first week in the program to complete the following: "a succinct summary (approximately four pages) of your career path. How did you select student affairs work?" (p. 182). Hunter found that 73% of participants described receiving encouragement to pursue a career in student affairs from a student affairs professional working at their undergraduate institution. Taub and McEwen (2006) studied 300 master's degree students in 24 randomly selected graduate programs. The participants completed the Student Affairs Entry: Factors Affecting Career and Graduate Program Choice, a survey developed to gather information on the "process of career choice, graduate program selection, and future plans" (p. 209). Taub and McEwen found that 80.3% of the participants stated that they received encouragement from someone to enter a career in student affairs. The participants indicated that the majority of encouragers were professionals currently working within a student affairs department (Taub & McEwen, 2006).

The findings of these three studies suggested that professionals working within student affairs may serve as important tools for motivating potential new student affairs professionals to enter the field. In the current study, I learned about how student affairs professionals encouraged participants to pursue a career in health promotion in higher education.

Another factor impacting motivation for career entry was how the field attracts some future professionals because they find the work personally fulfilling, and it compliments their values. Hunter (1992) found that some study participants "acknowledged they were attracted to student affairs because they felt their values and priorities were consistent with those of the professionals who had served as role models for them" (p. 184). Taub and McEwen (2006) found that student affairs attracted their participants because they found the work fulfilling. The findings of these two studies suggested that having values and priorities that compliment



working in student affairs may motivate some to enter the field. I did not specifically ask participants about their values during the interviews.

A final factor that may motivate someone to enter student affairs was for an undergraduate student to have a positive experience as a student leader on campus. Taub and McEwen (2006) found that the majority of the participants indicated that while they were undergraduate students, they had been involved with student affairs departments as student leaders or paraprofessionals. Similarly, Hunter (1992) found that student affairs departments including admissions, residence life, and Greek life employed participants as undergraduates. Some participants stated that their experiences working in these departments encouraged them to begin a career in student affairs (Hunter, 1992). Some SSAOs began their career in student affairs by working as a student worker in residence life, Greek leadership, or student government (Blimling, 2002). The findings of these three studies indicate that involvement in a department within student affairs as an undergraduate student may be an important motivating factor for career entry.

Student affairs professionals could play an important role in motivating current students to consider entering a career in student affairs. Professionals can encourage undergraduate students to become involved with a student affairs department as a student leader or a paraprofessional.

### Career Mobility & Persistence

At most institutions of higher education, the highest-level positions in student affairs is the Senior Student Affairs Officer. This position often varies in title from institution to institution and can include dean, assistant/associate vice president, or vice president of student affairs (Biddix, 2011). In order to reach one of these high-level positions, a professional must

persist in the profession. However, some professionals may not plan to advance beyond their current lower-level position or department. Blimling (2002) suggested that participants advanced to SSAO positions because of their skills and interest in working in these positions. Additionally, participants described the desire for a new challenge, rather than having planned for advancement all along. However, Daddona, Cooper, and Dunn (2006) conducted a study of 101 chief student affairs officers completing a doctoral degree. The researchers asked each participant to complete a web based survey on degree earned, job status, job goals, and salary expectations. The researchers found that more than half of the participants' goal was to advance to a dean, associate vice president, or vice president position upon completion of their terminal degree. Whether professionals plan career advancement or it happens by chance, there are several factors that impacted student affairs professionals' ability to persist and advance in their career.

# Relationships

One factor in career advancement within student affairs was building relationships with others in the field. As previously discussed, building relationships with professionals or mentors can be motivating for those choosing to enter both nontraditional careers and the field of student affairs. Renn and Hodges (2007) completed a grounded-theory qualitative study of 10 graduates of a student affairs master's program. Eight women and two men completed the interviews. Participants indicated that staying connected to a mentor continued to be important once working in the field and that building relationships with students and colleagues was important for a professional working with in a department in student affairs. These relationships could help professionals both personally and professionally (Renn & Hodges, 2007).



Similarly Renn and Jessup-Anger (2008) completed a mixed-methods study on the preparation of new student affairs professionals; however, the study did not include health promotion professionals. The researchers had participants complete an initial survey on educational background and current position. Every month after their initial contact, the researchers sent participants a question and asked them to complete a 300-500-word response. The researchers utilized open coding to analyze the written responses. Renn and Jessup-Anger found that "new professionals repeatedly discussed the crucial role that mentors and supervisors played in their lives during their transition into the field" (p. 328). The participants indicated that mentors and supervisors supported them, offered important advice, and offered feedback.

One study focused specifically on entry-level men entering student affairs. The purpose of Calhoun and Taub's (2014) research was to "understand the experiences of men at the entry level (first 5 years) of the student affairs profession" (p. 38). The researchers completed 18 semistructured interviews and a focus group with four other participants. The researchers explored identity and self-awareness with the participants. All worked at 4-year institutions throughout the United States. The participants represented various areas of student affairs, though the researchers did not include participants from health promotion. The researchers utilized member checks and an external peer debriefer to address trustworthiness. They found that participants assumed one of three different roles or identities: the traditional male role, the mentor role, or the change-agent role. Participants who identified with the mentor role said that "leadership, mentorship, and role modeling [helped] define their experience in student affairs" (p. 44). In turn, these participants expressed interest in becoming mentors to newer men in the field. In the current study, I asked about factors that impact persistence in the field and discuss these factors in subsequent chapters.



#### **Professional Involvement**

Another factor that may positively impact persistence in the field of student affairs was involvement in the profession outside of the home institution. A study conducted by Chernow, Cooper, and Winston (2003) examined student affairs professionals related to their career advancement and involvement in professional associations. The researchers surveyed 139 entry-, mid, and senior-level professionals using the Student Affairs Professional Involvement Questionnaire designed specifically for the study. The researchers recruited participants using the 1998 Directory of Student Affairs Personnel in Georgia. The researchers did not state the gender or other demographic information of participants. Therefore, the findings are not generalizable to all states or all student affairs professionals. However, the findings may be transferrable. The researchers found that "entry-level professionals tend to be at the formative and application stages where involvement in professional associations is crucial to a sound start in the field" (Chernow et al., 2003, p. 53). Additionally, both newer and more seasoned professionals stated that they engage in professional organizations by attending conferences, networking, being involved with committees, and participating in online discussions and listservs (Chernow et al., 2003). Involvement in a professional organization may therefore be important for professionals at all levels because it allows them to connect with others who work in similar careers as well as engage in professional development opportunities. I did not ask participants specifically about involvement in the field professionally; however, several of them spoke about their involvement in professional organizations.



### Career Satisfaction

Career satisfaction and morale were also factors that may impact persistence and career mobility. Rosser and Javinar (2003) surveyed 1,166 randomly selected mid-level student affairs professionals from across the country in regard to morale, work satisfaction, and intent to leave their career. The researchers used a morale and departure survey but did not identify the name of the instrument in their journal article. Rosser and Javinar found that there were several factors that led to professionals' decreased satisfaction and morale including working at one institution for a long period of time, feeling unhappy with life in their profession and at their institution, feeling discriminated against, being unrecognized for their professional work by their institution, and having negative relationships with other professionals and/or departments at their institution. Working at an institution where professionals encounter factors that lead to low satisfaction and morale impacted their ability to persist in the field and may lead to a decision to leave their institution or their career (Rosser & Javinar, 2003).

The literature indicated that involvement with professional associations, and having relationships with other professionals may assist those who want to persist and advance in their career. Additionally, the research indicated that a high level of satisfaction and morale might assist with persistence. I asked participants about benefits and challenges in their work and found that some of the factors described in the literature were true for them.

# **Gender and Student Affairs**

Turrentine and Conley (2001) completed a study that reviewed the Integrated Postsecondary Education Data System, an online directory of practitioners maintained by the American College Personnel Association. The researchers reviewed 31,276 participants who

completed degrees in 1995-96 and 1997-98. Turrentine and Conley examined the diversity of entry-level professionals in student affairs and found that the majority of graduates from student affairs-related graduate programs were female. The findings of two more recent studies indicated that the majority of participants enrolled in student affairs graduate programs were women (Renn & Hodges, 2007; Taub & McEwen, 2006). Though women composed the majority of the preparation programs, Turrentine and Conley also found that men held the majority of senior-level student affairs positions. While more women than men prepared for entry into the field of student affairs, they did not appear to advance into senior-level positions.

Taub and McEwen (2006) focused their study on why graduate students chose to enter the field of student affairs, and the majority of their participants were female. However, the findings could also be transferrable to men interested in student affairs. Both the male and female participants stated that someone influenced them to enter into student affairs, and most of these influencers worked in student affairs. Additionally, both genders reported that they were confident in their choice to enter the field and planned to remain in the field long term (Taub & McEwen, 2006).

Calhoun and Taub (2014) focused their research on entry-level men in student affairs.

The men in the study indicated that mentors positively impacted their experience in student affairs, specifically male mentors. The researchers stated that there is a need to recruit more men into the field and that there is a misperception in the field that student affairs is not a legitimate career for men. Additionally, they suggested that more research needs to be completed on men in student affairs.

A literature review completed by Hamrick and Carlisle (1990) indicated that while more women than men were entering careers in student affairs, more men advanced into senior-level



administrative roles than women. Additionally, the researchers stated that a man's career in student affairs may be "devalued," as it is seen as a female-dominated profession. Engstrom, McIntosh, Ridzi, and Kruger (2006) completed a quantitative study using the 2001-02 NASPA Salary Survey. The researchers examined the salaries of 405 SSAOs using a web-based survey. The researchers recruited participants through the NASPA. The survey included questions about years of experience, number and type of positions held, and base salary. Engstrom et al. (2006) found that while women were well represented in student affairs, women were not as well represented among SSAOs. Based on the findings of these studies, women dominated the lower levels of student affairs while men dominated the SSAO positions.

#### **Health Promotion**

According to Green and Kreuter (1990), health promotion is the "combination of educational and environmental supports for actions and conditions of living conducive to health" (p. 313). Based on this definition, the focus of health promotion in the setting of higher education is on creating an environment that is conducive to the health of the campus community. The Council for the Advancement of Standards in Higher Education (2012) stated that health promotion services should focus upon creating a supportive, just, and equal living-learning environment in terms of health and wellness. Health promotion should also focus on primary or universal prevention policies and initiatives to prevent health and wellness problems (Council for the Advancement of Standards in Higher Education, 2012).

The first mention of a health educator employed at a college occurred in 1950 when Jackson College reported employing a health educator during a survey of state colleges in Mississippi (Robertson, 1950). According to Zimmer (2002), in 1954, Dr. Ruth Boynton at the



University of Minnesota developed the first health educator position in higher education. The third institution to report having such a position was Amherst College. This institution added a health education position in the 1960s. By the mid-1980s, 68 health educator positions existed at higher education institutions in the United States (Zimmer, 2002). By the early 1990s, the ACHA found that slightly more than 20% of participating institutions reported a health promotion program on their campus (Sloane & Zimmer, 1992). By the end of the 1990s, almost 80% of institutions involved in the ACHA reported a health education or promotion service on their campus (Zimmer, 2002).

The Task Force on Health Promotion in Higher Education, appointed by the ACHA, developed, implemented, and analyzed the National Survey on Health Promotion and Education in Institutions of Higher Education in 1998 (Zimmer, Hill, & Sonnad, 2003). The task force developed the survey to obtain information on strategic planning in health promotion departments, current strategies and resources used in health promotion departments, performance measures for best practices, gaps in a department's ability to meet goals, and the education and preparation of health promotion professionals (Zimmer et al., 2003). The task force found that the majority of the randomly selected institutions had some form of health promotion or health education programming. However, they also found that the majority of institutions did not have strategic planning and evaluation of health promotion services, a mention of health or quality of life in their institutional missions, or campus-wide networks devoted to health promotion initiatives. The findings of the task force indicated a lack of standardization in the field of health promotion in higher education and led to the development of the Standards of Practice for Health Promotion in Higher Education (SPHPHE) (Zimmer et al., 2003).



The *Standards of Practice for Health Promotion in Higher Education*, first published in 2001, is currently in its third edition, published in May 2012 (American College Health Association, 2012b).

Like previous editions, the third edition serves as a guideline for assessment and quality assurance of health promotion in higher education. The third edition also acknowledges additions to the body of knowledge and makes explicit the scope of practices and essential functions for the field. (p. 1)

Professionals use the SPHPHE in several ways including assessing their professional development needs to meet the standards, assessing health promotion departments, and as a way to help others outside of health promotion to understand the job of a health promotion department on a college campus (American College Health Association, 2012b). The SPHPHE is composed of seven standards of practice including Alignment with the Missions of Higher Education, Socioecological-Based Practice, Collaborative Practice, Cultural Competency, Theory-Based Practice, Evidence-Informed Practice, and Continuing Professional Development & Service (American College Health Association, 2012b). Each of the standards is composed of several sub standards. According to the *ACHA Health Promotion Section Needs Assessment* 2012 (American College Health Association, 2012a), 74% of participants use the SPHPHE in their professional practice. Being aware of and understanding the SPHPHE may therefore be important for those men who are working in health promotion in higher education.

The ACHA's Health Promotion Section developed the *Guidelines for Hiring Health*Promotion Professionals in Higher Education in 2008 based on the SPHPHE and the CAS standards. Updated in 2014, it states that

qualified health promotion professionals in higher education possess specific competencies that make them best suited to support student success through the practice of prevention – that is, by preventing the development of person and campus population-



level health problems, while enhancing individual, group, and institutional health and safety. (American College Health Association, 2014, p. 1)

The Guidelines for Hiring Health Promotion Professionals in Higher Education includes specific competencies possessed by qualified entry-level and senior-level health promotion professionals including developing research and strategic plans, using the socioecological model and evidence-informed practices, working with other departments on campus, and displaying cultural competence. It also states that there are primarily two types of health promotion positions: a staff role and a director role. "The director of health promotion provides direct oversight of all health promotion activities that address priority student health issues and support the missions of the university" (American College Health Association, 2014, p. 2). The staff role conducts the health promotion activities. While the guidelines list two distinct roles, the document indicates that some institutions may develop assistant director-level positions or senior administrative-level positions, and professionals can use the ACHA competencies as a foundation to develop these positions. The titles of health promotion professionals may not be consistent from school to school. For example, I worked at one institution where the directorlevel position had the title of coordinator, and at my current institution, the director-level position has the title of executive director.

I did not ask specifically about the *Guidelines for Hiring Health Promotion*Professionals in Higher Education when interviewing participants, but it was mentioned by some participants. The fact that there are only two roles described by the ACHA may mean that in order to advance within higher education, professionals may need to leave health promotion. If professionals want to pursue advancement in health promotion beyond the director, they may need to leave higher education.



Though the ACHA developed competencies, there were currently no academic preparation programs for those who want to pursue a career in health promotion in higher education. However, a dissertation completed by McNeil (2012) indicated that there was an institution of higher education in the United States that offered a course on health promotion in higher education as a part of its master's and doctoral degree of public health programs. The same institution was in the process of developing a health promotion in higher education certificate program. The purpose of McNeil's intrinsic case study was to "determine if the graduate-level academic preparation in health education is congruent with the essential and preferred skills for a Health Promotion Director, or sole practitioner, defined in the ACHA Hiring Guidelines" (McNeil, 2012, p. 7). Using the syllabi from the courses offered in the public health program and the Guidelines for Hiring Health Promotion Professionals in Higher Education, he found that the courses did not cover all of the competencies. Further, findings from interviews with three members of the graduate faculty at the institution and three of the developers of the Guidelines for Hiring Health Promotion Professionals in Higher Education indicated a need for a graduate-level academic preparation program for health promotion in higher education (McNeil, 2012). The lack of a formal academic preparation program may contribute to how few people, including men, enter the field of health promotion in higher education.

I found one published study that focused on health promotion professionals in higher education. Davidson (2008) completed a qualitative study of 141 health promotion professionals in higher education regarding "perceived continuing education needs and perceived job relevance, as related to the National Commission for Health Education Credentialing, Inc. (NCHEC) Responsibilities and Competencies" (p. 198). Some of the responsibilities and



competencies included, but were not limited to, administering, planning, implementing, and evaluating initiatives and programs; conducting research; communicating and advocating for the field; and serving as a resource person. The majority (82%) of respondents were female.

Davidson found that female participants reported a greater need for more continuing education in five of the seven responsibilities compared to their male counterparts. However, few men completed the survey. Additionally, Davidson "found no statistically significant differences between perceived job relevance and sex" (p. 201). I did not ask participants about their perceived continuing education needs. However, I did learn about participants' thoughts on education and training needed for the field.

#### Men in Health Promotion

The NASPA completed *Wellness and Health Promotion Benchmark Survey 2011* in 2011. The survey, completed by health promotion department directors, asked questions about current practices, demographics of the university, and demographics of the individuals answering the questions. The researchers found that 23.5% of respondents were male. Similarly, as stated earlier, the vast majority of the members of the Health Promotion Section of the ACHA were female (A. Tinker, personal communication, November 5, 2012). While these two samples may not encompass all who work in the profession, they suggested that health promotion in higher education is a female-dominated profession.

While I was able to find one study on professional development needs of health promotion professionals in higher education, I was unable to locate any research studies focused specifically on men working as a health promotion professional in higher education. However, I located two studies of male peer health educators. Peer education is educating youth and young



adults by using their peers as the educator (Beshers, 2008). While these two studies did not focus specifically on male health promotion professionals working in higher education, their information is transferrable.

Cupples et al. (2010) wrote a program review on recruiting, training, and evaluating the use of college-age male sexual health, peer health educators. The peer educators did outreach work, conducted educational sessions, and developed marketing and educational materials. The educational sessions were conducted for men ages 13 to 25 at schools and universities. Cupples et al. found that the college-age male peer educators were popular with those they worked with and with the faculty. Additionally, the peer educators connected well with young men. The researchers also found that retention was an issue because the peer educators eventually graduated from college. However, the current peer educators encouraged more men to enter the program to serve as future peer educators (Cupples et al., 2010). The findings of the study indicated that having older males available to work with younger males could be helpful when discussing health issues. Additionally, male involvement could help to recruit more men to the field.

A second study that examined male peer health educators focused on the lack of men involved in sexual health education groups (Beshers, 2008). This study focused on sexual health peer education groups working with community youth rather than the higher education population. Beshers (2008) administered surveys to 40 sexual health peer education programs. The survey included demographics, questions about the program structure, and questions on recruitment. The researcher tested the survey for face validity prior to administration. Beshers (2008) found that the average percentage of male peer educators in the groups studied was 24.9%. The researcher indicated that some barriers for male participation included sexual health



education being perceived as a women's issue, a lack of confidence with communication and relationship-building skills, and an inability to express their male personality characteristics (Beshers, 2008).

The focus of both of these studies was on peer health education, and more specifically, sexual health education. Though these two studies did not focus on health promotion in higher education, they did allude to the fact that there is a lack of involvement among men. While both articles identified barriers to participation for male practitioners, both discussed recruitment efforts centered on men. I asked men about their experiences in college. I learned that some participants were involved in peer health education groups and other student leadership opportunities.

#### Conclusion

Much of the research reviewed focused on a limited number of nontraditional careers for men. Additionally, the majority of research focused on why men enter nontraditional careers and barely cover why men maintain their nontraditional career over time. "Methodologically and statistically rigorous investigations are critically needed to better understand predictors and mediators of nontraditional career choice as well as outcomes of those choices" (Heppner & Heppner, 2009, p. 60). Additionally, while there was literature available on career entry and persistence for student affairs professionals, I was unable to locate information on career entry and persistence in health promotion in higher education.

The studies reviewed in this chapter indicated that the topic of men in nontraditional careers is an important focus area. The information obtained through this research could be important for colleges and universities that prepare students for future employment, for



employers looking to diversify their fields, and for men considering a nontraditional career. I asked participants about their motivations for entering a career in health promotion in higher education and why they remain. I also asked what factors positively and negatively influenced their career entry and persistence. By learning this information, I am now able to share findings with practitioners in the field and faculty in academic preparation programs. These professionals could utilize the findings to recruit and retain male health promotion professionals. More broadly, practitioners in other areas of student affairs who are interested in recruiting and retaining male professionals may be able to utilize the findings of this study. In the following chapter, I discuss the methodology I utilized to conduct this study. Subsequent chapters outline the findings, and I discuss the findings in the context of the literature.

#### **CHAPTER 3**

#### **METHODS**

Previous research found that men connect well with other men on health issues and that older men may positively influence young men (Beshers, 2008; Cupples et al., 2010; Harris & Edwards, 2010). However, men are underrepresented in the field of health promotion in higher education (National Association of Student Personnel Administrators, 2011; A. Ticker, personal communication, November 5, 2012). Based on previous research, having male health promotion professionals could be important for institutions of higher education. Therefore, there is a need for more information on why men choose to enter and remain in health promotion in higher education. Having men on a health promotion staff may well assist with connecting male students on campus with health resources. Institutions of higher education could utilize the data I obtained to better recruit and retain male health promotion professionals.

In the first two chapters, I introduced the purpose of the study, the significance and need for this research, and I provided a review of the related literature. In this chapter, I discuss the rationale for using a basic interpretive qualitative research methodology, the research sample and setting, methods for data collection and analysis, and issues regarding trustworthiness of the data.



# Purpose

The purpose of this basic interpretive qualitative study was to explore the perceptions of men working as health promotion professionals in the higher education setting regarding their entry into and persistence in their chosen profession.

### **Research Questions**

In order to conduct this research and address the stated purpose above, the following research questions guided this study:

- 1. What influenced participants in their decision to enter health promotion in the setting of higher education?
- 2. What continues to influence participants to remain in health promotion in the setting of higher education?
  - 2a. What benefits do men perceive working in health promotion in higher education?
  - 2b. What challenges do men perceive working in health promotion in higher education?

### Methodology

According to Creswell (1998), qualitative research "is an inquiry process of understanding based on a distinct methodological tradition of inquiry that explores a social or human problem" (p. 15). The purpose and research questions developed for this study lend themselves to a qualitative approach, as I sought to explore and understand the perceptions and experiences of men entering and remaining in the field of health promotion in higher education.

I chose a basic interpretive qualitative approach because my research focused on understanding the experiences and perceptions of men currently working as health promotion professionals in higher education. The goal of a basic interpretive qualitative study is to discover and interpret the meaning of experiences of a group of participants (Merriam, 2009). According to Merriam (2009), the purpose of this approach is to better understand "1) how people interpret their experiences, 2) how they construct their worlds, and 3) what meaning they attribute to their experiences" (p. 23). Similarly, Bogdan and Biklen (2007) state that researchers using the basic interpretive qualitative research approach gather data in order to learn how people understand their lives and experiences. In basic interpretive qualitative research studies, the researcher analyzes the data to look for recurring or common themes and the findings reference previous research (Merriam, 2002). I used the literature reviewed in Chapter 2 as context for my findings.

I utilized a basic interpretive qualitative approach and I also employed a general phenomenological perspective. According to Patton (1990), a general phenomenological perspective focuses on what people experience and how they interpret that experience. "One can employ a general phenomenological perspective to elucidate the importance of using methods that capture people's experience of the world without conducting a phenomenological study that focuses on the essence of shared experience" (Patton, 1990, p. 71). I focused on how men interpreted their experiences and perceptions as men working in health promotion in higher education. This was an exploratory study; therefore, I did not focus on shared experiences.

I chose a basic interpretive qualitative research approach because it allowed me to obtain information on the participants' understanding and meaning of their experiences and perceptions while working as a health promotion professional in higher education. This approach allowed



me to explore, in their own voices, how participants entered into their career and why they remained.

#### Research Procedures

### Research Participants

The research participants were five men working as health promotion professionals in higher education. Each was required to have a minimum of 5 years of experience in the field. I did not include men who left this field, retired, or who moved into other positions within higher education. I excluded those who left the field, because they would not have been able to speak about persistence. Additionally, I chose to exclude those who retired because they may not have been able to speak about factors regarding entry and persistence in the same way as men currently employed. All participants worked at public or private 4-year higher education institutions in the United States.

The first selection criterion was that all participants self-identified as male. For the purposes of this study, being a man, or male, was a socially constructed role in a given society rather than a biological sex (World Health Organization, 2013). Therefore, the participants may or may not have been born biologically male. All five participants self-identified as male.

The second selection criterion was that all participants must work at a 4-year institution of higher education in the United States. I chose this criterion because I focused my research on institutions of higher education in the United States. I chose to only include 4-year institutions because the majority (96.9%) of health promotion professionals who completed the *ACHA Health Promotion Section Needs Assessment 2012* (2012a) indicated that they serve at a 4-year



institution. The number of men in this field is not large; therefore, I sampled participants from across the country and at both public and private 4-year higher education institutions.

In order to select participants who had a strong understanding of health promotion in higher education and their persistence in the field, all participants were required to have a minimum of 5 years of professional experience in health promotion in higher education. I chose this because those in the profession for several years should be information-rich cases because they have more experience in the field (Patton, 1987).

I utilized purposeful sampling to recruit and select participants. "Purposeful sampling is based on the assumption that the investigator wants to discover, understand, and gain insight" (Merriam, 2009, p. 77). I chose this method so I could select those information-rich cases from whom I could gain the most understanding and insight (Patton, 1990). I identified eight men who worked as health promotion professionals who I was aware of through reviewing the member directory of the ACHA's Health Promotion Section. I asked the eight men to participate in the study through an initial e-mail that explained the study, the selection criteria for inclusion, and the nature of the questions I would ask (see Appendix A for a copy of the recruitment e-mail). Five men agreed to participate, one refused, and two never responded.

### **Data Collection**

Prior to beginning data collection, participants completed an informed consent form (Appendix B). The informed consent form includes the purpose of the study, the estimated amount of time that data collection would take, as well as how I planned to use the results (Creswell, 1998). As participants worked at institutions around the country, I e-mailed the consent form to them and instructed them to sign it and then either e-mail a scanned copy back



to me or fax it to me. I also instructed participants to keep a copy of the signed consent form, and I kept the signed forms in a locked file separate from the raw data collected.

I conducted semistructured interviews because they allowed for flexibility with the questions and their order while gathering specific data (Merriam, 2009). I wanted to gather specific information from participants about their career path and their experiences and perceptions as men working in health promotion in higher education. I also wanted to have the flexibility to change the order of questions so the conversation could flow smoothly and to ensure I had the ability to ask follow-up questions. Semistructured interviews allowed me to ask various questions of substance based on previous and potentially unexpected responses from participants (Merriam, 2009).

For the purposes of this study, I utilized Seidman's (2006) three-interview series. In this series, the first interview focused on the participants' life history, the second on their experiences directly related to what was being studied, and the third allowed the participants to reflect on the meaning of their perceptions and experiences (Seidman, 2006). Each interview lasted between 45-90 minutes and took place over the course of 3-5 weeks for each participant. I did not give participants a copy of the questions ahead of time.

I chose this approach for several reasons. First, it allowed me to build rapport with the participants. According to Glesne (2006), part of building rapport with participants is spending enough time with them to show that I was interested and invested, nonjudgmental, and that I would keep the promise of confidentiality. Additionally, the three interviews allowed me to obtain depth of information. I had the time needed to "capture the unseen that was, is, will be, or should be; how respondents think or feel about something; and how they explain or account for something" (Glesne, 2006, p. 105). Finally, by conducting three interviews, I gained a holistic



perspective of the issue so that I could understand the factors that influenced each participant's decisions to enter into and remain in the field of health promotion in higher education (Patton, 1990).

I structured the interviews using an interview guide that provided a list of questions and topics to explore with each participant during each interview (Patton, 1987). The guide allowed me to "build a conversation within a particular subject area, to word questions spontaneously, and to establish a conversational style – but with the focus on a particular predetermined topic" (Patton, 1987, p. 111). I developed guides for each interview (Appendices C, D, & E) to give structure to them while allowing for flexibility in how and when I asked the questions.

I interviewed participants located at institutions of higher education across the United States; therefore, I utilized telephone interviews. I used a speakerphone so I could record the interviews using a digital recorder. In order to maintain confidentiality, I conducted the interviews in a private space with a white noise machine running in the background. I also took notes throughout. Recording and making notes allowed me to review both what was said by each participant and any thoughts that I had during the interviews (Merriam, 2009). I kept all audio files in password-protected files on a password-protected computer and will delete all audio files after 3 years following completion of the study. I transcribed all recordings so I could become more familiar with the data and could begin data analysis (Merriam, 2009).

I began the first interview by reiterating the purpose of the research study, and I confirmed my intent to maintain confidentiality of all information collected (Bogdan & Biklen, 2007). For the first interview, "the interviewer's task is to put the participant's experience in context by asking him or her to tell as much as possible about him or herself in light of the topic up to the present time" (Seidman, 2006, p. 17). I focused my questions on influential



experiences during participants' childhood, time in school, and professional experiences. I also asked about their role in their current position. Finally, I asked some demographic questions relevant to the study (Merriam, 2009). The first interview guide is in Appendix C.

The second interview focused on experiences that related to the topic of this study (Seidman, 2006). My questions focused on why participants chose a career in health promotion in higher education and why they remained. I also asked questions regarding their perceptions of the benefits and challenges of being a male in their career. Finally, I asked participants questions about experiences and relationships with students, colleagues, friends, and family. An interview guide for the second interview is in Appendix D.

During the final interview, I asked participants to "reflect on the meaning of their experience" (Seidman, 2006, p. 18). The focus of the final interview was on what we spoke about during the first two interviews, and I asked them about specific experiences or perceptions that they discussed previously. "The combination of exploring the past to clarify the events that led participants to where they are now, and describing in concrete detail of their present experience, establishes conditions for reflecting upon what they are now doing in their lives" (Seidman, 2006, p. 19). I also asked questions about their future career goals in order to determine whether they planned to persist. Additionally, I asked specifically about the influence their gender had on their career. An interview guide for the third interview is in Appendix E.

I gave participants the chance to review the transcripts and my interpretations of the data through member checks (Merriam, 2009). Additionally, when I needed other clarifying information after completing an interview, I contacted the participants via e-mail.



## **Data Analysis**

I utilized a constant comparative approach when analyzing data collected. This method of analysis consisted of "comparing incidents applicable to each category, integrating categories and their properties, delimiting the theory, and writing the theory" (Glaser & Strauss, 1967, p. 105). While the constant comparative method is most often used when conducting grounded theory studies, Merriam (2009) states that it has been utilized when conducting other types of qualitative research. First, I reviewed the interview transcripts to determine what themes and issues reoccurred within each interview. Glaser and Strauss (1967) refer to these themes as incidences. Then, I created categories for the themes and compared across the interviews to determine which categories were consistent and that there was consensus among the participants (Glaser, 1978). Additionally, the approach required me to continue to collect data until saturation or the point at which I discovered no new information (Creswell, 1998).

Because the constant comparative method calls for analysis to begin prior to the completion of data collection, I coded data continuously throughout the process of data collection (Glaser, 1978). I coded all transcripts by hand. I utilized the open, axial, and selective coding method discussed by Corbin and Strauss (2007). First, I utilized open coding to develop categories based on the data collected (Creswell, 1998). I reviewed the data to look for "words, phrases, patterns of behavior, subjects' way of thinking, and events that repeat and stand out" (Bogdan & Biklen, 2007, p. 173). According to Merriam (2009), these are called units. I highlighted the units in the transcripts of each interview by hand.

The next step of data analysis was axial coding. I used the units from the open coding process to develop categories based on the patterns that repeat throughout the data collected (Merriam, 2009). This required me to focus on one category at a time and to bring the units



together into themes (Merriam, 2009). During the last phase of the coding process, I conducted selective coding in which "the researcher identifies the story line and writes the story that integrates the categories in the axial coding model" (Creswell, 1998, p. 57). I utilized the previous research discussed in Chapter 2 when I conducted selective coding. By using this data analysis plan, I gained a better understanding of why participants chose to become health promotion professionals in the setting of higher education and why they remained.

## Conceptual Framework

In the current study, I explored why men enter and remain in a career in health promotion in higher education. I studied men working in this field in regard to their motivation for entry, factors impacting their persistence in their career, and any issues and challenges they encountered when working in a female-dominated profession. I found literature on career entry and persistence on men working in other nontraditional careers and on professionals working in student affairs that I utilized when completing the analysis of this study. I utilized the literature as a starting point rather than focusing only on factors discussed previously.

Within the literature reviewed in Chapter 2, I found many factors that influenced career choice. I utilized factors impacting career choice to frame the findings of this study. Some of these factors included educational interests (Carmichael, 1992; Sargent, 2004; Werhan, 2010), the environment (Chusmir, 1990), perception of the career (Gottfredson, 1996; Williams, 1995), contact with an influential person working in a nontraditional career (Betz & O'Connell, 1992; Blimling, 2002; Calhoun & Taub, 2014; Gillingham, 2006; Hunter, 1992; Rice & Goessling, 2005; Taub & McEwen, 2006; and Werhan, 2010), and having a positive experience as a



student leader while in college (Blimling, 2002; Hunter, 1992; Taub & McEwen, 2006). I discuss these factors in Chapter 5.

My other issue of interest was persistence of men working in health promotion in higher education. The literature I reviewed discussed both factors that assist with and factors that challenge persistence. Factors that have a positive impact on persistence include developing relationships and having a mentor when beginning a career (Betz & O'Connell, 1992; Calhoun & Taub, 2014; Dohner et al., 1990; Galbraith, 1992; Gillingham, 2006; Koberg et al., 1998; Renn & Hodges, 2007), having a high satisfaction and morale (Rosser & Javinar, 2003), involvement in professional organizations (Chernow et al., 2003), connecting with other men in the field (Gillingham, 2006; Koberg et al., 1998), and male privilege (Giesler, 2013; Loschiavo, Miller, & Davis, 2007).

Factors that negatively impacted participants include low morale, low job satisfaction and not feeling valued (Calhoun & Taub, 2014; Hogue et al., 2010; Rosser & Javinar, 2003); lack of understanding of the field (Blimling, 2002); lack of advancement opportunity (Blimling, 2002; Calhoun & Taub, 2014); moving into more administrative roles (Carmichael, 1992; Sargent, 2004; Simpson, 2005; Williams, 1992); balancing masculinity, gender, and the work (Calhoun & Taub, 2014; Cross & Bagilhole, 2002; Giesler, 2013; Sargent, 2004; Simpson, 2005; Werhan, 2010;); and being stereotyped (Calhoun & Taub, 2014; Carmichael, 1992; Sargent, 2004; Simpson, 2005; Werhan, 2010). I discuss these factors in detail in subsequent chapters.

#### **Trustworthiness**

The entire study including recruiting participants, data collection, analysis, and reporting must be trustworthy (Merriam, 2009). Collecting and analyzing data and reporting findings in a transferable, dependable, and credible way assists with the trustworthiness of a study (Lincoln & Guba, 1985; Merriam, 2009). I utilized several strategies to ensure trustworthiness.

I performed member checks with participants in order to establish credibility (Lincoln & Guba, 1985). Credibility is established by ensuring that participants agreed with my interpretation of their experiences. According to Lincoln and Guba (1985) conducting a member check "is the most crucial technique for establishing credibility" (p. 214). Member checks required me to share data collected and the analysis with participants and to ask for their feedback (Creswell, 1998). I needed to establish that my data and analysis were recognizable to my participants. According to Merriam (2009), "participants should be able to recognize their experience in your interpretation or suggest some fine-tuning to better capture their perspective" (p. 217). I followed-up with all participants via e-mail after the interviews to ensure the interviews were transcribed correctly. I shared my analysis on two separate occasions with participants, and I received feedback from two of them. By working with my participants to see that my interpretations of their experiences were correct, the data collected should be credible (Lincoln & Guba, 1985).

I also utilized peer debriefing to enhance the credibility of the findings. Peer debriefing is a process that involves "locating a person (a peer debriefer) who reviews and asks questions about the qualitative study so that the account will resonate with people other than the researcher" (Creswell, 2003, p. 196). According to Lincoln and Guba (1985) peer debriefing helps keep the researcher honest because "biases are probed, meaning explored, [and] the basis

for interpretation clarified" (p. 308). I utilized peer debriefing to discuss and defend my findings and to discover any assumptions I made regarding the findings (Lincoln & Guba, 1985). By completing peer debriefing I was able to ensure that I had not made assumptions regarding participant experience and that I had not missed any themes in my analysis of the data. Lincoln and Guba (1985) recommend choosing a peer debriefer who is neither a supervisor nor supervisee, but is someone who will take the job seriously. Therefore, I asked a health promotion in higher education colleague at another institution to review my interpretations of the findings in order to determine whether the findings resonate with others.

Finally, in order to establish transferability I used thick descriptions of the data as a trustworthiness strategy. According to Lincoln and Guba (1985), using thick descriptions allows for both the researcher and the reader to make judgments about the transferability of the conclusions. Thick descriptions "may transport readers to the setting and give discussion an element of shared experience" (Creswell, 2003, p. 196). Merriam and Simpson (2000) state that thick descriptions include enough detailed information about the participants and findings so that it is obvious to the reader why I made my interpretations. In subsequent chapters, I discuss participants' background, perceptions, and experiences in order to create thick descriptions for the reader; therefore, establishing transferability.

#### **Ethical Considerations**

I utilized the "Ethical Issues Checklist" discussed by Patton (2002) that includes factors such as obtaining informed consent from study participants and maintaining confidentiality, among others. I discussed the purpose of the study with all participants, and all participants signed an informed consent form prior to participating. I kept interview transcripts, as well as



my interview notes, in a locked drawer separate from the signed informed consent forms. I kept all audio and electronic files in password-protected files on a password-protected computer.

The only other people who had access to the raw data were my dissertation committee members. I also gave the participants the opportunity to review their own raw data. Conducting the study in an ethical manner should increase the trustworthiness of study findings (Lincoln & Guba, 1985; Merriam, 2009).

### **Summary and Conclusion**

I utilized a basic interpretive qualitative methodology to explore the experiences and perceptions of men working in health promotion in higher education. I recruited participants using purposeful sampling. The criteria for inclusion included self-identified males currently working as health promotion professionals in 4-year public or private higher education institutions in the United States. All participants had a minimum of 5 years of professional experience as a health promotion professional in higher education.

The participants engaged in semistructured interviews regarding this entry and persistence in their career. I utilized the constant comparative method to analyze the data collected in the interviews. The methods discussed above allowed for the data collected and the findings to be credible and trustworthy. I utilized previously completed research as a starting point for analyzing the data; however, I also looked for new information on career entry and persistence not previously documented in the literature. The findings of the current study provide a better understanding of the experiences and perceptions of men working in health promotion in higher education. In the next chapter, I discuss the data collected, and in the final chapter, I provide conclusions and suggestions for further research and practice.



#### **CHAPTER 4**

#### **FINDINGS**

The purpose of this chapter is to reveal the findings of this basic interpretive qualitative study. I explored the following research questions.

- 1. What influenced participants in their decision to enter health promotion in the setting of higher education?
- 2. What influences participants to remain in health promotion in the setting of higher education?
  - 2a. What benefits do men perceive working in health promotion in higher education?
  - 2b. What challenges do men perceive working in health promotion in higher education?

In this chapter, I briefly describe each of the five study participants. Subsequent sections outline the findings of the two research questions. The purpose of this study was to explore the perceptions of men working as health promotion professionals in the higher education setting regarding their entry into and persistence in their chosen profession.

# **Participants**

For the purposes of this study, I interviewed five men currently employed in health promotion in higher education (see Table 1). All names used are pseudonyms.



Table 1
Participant Demographics

Name	Age	Years in field	Race	Institution type	Position type
David	37	11	Caucasian	Public	Director
Harry	35	14	Caucasian	Private	Director
James	36	11	Black	Public	Director
Mark	42	18	Caucasian	Public	Director
Matt	47	16	Filipino	Private	Health educator

All participants indicated involvement either locally or nationally with health promotion, wellness, or student affairs organizations. I knew Harry, James, and Mark through membership in national health promotion organizations. While I did not know David or Matt prior to this study, I knew their names through their involvement with the ACHA.

Health promotion in higher education is a female-dominated field; therefore, the men in the profession are easy to identify. In order to preserve the confidentiality of the participants, I discuss their education, career progression, and current institution at a minimum. The following description of each participant serves to provide context for the perceptions and experiences discussed later in this chapter.

#### David

I spoke with David during the final weeks of the Fall 2014 semester. As a child, David wanted to be a paleontologist, but as he got older, his interest in science led him to medicine. He worked at a hospital while in college but determined that was not the right fit for him. He learned about public health while exploring other options and "thought it was an intriguing notion to work on the health of populations and not just individuals." After completing a



bachelor's of science degree in biology with a chemistry minor, he explored graduate programs in community health science and, more specifically, health education. He completed a master's degree in public health and participated in the Peace Corps Master's International Program. He served in the Peace Corps for over 2 years in Africa in a health education role. Upon returning to the United States, David did not have a career goal. While speaking to a friend about the possibility of going back to school to pursue a second master's degree, David's friend stated, "it sounds like you want to be in the collegiate environment but not as a student." He came across a "posting for an alcohol, tobacco, and other drug prevention specialist" and he applied, got the job, and began his career in the field.

David worked at four institutions of higher education, both public and private, located within different geographic regions of the United States. Additionally, he earned a second master's degree in higher education leadership. At the time of the interviews, David had recently started a new director-level position at a new institution. He indicated that much of his work consisted of relationship building, attending meetings, supporting and supervising staff, and working with students. In the future, David wanted to continue his work in health promotion in higher education. He planned to earn a Ph.D. degree in public health science and "maybe have a faculty-type role but still maintain my role as a practitioner."

### **Harry**

I completed Harry's interviews at the beginning of the Fall 2014 semester. We began all three interviews with a brief conversation about the activities related to the beginning of the new academic year. Harry seemed excited about participating in the study and indicated interest in hearing the findings.



Harry had worked in health promotion in higher education throughout his professional career. He indicated an interest in education from the time he was a child; however, his interest pertained to the K-12 field. Harry never considered higher education as a career until graduate school. Though he earned a bachelor's degree in education to teach elementary school, his involvement in peer health education and experiences with student affairs influenced his career change. While considering positions in primary education, he also did "a little bit of poking around of entry-level gigs mostly around areas like res life and judicial." He said that he had not considered a position in health promotion in higher education until he worked as a graduate assistant in the health promotion department with peer health educators.

Harry was the director of his department at the time of the interviews. He oversaw two departments, one focused on alcohol and other drug education and the other focused on general health issues and sexual assault prevention. He supervised both professional and student staff and worked collaboratively with many other campus departments. Harry indicated much of his work centered on meetings with other professionals, community organizing, supervision of staff, administrative duties, and working with the peer health education group. He spoke passionately about his work, especially his interactions with the students. While Harry stated no intention of leaving higher education, he mentioned feeling "tension between remaining in health promotion and continuing to advance." Harry discussed going back to school for a terminal degree, possibly in public health, as he had "aspirations at some point in time of potentially being a dean of students or a chief student affairs officer."

#### James

I also interviewed James during the first part of the Fall 2014 semester. James spoke about his intention to look for opportunities outside of health promotion in higher education during our first conversation. However, he also talked positively about his work. James mentioned an interest in learning more about men in this profession based on the findings of this study.

James originally "flirted with the idea of going to med school"; however, while in college, he determined "what clinicians did wasn't exactly what I was looking for." His experiences while growing up and involvement as a student leader on campus helped him recognize his "passion to create change and help improve the lives of other people" which led him to public health. He worked as a health educator in the topic of sexual health at a community organization while he earned his master's degree in public health. After completing his degree, he chose to leave his position because he wanted a new challenge. He accepted a position as a health educator at his alma mater. He stated that "the idea of working with college students and different issues just seemed appealing."

During the course of his career, James worked at two institutions. At the time of the interviews, he was the director of his department. In his role, James supervised multiple professional staff members, collaborated with other departments, worked with regional and national college health organizations, and served on many committees on campus. His department focused on alcohol and other drugs, sexual health, stress management, nutrition, and sexual assault prevention.

James indicated that in the future, he did not see himself "in an environment where I'm educating college-age students." He spoke about his interest in positions in organizations



outside of higher education, but he "wouldn't go as far as saying that I won't be in public health." In higher education, he felt he had "reached a ceiling...in career growth specifically" and that he needed to feel like he was "professionally growing and maturing in sophistication of work"

#### Mark

I spoke with Mark during the middle part of the Fall 2014 semester. He indicated excitement to participate in the study. Originally, Mark explored careers in cardiology and the military; however, due to health issues the military was not a feasible option. During high school, Mark attended a substance-abuse prevention camp, though initially, he "hated being there." He stated that he gave it a chance and grew to love the work. Mark then began involvement in both the camp and its "sister organization" as teen staff. These experiences fostered his interest in substance-abuse prevention. After earning his bachelor's degree in psychology and communication studies, Mark worked at a substance-abuse prevention organization for a year and a half before returning to school for his master's degree in clinical psychology. During this time, Mark worked at the health promotion office on campus as a graduate assistant. That position eventually became a full-time, and he had been there ever since. He completed a Ph.D. degree in public health and was the only participant with a terminal degree.

At the time of the interviews, Mark supervised three full-time staff members and multiple graduate students. His job consisted of running the department and a state-wide alcohol and drug prevention grant. In his role, he spent most of his time supervising staff and students, "putting out fires," handling administrative duties, collaborating with professionals on campus



and across the state, and assisting with assessment. Additionally, Mark taught classes at the college.

Mark indicated uncertainty regarding his future goals. "My career aspiration was to kind of be a director of a prevention unit...I achieved that rather early in my career." He expressed thoughts of pursuing a career in teaching; however, he indicated his dream job was "to have a dual appointment with a college health service and an academic health education/health promotion program or a college student affairs program."

#### Matt

I interviewed Matt near the end of the Fall 2014 semester. Like Mark, Matt spent his entire career at the same institution. Matt said that he loved his work with students and stated no interest in moving into a more administrative director role. He considered himself a "health and happiness educator."

Matt's interest in career always centered on doing work that focused on self-improvement. Experiencing "dorm-based programming" in college as well as the "gurus of self-help in the 70's, 80's, and even 90's" drew him into that type of work, especially work regarding the topics of sex and sexuality. He earned a bachelor's degree in psychology and a master's degree in sociology. Upon graduation, he entered a position at the county health department and then a position at a community organization; he did AIDS prevention programming in both positions. Mark's presentations and education at the university while working at the community organization came to the attention of the university's director of the health promotion department. She approached him when a health educator position became available.



Matt's worked in 80% part-time position. He was the only participant not employed full-time. His position consisted primarily of doing educational presentations, meeting one-on-one with students, and meeting with small groups. He also did some educational work in the surrounding community at schools and other community organizations. Matt stated that he planned to continue doing this work as long as he can. He had no interest in moving up in the organization and was adamant that his place was educating, not in meetings. He expressed a keen interest in learning more wellness topic areas and increasing his skills with blogging and writing.

# Factors of Entry Into Health Promotion in Higher Education

I first examined factors that influenced each participant in their decision to enter health promotion in the setting of higher education. During the course of data analysis, I found several factors regarding participants' decision to enter a career in health promotion in higher education. Six major themes emerged from the data including (a) an attraction to or interest in teaching, helping others, and health; (b) being one of very few men; (c) it is more than just a public health or student affairs degree; (d) influence from others; (e) experiences growing up; and (f) experience as a student leader (see Table 2). This section provides evidence regarding each of the themes.

### Attraction to or Interest in Teaching, Helping Others, or Health

All five participants described an attraction to or interest in teaching, helping others, or health. Matt stated that "You know, I grew up in the 70's and 80's during the self-help



Table 2
Factors of Career Entry

Theme	Subthemes			
1. An attraction to or interest in teaching, helping others, and health				
2. Being one of very few men	2a. A lot of women in graduate school 2b. Female-dominated			
3. It is more than just a public health or student affairs degree	<ul><li>3a. What is college health?</li><li>3b. Coursework did not prepare me for this</li></ul>			
4. Influence from others				
5. Experiences growing up				
6. Experience as a student leader				

movement and was really affected by that. So I think, looking back, I was very attracted to self-improvement." While Harry initially wanted to work in K-12 education, he stated, "I've always had a tendency toward education and towards teaching...It felt like a very natural place for me and a very good fit. Coupled with the fact that I just had this interest in health and wellness issues." Similarly, Mark did not initially intend to work with college students; however, he became interested in health education at an early age:

Really when things became serious, and then became real, was probably when I was a sophomore and had gone to a week-long substance-abuse prevention camp the prior summer; I was going back the second summer as teen staff. That's really, kind of, really where I decided I wanted to do something with a helping profession, social services where I'm, "Ok, I want to do prevention, health education, health promotion for a living"... So, I got involved in that and then I went back the next summer, had the time of my life and, just kept staffing that camp and staffing the community program through high school and through college. And really decided that substance-abuse prevention was what I wanted to do.



David and James both planned to enter medical school but realized early on that it may not be the right fit.

I always had a desire to go to medical school and become a doctor. I went into premedicine when I started college, and at the end of my freshman year in high school I
started working at the local hospital. I just became disillusioned very early on with the
medical system. So I started, finally, for the first time, thinking, "Hmm, maybe medicine
isn't the way to go," even though everyone was definitely pushing me in that direction
when I expressed an interest in it. So I started exploring other fields casually in the
healthcare discipline, and I had never really come across public health, but I found it and
thought it was pretty cool. But I just really fell in love with public health, so I started
exploring graduate programs and the different disciplines within public health as a field
and really settled on the community health science, health education component of it.
(David)

That was kind of my original thought, being a physician. Then, you know as things progressed in my academic career, I was realizing, I guess, that I was experiencing, you know, a struggle. And, with medical professionals, what clinicians did wasn't exactly what I was looking for...Once I made my decision to switch majors and kind of found health and science education, which was my undergraduate [major], you know, all of the classes were really focused on giving students a broad perspective on very topically centered things like alcohol education and drug education. (James)

While none of the participants planned to enter a career in health promotion in higher education, all described career interests that eventually led them to the field. This factor did not connect directly with participants' gender; however, literature on men entering nontraditional career did discuss this factor. I discuss this in Chapter 5.

#### One of Very Few Men

All participants discussed the female-dominant nature of health promotion in higher education and being one of very few men. Additionally, several participants found a similar gender make-up in their graduate program.



### Female-Dominated

In regard working in health promotion in higher education three participants said the following:

I am often the only man in the room. But I think when you find yourself among your peers, and there are so many groups in which this is the case, I could very well be sitting in a meeting with 25-30 people and be the sole male-identified voice there. And I think with that brings its own set of challenges and opportunities. I don't mind it...it's taken a little while, I'd say professionally, to get accustomed to it because it does, it tends to be a female-dominated profession. (Harry)

It's definitely something that I tend to notice every so often, or I may think, like, wow, I work with a lot of women. It's even more pronounced at annual meetings and ACHA where I'm one of very few men. It's one thing when I'm with my four other direct reports, but when it's me and, like, 40 other people in a session, and I'm, like, one of maybe 5 males, yeah, you can, you can kind of see that distinction. And that's where you're like, "Wow, there really are a lot of women in this profession." (James)

The field tends to be very female-dominated. The majority of folks that I tend to work with, seem to work with, are female. Very few males...You know a lot of times you don't think of a male Caucasian as being a minority, but in health education, I really do kind of feel that way. (Mark)

All of my participants perceived the field as being female-dominated, especially when attending meetings with other professionals. However, none of the participants indicated that this perception discouraged them from entering their career.

### A Lot of Women in Graduate School

Several participants discussed the female-dominated environment while in college.

Harry said that "As an education and human development major, I was often the only guy in my academic courses as well." David and James had similar experiences in graduate school:

I was really surprised when I got to graduate school, and it was me and maybe one or two other guys in my department, which was community health science. There were a lot of men in the epidemiology programs. They were represented in biostatistics. They



were represented in tropical medicine. In the softer or more social science disciplines, aspects of public health, more of the helping aspect, less of the empirical, more of the helping or practically based, I was really surprised to see there weren't more guys. (David)

Maybe grad school was where I noticed a little bit, more of a, there were more women in the program than there had been in the bachelor's level. So I didn't walk in knowing that, right? It became something that over time I realized, "wow, there really are a lot of women in public health." And, in fact, it's more accurate to say there aren't many men in public health. (James)

None of the participants stated that the female-dominated nature of their degree program or the field influenced their decision to enter; however, it could impact men considering the field of health promotion in higher education. I discuss this in Chapter 5.

### It is More Than Just a Public Health or Student Affairs Degree

Participants described two distinct challenges to entering this career related to their degree programs. These included not learning about it as a potential career and completing degree programs that did not cover all the knowledge and skills needed to enter.

#### What is College Health?

Participants indicated that preparing for a career in health promotion in higher education may be difficult because people may not be aware of it as a career option. Two said,

I had no idea that [college health] was even a thing. In public health school, I knew there was school health, but everything that we ever talked about, or case studies, or peer reviewed articles that I've read tended to focus more in the K-12 system. I don't think once in graduate school I came across an article about college health, or nothing that I remember coming across, at least. Or if I did, it didn't make an impression. (David)

I guess, through the process of college and learning in different classes and organizations I mentioned, I kind of really just stumbled across health education per se, and then more specifically, the field of public health. I didn't have that vision of, like, being a professional in health education or higher education, by any means. (James)



The majority of participants indicated that they did not learn about health promotion in higher education during graduate school. However, all participants entered the field more than 10 years ago; therefore, I do not know if there still is a lack of awareness of the field.

# Coursework Did Not Prepare Me for This

In addition to not knowing about the field, several participants mentioned that a degree in health promotion or higher education student affairs may not cover the skills and knowledge needed for a career in this field.

I started to see the work I do in higher education as two sides of the same coin. One isn't necessarily more important than the other. On one side, you know, I'm a public health guy. I use that stuff on a day-to-day basis. The scope of my work is public health, health education. However, I also see myself as a student affairs person. So anytime we address issues, and I encourage my staff to do the same thing, we try to look at problems from two different lenses. Number 1, the determinants of behavior and the underlying theory, we look at all the public health evidence-informed, theory-based practice. But we also look at our population, and we look at it from a student affairs developmental theory perspective as well. So we also take a look at that when we are designing programs as well and what is developmentally appropriate for, you know, the particular group that we're dealing with, coupled with the public health theories as well. (David)

The higher ed admin programs are great but you've got to sort of carve out your own nitch and your own path. It's still important to learn the developmental theories and higher ed law and public policy, and the HR stuff. You know, all of that stuff is great if you do eventually want to be a dean or a manager, and those sort of skills have been invaluable for me. But none of my coursework prepared me for this. I mean, maybe my undergraduate, better than my graduate degree, to be honest. You know, my psych background and education background that I think are sort of strong transferrable skills to what I actually do day-to-day. Curriculum design, assessment, and, you know, things like that. (Harry)

I think so much of health has to do with other human beings, not just us, what other people are doing. If other people are wearing bicycle helmets, then my students would wear their bicycle helmets. You know? And so I really think that sort of a sociology, anthropology background or a feminist studies background that deconstructs culture would be very helpful because there is so much in the culture. So those are the fields that I would put in there. Human biology, I guess, too. (Matt)



The participants did not indicate that these factors negatively impacted their career entry.

However, they suggested that there is a lack of knowledge about the field and the need for more training. These factors did not directly relate to the gender of the participants; however, they all discussed how training and education outside of public health and student affairs may be needed for those individuals entering the field.

### **Influence From Others**

Three participants discussed how others influenced the direction of their career. Both David's family and educators at the school he attended while growing up encouraged his interest in a medical profession:

And when I was in high school, I went to a private Catholic high school 'cause the schools in my area, the public schools I should say, weren't terribly good, so, anyways, I went to a private school and, you know, they were all about the alumni and creating successful graduates and stuff. I mentioned medicine, and I never really got any sort of, you know, "have you considered anything else?" They just thought that would be a great profession to pump graduates from the high school into.

#### Harry spoke about family:

My grandmother was a long-time educator and a principal, and we lived with my grandparents for quite a while, and I definitely admired her as a role model. And my mother was, and still is, an elementary school teacher as well, so I feel like I have it in my blood, and I was always interested in pursuing that career path, and eventually you know ended [up] there in a little of an unusual way, I suppose, but still all related, I'd say.

Mark described an influential relationship that developed through his involvement in the community prevention organization:

When I first met him, he was a prevention coordinator, kind of helping run the chapter that I got connected to. I just did a lot of work with him in planning events, alcohol-free alternatives, and social events. So I probably got to interact with him at least on a weekly basis, if not more. So in addition to the prevention, he was an Eagle Scout, so we could



also talk a lot about scouting, different things like that. And to this day, I seek his guidance and counsel for different issues, different professional matters. He was just a good role model for me...Probably because of him I ended up going into psychology and speech comm as undergrads and really was open to substance-abuse prevention as a career. His background was more psychology/social work, not very much health promotion/health education, so I think that's really why I kind of went that direction regarding discipline rather than just going directly into health promotion, health education.

For these participants, the influence of others encouraged their decision to pursue a degree program that eventually led them to health promotion in higher education. Though this factor did not relate to the gender of the participants, it did have an impact of their career entry.

Additionally, none of the participants mentioned the gender of their influencers.

# **Experiences Growing Up**

Several participants described experiences while growing up that influenced their overall career goals. Family dynamics impacted Harry and Mark:

Personally, one of the reasons I decided to focus so much of my career specializing around substance-abuse prevention is having grown up in an alcoholic home. And always sort of having that as a source of influence in my life. And, you know, it is such a huge determiner in terms of my traits and characteristics, who I am and how I operate. You know, if there is a way in which I might be able to help support a student who's gone through, you know, a similar sort of experience or potentially help prevent a student from heading down that path themselves, I think that has always been something that is sort of an additional inspiration for me to keep fighting the good fight. (Harry)

I think the other reason that I'm having some difficulty talking about this is that part of the pull for me to go into substance-abuse prevention, which led to the health promotion in higher education, is that I'm an adult child of an alcoholic. So I think really trying to deal with that and finding a home in prevention were really strong factors that led me into this career path. (Mark)

James spoke about his experience of moving to a new community and how the environment exposed him to health-related issues he had not seen previously:



I guess I stared certain health issues in the face without realizing it, you know? I'm seeing, literally, I'm in 7th grade, and there's a girl pregnant next to me. I'm seeing, I'm hearing about gun violence, I'm hearing about things that are just effecting the community on an everyday basis, not just on a regular, but on an everyday basis. So I think those are the types of things that, I guess, helped trigger my or gave me the opportunity to then demonstrate that I had a desire to be, to effect change in a social capacity. To really, like, it doesn't have to be like this. So a lot of that certainly is rooted in racism and discrimination and things like that I kind of experienced, and it was more in my face at that time of life.

While different aspects of their environment growing up impacted the participants, the experiences influenced their ultimate career direction. Though this factor did not relate to the participants' gender, each described how their environment impacted their decision to enter health promotion in higher education.

### Experience as a Student Leader

Three participants discussed student leadership experience. Harry's leadership experience began in middle school and continued through college:

My middle school had a really neat peer leadership class, and a conflict resolution program that I joined which first really introduced me to that skill set, the type of work that peers could do to support one another. I was able to continue that work through high school with an actual for-credit course on peer leadership, which was amazing. Much of that focused on sexual health and HIV/AIDS at the time. This sparked my interest in health promotion, and as soon as I got to my undergraduate institution, I immediately sought out the peer education program on campus. There, it was more of an umbrella organization covering a whole cadre of issues from alcohol to sexual assault, body image and eating, and it was a pretty all-inclusive organization and a good training ground for me. So while I was studying to be a school teacher at the time, it, at some point within my late undergraduate years, it became apparent that you can actually do student affairs health promotion work on a college campus professionally and get paid for it! It was a pretty wild moment of recognition for me to realize that I could kind of combine my skills and passions that I had in terms of being an educator and possibly work with the big kids instead of the little kids.



Similarly, Mark's experience as a student leader during high school influenced his career trajectory. Additionally, Mark's involvement in health promotion in higher education began while working as a graduate assistant in a health promotion office:

Even when taking the graduate assistantship, I really didn't think of it as a long-term commitment. It wasn't really until the second year of my program where, you know, I was enjoying what I was doing at my assistantship, but when I was at my internship, I was just really kind of miserable. I think the other aspect of the clinical internship that really dissuaded me from counseling and psychology to health education and health promotion was that I found counseling to be monotonous. So I found that a lot of times, my clients would come in and it was just a broken record. Every session would sound very similar to the session before it. I just wasn't helping facilitate any headway with my clients, but I felt back at my assistantship working with students, I was getting students to have epiphanies, that I was helping guide them to greater awareness and greater behavior change. It was just more exciting; it seemed to be a better fit.

Unlike Harry and Mark, James was not involved in a health-related group. However, a different involvement influenced his career trajectory:

I was getting more involved with organizations on campus, and just being more of a student leader. And you know, I think that kind of pointed me towards a passion to just be part of organizations that create change. I just enjoyed that process of things. I think I always had a knowledge that I wanted to help improve the lives of other people, you know, in a very broad sense.

While they had different types of leadership roles, experience as a student leader influenced career choice for these participants.

In the following chapter, I discuss each of these factors of career entry in more detail.

Additionally, I discuss them within the context of the literature and provide details on implications for practice and future research. While not all of these factors related directly to the participants' gender, the literature on men entering nontraditional careers indicates that many of the factors are consistent with previous research.



# Factors of Persistence in Health Promotion in Higher Education

I also explored factors impacting career persistence. Some factors related directly to the participants being men, while others seemed more generalizable to women as well. I first describe the factors that participants perceived as benefits and then those that participants perceived as challenges (see Table 3). Finally, I describe findings regarding how some participants did not consider their gender in terms of their work prior to this study.

Table 3
Factors of Persistence

Туре	Theme	Subthemes
Perceived benefit	1. Fulfilling and enjoyable	Subtricties
r crecived ochem	2. Mentorship	<ul><li>2a. Students and newer professionals</li><li>2b. Participants' mentors</li></ul>
	3. Professional relationships	•
	4. Support from friends and family	
	5. Privilege that comes from being a	
	man	
Perceived challenge	1. Feeling unsupported	
· ·	2. Salary	
	3. What do you actually do	
	4. Career aspirations	
	5. Moving up means more administration	
	6. Balancing being a man and a lack	6a. Gender and the work
	of men in the profession	6b. A "different kind of man" 6c. More men
	7. Sexual identity	



### Perceived Benefits

During the course of the interviews, I found several factors that benefitted the participants. The five major themes included (a) participants finding the work fulfilling and enjoyable, (b) mentorship, (c) professional relationships, (d) support from friends and family, and (e) the privilege that comes from being a man.

# Fulfilling and Enjoyable

All participants shared aspects of their work that they found to be fulfilling and enjoyable. They stated that the challenge and diversity of the work kept them interested in the field. Three spoke of how the work made them feel:

I think the most positive would be just seeing projects come to fruition...I'd say I really enjoy that part of my job is, you know, seeing the things that can be challenging at times certainly, but to actually get to a point where, you know, you're accomplishing major milestones with your projects or you're getting the burst of creativity and it kind of manifests within the team in some of the work that we do. So I find that rewarding when we've achieved that sense of accomplishment in any given project or task. (James)

I am happiest, I feel most alive, when I am presenting, when I am training, when I am teaching, and I know I'm in the zone and I know that the audience or that the class or whatever group that I am delivering whatever I'm doing to, is there with me on every word. That is when, I just, that is, like you know, "Ok this is what I was meant to do. This is where I feel most alive." (Mark)

My job is fulfilling every single day. And every single day, I'm really able to get to the heart of where students are suffering. I mean, we are talking in an hour, an hour and a half, not to solve the problem, obviously, because that's not possible, but to really have them go, "Wow, I really clearly see that what I'm doing is painful but it's also normal. And then they can start to go about their life going wow, this is part of the human experience." And then trying to get help to start to move forward. So, very fulfilling. (Matt)

Participants discussed specific parts of their job that they found particularly rewarding and motivating including their interactions with students, developing new initiatives, working



through challenges, and mentoring students and newer professionals. While this factor did not relate to participants' gender, all five discussed how enjoying aspects of their jobs assisted with their persistence.

## Mentorship

Participants discussed mentorship as an important aspect of their work. Participants discussed two distinct areas of mentorship including mentoring students and newer professionals and their relationship with their mentors.

Students and Newer Professionals. In terms of students, four participants spoke about the positive impact of working with students. Harry and Mark discussed working with peer educators. Mark stated, "I really liked the peer education aspect of working with groups, program planning, program design; you know, developing goals and objectives and figuring out how you can meet those things. I also liked the mentoring aspect." David, Harry, and Matt all spoke about interacting with the general student population. Harry described one instance in which he connected with a student leader during new student orientation:

We got to chatting, small talk. I could tell something else was on his mind and he said, "you know, you also met with me my freshman year, and I don't know if you remember the details, but I was a mess and had made some really poor decisions and was drinking in a really dangerous way, and we met, and what you were able to share and had me think about really turned things around for me, and I've never had the opportunity to tell you that." And now as a junior or a senior, he said, "I'm in a totally different place, you encouraged me to connect with some leadership organizations on campus" which is what he has done and all that...It was just a sweet, really, really sweet and very, very rare type of moment...You know, rarely, if ever, do you hear what the outcomes are, and to have had an actual live conversation with someone who felt compelled, for whatever reason, to share more about their experience and how whatever I may have shared in that moment, who knows what it was, you know, made that difference and helped that one person get back on a better path. I mean, it was a pretty special thing for me.



In addition to mentoring students, participants discussed the importance of mentoring newer health promotion professionals. David stated, "Despite being a director, I have an open door policy, and, you know, if my door is open, it's an invitation. So I really pride myself on being accessible to my staff, student staff, and just students in general." Similarly, Mark said,

I think I've always interacted with younger professionals, but I kind of feel like younger professionals are seeking me out for wisdom and guidance. I can kind of be their Obi-Wan Kenobi, and I get a lot of satisfaction and reward working with younger professionals and seeing them grow and develop. They're the next generation to come, and, you know, part of my mentality is my legacy will be what they do and that's important. One [mentor relationship] that comes to my mind is basically the relationship that I have with my current, with one of my current assistant directors. She was one of our grad assistants, and now, you know, she has been my assistant director for 2 years. So just seeing her grow and develop. You know, it's kind of funny. She is beginning to have a lot of "Mark-isms." You know, we'll be in meetings, and something will happen and then I don't have to say what I want to say because she's saying it. Or, you know, we're to the point where we don't have to have a lot of those developmental conversations because she's bringing up those points before I need to, so that's awesome.

None if the participants talked about mentoring male students and professionals.

However, one participant did mention that a lack of men in the profession may make it more difficult for him to mentor men.

<u>Participants' Mentors.</u> Participants discussed the positive impact of their mentors. Harry talked about professional support he received when starting in his first position;

I got a lot of support from family and a lot of support from the dean whom I was interacting with at the time... I felt pretty lucky as well. I certainly worked very hard to earn what I've been able to achieve, but, you know, you know, a lot of time it's good timing and a lot of luck and someone who is willing to take a chance on you. You know, my supervisor at the time, I'm forever grateful to him for taking a chance on me and just welcoming me with open arms. It was just a great learning experience, the couple of years that I spent there.

Mark stated that his mentor from his high school prevention program remained an important mentor throughout his professional career. Participants discussed mentors in student affairs and



outside of higher education; however, I found that no participants had had a mentor in health promotion in higher education. I discuss this in the next chapter.

Three participants discussed gender as it related to their mentors. Harry stated that all of his mentors had been male. Similarly, James stated, "I have a few mentors and people that I consider trusted colleagues on campus that are male and even in my professional network that have nothing to do with the institution." Unlike Harry and James, Matt said that "My supervisors have almost always been women. One of my supervisors at the county was a gay man...Again, different energy, right, then maybe a heterosexual guy or, you know, different feelings, different awareness." While participants discussed male mentors, none indicated gender as a requirement for mentorship.

# <u>Professional Relationships</u>

Participants discussed connecting with other professionals. They spoke specifically about connecting with colleagues in health promotion in higher education. David discussed making professional connections:

I've always been very involved within my regional affiliates of the American College Health Association. So you know, you go to the regional conference or even the national conferences, and I'm certainly not the hotshot in the field of college health by any stretch, but I do know a lot of people. I'm very well connected. Former colleagues or associates within the field have always, you know, I occasionally get e-mails from them, a personal note, or just, "Hey, I know you did this kind of work in the past, and I'd really like to get your opinion on it." Those sorts of things. So I've cultivated good professional relationships.

Mark spoke about how his connections in the field assisted him professionally and personally:

I think one of the reasons I continue doing what I do is because of the friendships and professional relationships that I have with many others. Even though they are people that I only get to see a few times a year...You know, the people I think that I really regard as my true friends are other health educators and other health promotion people, which



makes staying in the field valuable and going to different conferences and trainings worthwhile. And if I left health promotion, not that those friendships would end or cease, but there'd just be less opportunity to see those folks and to run into them and, you know, co-present with them and hang out at conferences and different things like that, so I think a lot of the connections are important... I mean, they're great professionals, they're great individuals, and they're great friends. I just can't hold these people up high enough, they just mean so much, and that helps [me] remain in the field, it helps wanting to stay in the field, it kind of helps keep me on my A-game.

While participants mentioned that connecting with other professionals was positive, they also noted that conferences and meeting with other professionals was when it was most obvious that they are the gender minority in the field. Therefore, in terms of gender, this factor can be both a benefit and a challenge.

# **Support From Friends and Family**

In addition to support received from mentors, participants also discussed support from friends and family. Participants talked about their spouses, other family members, and their friends outside of the field. James shared that while his friends may not completely understand his job, he can make connections with them about aspects of his work. "A lot of my friends are attorneys, so when I talk about policy development, that makes sense to them. Some of my friends are in business, and [when] I'm talking about marketing and communication, we can speak the same language then." Mark spoke about receiving support from friends and family in his life:

I mean, most of my friends, most of my family, particularly those who have known me since high school and, you know, were doing prevention with me in high school and prevention with me in college so, you know, it's the fit...So I think those people that know me affirm that it is a good fit. People have told me that they can't see me doing anything else. Good friends have told me that, you know, that they couldn't see me doing anything else and if I tried to do something else, they didn't think I would be happy. They think I would be miserable. So, I mean, most of my friends in my inner circle really think it's a good fit.



Overall, participants indicated that friends and family supported the career and their work. However, as indicated by James, they did not always understand the career. I discuss this further in the challenges section. Though this factor does not directly relate to participants' gender, it does contrast some of the literature on men in nontraditional fields. I discuss this in Chapter 5.

# The Privilege That Comes From Being a Man

All participants discussed their perceived privilege as men. David perceived that he may benefit during interactions with senior administrators:

You know, I was talking about how I'm very aware of being, you know, what privilege comes with being White, being male, being straight-identified. I understand being a straight, White male in today's society and in today's professional work environment has advantages. It really bothers, me but at the same time, I'll be perfectly frank, it's kind of nice. So with college health, the health center, those communities not-withstanding, a lot of my interactions or, especially as I become more of an administrator and as I've risen in position in higher education, a lot of my interactions, quite frankly, are with, you know, other directors, vice presidents, or vice chancellors, you know, depending on the school and beyond. A lot of those senior positions are still occupied by men, and, for better or worse, it confers some advantages. I don't think that you know if I walked in as a woman, there would necessarily be, like, a different outcome to a meeting or anything like that. I think it's a lot more subtle than that, but I definitely sense that.

#### Harry and Mark described the following:

I suspect there are parts of my presentation and my being, just the way in which I am out there as a White, male, heterosexual administrator that I'm not even aware of, as self-aware as I try to be in terms of how I am perceived or treated, perhaps as compared to my female counterparts and/or staff members who are primarily female as well...but I do imagine that there are some, there are some differences just based on the privilege that I have. And, you know, I really just try to be mindful of that. Mindful of that space as best I can and contribute what I think I can to the conversation and really just be mindful of, I suppose, of any power dynamics or, you know, things that just might feel a little bit different in a space like that. (Harry)

I'm sure I have benefitted as a result of being a male. Have there been times when I've intentionally stopped and gone "Oh I've benefited because I'm a male?" Um, I really



don't recall ever having that mental conversation with myself. You know, I've got some colleagues who are, like, you probably make more money than me because you're a male. I mean, I've had those conversations before. Knowing my campus culture and campus history, it wouldn't surprise me if I'm making more than some of my colleagues who are female because I am male. (Mark)

Matt said, "I mean, I don't doubt that I benefit from some kind of male privilege." However, participants struggled to describe specific instances of that occurring.

I found that participants believed that they benefitted simply from being men. However, they could not always indicate ways in which they had benefitted. I found that some participants felt that the field was not valued as highly as other areas of student affairs and that felt that other factors of their being may mitigate male privilege. I discuss this in the next chapter.

### Perceived Challenges

The participants described a number of factors that challenged them in their role as health promotion professionals in higher education. The seven major themes included (a) feeling unsupported, (b) salary, (c) others not understanding what they do, (d) career aspirations, (e) moving up into more administrative roles, (f) balancing being a man and a lack of men in the profession, and (g) sexual identity.

## Feeling Unsupported

All participants described times in which they did not feel supported by their institution, department, or supervisor. David felt that professionals in health promotion are viewed differently than other professionals working in student affairs:

I don't think, as a field of health promotion or even college health in general within higher education, it has as much clout as some other fields in student affairs in senior level positions. So it's actually a great illustration of my point because I'm a director



here but I don't feel like I'm a real director. So my boss, the other day, said something like, "I'm sorry David, I need to cut our meeting short by about 5 minutes because I've got to get across campus for a department heads meeting." And I remember thinking, "Oh, I'm a department head, too. Why am I not at the meeting, you know?" And it's kind of the concept that we're not really a real department. We will never have our own building here or probably not for the foreseeable future. We're probably always going to be housed either within the health center building or within the counseling building or within someone else's departmental building. When I worked at my second institution, I had a director title, but I wasn't...me and a couple of other positions at the university who had director titles weren't real directors. We weren't the res life director, the director of judicial affairs or student conduct, we weren't the director of student activities, we weren't those traditional old-school, hard-core student affairs departments. So partly ranting a little bit, and it's a bit of a frustration. So maybe if we could create health promotion offices that are their own entities and that have a comparable level of influence and prestige as some of these other student affairs departments within the university structure, that might entice people to stay.

Mark discussed a challenge of health promotion in higher education:

In higher education, the goal is to graduate students, to make responsible citizens, to have them be academically successful, to have them develop leadership skills, to have them develop critical thinking skills, to be good writers, and the main goals of the institution are not necessarily to promote health and wellbeing. Some do, but many do not hold that as their key priority, so in higher education we have to be able to connect what it is that we do, how what we do helps contribute to those goals. Sometimes that's a lot more difficult than what it sounds.

Matt stated that "We're a very small department. We keep getting smaller every day. Prevention, again, is just not valued."

I asked specifically about challenges related to working in health promotion in higher education. Two participants described specific experiences. David described his experience of working toward a campus-wide tobacco ban:

Honestly, I thought I was going to get fired by the Board of Regents [for pursuing the university tobacco ban]. I was starting to get personal invitations [to attend board meetings] from this particular board member, you know, around this issue, and I was very fearful. In fact, I started looking for other jobs in part because I thought it was a very real possibility that I would lose my job over this...So that was probably the most stressful year of my career, was that whole process. As they say, it was the good fight, and that was the fight I was willing to have. I really thought I was going to lose my job.



Harry discussed an instance that occurred during a presentation to faculty living in residence halls. A faculty member asked,

"All of this is fine and good, but at the end of the day, what are your two goals? If you boil this down to just two things, what is it that, you know, you're trying to accomplish, and what's your point?" So, I said, "Well, you know this is a period of time mentally where a lot of our students are starting to create life-long health habits for better or for worse. And, so, for many of them, some of those patterns aren't so healthy and can continue post-college. So if there is a way in which we might be able to intervene at a critical point in time and prevent, for instance, the onset of disease, alcoholism, you know, that would be one of my goals. The other one is, I don't want to lose someone on my watch if I have responsibility for this. With all of these great resources available, I don't want to have a student die of an overdose for fear of getting help." And his very candid response right there in the moment was, "Well, neither of those are going to be achievable at a place like this, and, you know, you should probably look for employment elsewhere because we will have a death, and there's no way that you'd be able to effectively intervene in the trajectory of someone who's going to drink to the point of that." So that was it, he just completely shut down the conversation.

Participants did not suggest intent to leave the profession because of a lack of support; however, they found it challenging. While this factor did not relate directly with participants' gender, all indicated that feeling unsupported was a potential challenge to persistence for professionals in the field.

#### Salary

Four participants discussed salary as a challenge. Additionally, several felt that for men, salary was especially important. Two participants said,

This sounds shallow to say; it's not the most lucrative field in the world. I suppose you can make a good salary depending on what you're doing or how you're doing it. I'm certainly, I'm not complaining about my financial stability, but I really struggled financially as a new professional for the first few years I was getting that initial entry-level experience. That was, that was a real challenge. I know people that, for example, just out of college with bachelor's degrees who are making almost twice as much as I was after 2 years in the Peace Corps and a public health degree, you know, and 2 or 3 years working as a public health professional in college health. So that was a bit of a



frustration. I guess I could have done my homework ahead of time, but that was a bit of a surprise. (David)

I don't know if my salary is, I'm assuming it's pretty, you know, competitive with other health educators. I'm not complaining. But having said that, for someone who lives in this area, I don't make a lot of money. I don't complain because I feel like, you know, for whatever reason, our family situation is such, it's enough. Like I said, you know, on the financial end of things, as a male or a female in this culture, you know, it might not be a bad idea to, um, think about having your own business on the side. (Matt)

One participant discussed low salary as a reason he considered leaving the profession.

Similarly, participants discussed salary in terms of gender. Three participants indicated that salary may be especially impactful for men. Mark said that

of those males that we've had as graduate assistants, none of them have gone into health education. They've gone into health administration. It's interesting because in talking to those males during their internships or graduate assistantships and afterwards, what I find is that they're really looking for more financially lucrative positions. They end up getting out of health education and finding health-related positions or other things that are paying a lot better. So, you know, as a field, we need to increase our salary.

Based on the feedback from participants, for men, salary could have a negative impact on persistence.

## What Do You Actually Do?

Participants indicated that health promotion in higher education is not a well-known profession. I asked questions about how participants explained their work to others. Four said that while they try to explain, most people do not understand the work. Participants said the following:

I think if I, back to our previous conversation, if I had pursued a career in tropical medicine and I was a disease hunter and I worked in West Africa and you know, was fighting Ebola, I think I would have all attention at every cocktail party and every family gathering. But I think people either don't understand it or [have] just somewhat of a passive disinterest in college health. (David)



I'm never one to really be the center of attention and don't strive to be, so usually my goal is to just find the common ground with anyone I'm interacting with, friends, family, and talk about that. But you know what I mean? My mother, like, to this day, I think, has very little clue of what I do. It's funny, it's very laughable to try and see her give me advice on things that, like, you realize, I do this professionally. (James)

I'll be honest; I find it easier to tell people at dinner parties that I'm an astronaut. Everyone knows what an astronaut is. Being an astronaut is not political. If they're really interested they will be like, "Now what is it that you really do?" And I say, "Well I work with students." If they are more interested, they will continue to pursue it. But I think it really is a challenge to tell people what it is that we do because we do so many things. To say we keep students healthy is so ambiguous and vague. Well, what does that really mean? A lot of people, once they find out I work at a university, they go, "Well are you a teacher?" "Well, I'm not a professor but I do a lot of teaching and education and focusing on learning." "But you're not a teacher?" "I'm a teacher, but I'm not a professor." "Well, what's that mean?" "It means that I'm not typically in front of a class in a classroom giving a lecture. I'm doing other things – communication campaign, one-on-ones, environmental strategies, socioecological approaches." Sometimes it's just easier to tell people you are an astronaut and pray that they move on. (Mark)

Participants felt challenged by this; however, none indicated that it impacted their persistence in the field. Not understanding the field was not directly connected to the participants' gender; however, the majority shared that it could be a challenge.

#### Career Aspirations

Four of the participants spoke about their career aspirations and how they may need to leave health promotion in higher education in order to further advance. Three said,

I don't really see myself leaving the field of higher education, but I could potentially see myself leaving health promotion eventually. What would maybe keep me in health promotion would be the possibility of being able to do those kinds of things but within the context of higher education health promotion. So for example, what's pretty cool about my job here, and makes me feel a little more satisfied, is that I actually have a director position. I'm not subsumed beneath a university health service. I'm not an office within a bigger department. I actually, you know, I'm a department head. That gives me a certain sense of career advancement where I'd never had that. It was never really a possibility at other universities. I could continue to do health promotion but still satisfy that senior-type leadership position. (David)



I still have aspirations, at some point in time, of potentially being a dean of students or a chief student affairs officer, though I need, I will likely need the terminal degree and proper credentials for that to happen along with some more experience outside of the realm of health promotion, and so those moving parts complicate matters a little bit. But I do feel like at the type of institution that I'm accustomed to working at and that I enjoy working at, I could probably elevate one more degree, to, like, an associate dean of students. Which I'd be open to at the right school in the right opportunity, but also it would have to be a position that had some oversight and some responsibility for a health promotion function. So, happy to sort of expand that portfolio and get some more general higher ed admin experience for which I was trained. But I would not be willing to completely let go of the health promotion piece. (Harry)

When I was an undergrad, you know, my career aspiration was to kind of be a director of a prevention unit, a prevention program. I achieved that rather early in my career. I've been happy with that. It's only recently that I've kind of begun questioning what is the next step? What is it that I want to do? Where is it I want to go? You know, I haven't really come up with any answers. When I went to get my Ph.D., I always said it was to open up doors. I think it has the strong possibility of doing that. I just haven't found any future pathways that I've fallen in love with. When I think of being director of a health service you know, I'm not finding that exciting. It, you know, initially it's not appealing to me. So, and I think it's just because I love health promotion and health education so much. I know if I were to become the director of a health service, it's dealing more with the clinic and dealing more with a wide variety of other things and doing more management, and more administration. I mean, yes, it all benefits our students and makes our campus a healthier and safer place, but it's going to be a big paradigm shift for me. (Mark)

Most participants indicated a desire to stay in the field; however, they also found a lack of advancement opportunity beyond their current role. One participant indicated this factor as a reason he considered leaving the field. Other participants stated that this factor could impact men as they may be expected or encouraged to advance. I discuss this in Chapter 5.

# Moving Up Means More Administration

While participants spoke about wanting to advance in their career, they also discussed the challenge of advancing into more administrative roles. David and James stated that in their



director role, they had less day-to-day direct student interaction. Two others described how advancement can be challenging:

I think as my career has evolved, perhaps it's just the result of maturing and becoming more sophisticated, but, you know, I start moving from more direct service to more client management, administration, leadership, guide others. So, unfortunately, you know, I think I spend more time doing administration management, with a dash of statistics, assessment, research. I really don't get to spend much time doing that program development, program implementation front-line types of activities. I miss doing those things; I miss the thrill of doing those things. I don't miss being on campus at 8 o'clock, 9 o'clock, 10 o'clock at night, or on the weekends. So I mean there's definitely give and take. But I miss it, I mean; it's why, it's why I got into this field. It's what I find exciting and invigorating.... Sadly, as a result of moving up, you know, I don't get those opportunities as much. And I can tell that my skill set is diminished because I don't get to practice and exercise that skill as much as I would love to. (Mark)

I don't have a lot of ambition in regards to climbing any ladders. I'm a pure educator which means, like, my place is in front of people presenting in some way. Or writing, I'm trying to do more writing, blogging, that kind of stuff. I need to be sort of outward facing towards the people I'm serving. I don't see myself as being very good at being at meetings or logistical anything...the more money you make, the higher up you get, the less education you do. You end up at meetings, you end up dealing with finances and budgets and whatever, logistics, but you're not doing much education directly...Maybe some people want to do that. For me, like, my place should be educating. That's what I do, and it wouldn't be health for me or for anybody else if I was not doing that. (Matt)

Therefore, while participants indicated an interest in advancement, they also found that taking on a more administrative role took them away from aspects of their career that they enjoyed. I discuss this contradiction in greater detail in the next chapter.

#### Balancing Being a Man and a Lack of Men in the Profession

I asked participants about their definition and perception of masculinity and how masculinity impacted work in health promotion in higher education. While each participant answered in a different way, they included similar characteristics when describing masculinity; strength, leadership, protection, security, provision, and compassion. Additionally, participants



shared that they believed society and people in the lives of men play a role in how men view masculinity. Harry said,

All the messages that maybe were conveyed from a very early age or influenced by the media or all of these other pieces — you know, I think don't actually paint an accurate picture of what it means to be a man and so for me, I think being masculine is less about aggressiveness or not crying or not emoting and more about the opposite actually and standing up for others and operating from an ethic of care and listening and being supportive and compassionate. And I think those are the, those are the factors to my mind [that] go into what masculinity should look like.

Participants discussed gender three ways including their gender in terms of their work, being viewed as a "different kind of man," and their desire to see more men in the profession.

Gender and the Work. I asked participants about how being male and their masculinity impacted their work. Two spoke about how their gender impacted work with other professionals:

I've also replaced female directors. People in my role have always been female, and just, you hear things when you talk to your staff. There seem to be closer relationships with the female staff to the director or supervisor who is a female. So I've got great relationships with my team here for example; however, I think there was a lot more personal interactions between my predecessor who was female and the team that I currently work with, and I'm not sure if it was because she was a woman or because of her personality. Maybe her personality lends itself to more social interactions with the staff. I'm not really sure. But sometimes I wonder if, you know, being a guy is kind of limiting some of those interactions, and what would it be like if all things being equal, if I was female instead of male? Had the exact same track record, same experience, everything, but I was biologically, um, a female; what would that look like? I'm not sure that I have the answer to that. I'd have to imagine it would be different, though, in some regards. (David)

The gender roles are factoring into conflict at work and then my ability to resolve conflicts at work, you know, is again related to gender differences. So whether that's stereotypically about, like, can I be "too rough" when I'm speaking to anybody in particular? You know, as I'm trying to discipline or take corrective action? Is there such a thing as me being too rough, too firm, too aggressive? So, my approach to conflict may be things that are, like ,"We're going to push through this, and we're going to fight through this, and, you know, we're going to, for lack of a better term, man up." That's how I may look at handling conflict. That's probably very true in terms of how I handle different situations outside of work, is this idea that sucking it up. All these different



stereotypical male things that I don't seem or I don't feel able to do or say the same way [when] talking to staff when resolving conflict. (James)

Similarly, Harry and Matt discussed times when their gender impacted working with students:

I found, was sort of a tendency of mine was just sort of, come to the rescue. Feel like I needed to assure, particularly people who were struggling sort of intro-psychically, with something that they, you know, they did or had done to them when alcohol was involved. And for me to almost rush to reassure or to feel like, you know, everything is going to be ok, or I'm sure it wasn't that bad...So, you know, it may not be harmful, but it may not contribute to having the best impact or outcome on the program, and so I've often sort of wondered is it just something about my style, my background, my training that contributed to that and/or is it more of a male instinct where sometimes you might feel the need to protect or to shelter or insulate or make things better when sometimes people just sort of need to experience what they experience? (Harry)

When I worked at the YWCA, we had a young women's empowerment group that I co led with my female counterpart, and she said to me after this one session, "You're standing up, you're the only male in the room, you're older than everyone else, you kind of had a loud voice, and it wasn't a participatory session." And it was humbling, you know? And there have been a number of situations like that, where, even recently, where I've had to be humbled because I was misusing my power in some way or unaware of my power in some way. (Matt)

The gender of both the participants and those they are working with may be important considerations for male professionals. Based on feedback from all the participants, their gender impacted interactions with students and professionals. While challenging, none of the participants indicated that these interactions negatively impacted their persistence.

A "Different Kind of Man." Three of the participants indicated that they are sometimes viewed as being a "different kind of man." James stated,

I think my training, my experiences in many ways make me...what's the word I'm looking for...I don't want to say enlightened, but I'm a different male, you know, when it comes to my perception of male norms and my exposure to the field and some of the many things that we do that are really more egalitarian in nature and really speak to breaking down gender roles... So I just think there's this idea of being more attuned to, again, what are very very traditional expectations, roles, behaviors, etc. But then, also



kind of saying things that are much more progressive and, I think, essentially allow you to...receive messages and perceive behaviors differently than that a "very traditional male."

Similarly, Matt stated that because of his work in emotional health, "I think my reputation precedes me. Like, I don't think I'm viewed as a normal guy. I don't make any sense. People don't know what to do with me." Participants did not say that this perception impacted them negatively or that it influenced their persistence in the field.

More Men. Four participants described the benefits of more male involvement in the field. Matt stated, "The male voice has a particular power to it, which, again, may be rooted in a problematic imbalance, but for now we can avail ourselves of that privilege to use it for good meaning." Other participants had the following thoughts on how more men would benefit the field:

I do think it would be a benefit to having more men in the field because, you know, there are so many, men play such a critical role on a number of different health issues whether it be for their own personal health and the expenses that, you know, that we as a society pick up for, you know, their lack of knowledge. I think having more men to speak more messaging and then, you know to, related to more male audiences would certainly benefit all of us as a whole. (James)

Personally, I would like to see more men. I mean, it is going to seem cliché but, you know, the more diverse you have of a population, the better off that population is going to be. So I mean, you're just going to get different viewpoints, different perspectives, people bring different competencies, different attitudes. Could we continue going and succeeding the way we are now? Oh yeah, definitely. (Mark)

Similarly, three participants spoke about how their gender benefitted their work. Two described specific experiences working with male student groups on campus and how having a male educator is beneficial. James stated,

I would also say that at times when I am doing presentations, I think my maleness is generally why I'm there. We recognize that students, male students, are more likely to



take to me, or it helps to have me there because I'm a man. And I think that's where it becomes an advantage.

Matt stated, "In an ideal world, you'd have a male and a female presenter and maybe a transgender presenter too." Participants indicated that having more men in health promotion positions would benefit the student population.

While being the only man in the department may be beneficial, some participants suggested that it can also be a challenge. David and James described times in which being the only person available to talk to a student could be challenging for the student:

I've worked on, historically, have been, dare I say "women's issues." Issues that perhaps disproportionately impact females. Notably, issues related to eating disorders and body image, working with survivors of sexual assault, relationship violence, and stalking. So students coming to me with those issues tend to be women, and, of course, these issues impact men as well but I would, arguably statistics would show that women are disproportionately impacted. So it does create, I think, a certain level of weirdness. I think women, female students with these issues are probably more comfortable talking to another woman. So I've never had any complaints. I've never really had students who've, that I know of, who have chosen not to come work with me on these issues. But I've also been sensitive to the fact that, if there, if there is a female student and she's impacted by these issues, she may be more comfortable speaking to a female colleague or staff member of mine. So I will make the referral or offer that as well if they so choose, and, you know, students will take me up on that sometimes. (David)

I think the most unique experience is when it's a female student coming in for something related to sexual health or reproductive health. I think that's, you know, typically, over the course of my career, there's always been that moment where it's, like, "Oh!" They may be a little uneasy with talking to me about things. And I think that I understand that, and I guess I'm used to it, so it's not a big deal. For me, it's not anything that is a challenge per se. And I think I immediately go in with a demeanor and professionalism that quickly establishes credibility. Now, I'll never be certain that's always the case, but I've never had anyone say that "I'd rather speak to a female," you know, I've never experienced that. If it did, I'd be ok with it, and it's fine. It's not something that really happens. (James)



While balancing masculinity and a lack of men in the profession did not negatively impact the majority of participants, one indicated that the lack of male energy was a reason he was looking for employment outside of this field.

### **Sexual Identity**

Three of the participants discussed the assumption of sexual identity as a challenge. I did not ask about sexual identity, but all shared their heterosexuality at some point during the interviews. Harry, James, and Mark mentioned a perception that the majority of men in health promotion in higher education were homosexual. Harry described an experience:

I think a lot of people questioned my sexuality, and why that would matter, I don't know. But I think people were curious and perhaps would perceive that if I am a male in health promotion, I must be gay. And again, why that would matter or what impact that would have on my job, I don't know. But I think that was something that I experienced sort of directly and indirectly... But I do remember fielding, whether it was in college or graduate school, you know, perhaps even early in the career, just fielding a lot of personal questions around so, you know, "So do you have a partner?" or, "Are you dating anybody?" You know, and not necessarily in the context of what could be construed as an appropriate conversation with someone you're developing more intimate professional relationship with. But sometimes sort of coming up where it just didn't seem like it fit, and people just probing....There's just something that also struck me, I guess, some assumptions that are made about young men going into health promotion.

While some participants indicated that this assumption was a challenge, the same participants said that they work with students and colleagues or are friends with people who identify as homosexual. The participants did not indicate that the perception of homosexual identity impacted their persistence.



#### Gender Not Considered

The purpose of this study was to explore the career entry and persistence of men in health promotion in higher education. Many of my questions related to the gender of the participants and how gender impacted their perceptions and experiences of their career. However, several participants indicated that their gender was not something they thought about regularly. One said, "until I've had these conversations with you, I haven't really given a lot of consideration or put special emphasis on my sex or my gender in regards to my work." The participants talked about a lot of information regarding their gender once I asked the questions; however, at times they seemed to struggle to think of specific challenges or benefits related to their gender. I discuss this further in the following chapter.

### **Summary**

In this chapter, I described the findings of this study and the evidence to support the findings. Through the process of open, axial, and selective coding, I developed categories that help explain why participants entered and remain in the field of health promotion in higher education. I found a number of factors that impacted participants' entry into the field.

Additionally, I found many factors that benefitted and challenged persistence in the field and that not all factors directly related to the participants' gender.

In the following chapter, I discuss my findings as they relate to the literature reviewed in Chapter 2 and how researchers and practitioners in the field of health promotion in higher education and student affairs might utilize the findings.



#### CHAPTER 5

#### **DISCUSSION**

The purpose of this final chapter is to discuss the findings described in Chapter 4 within the context of the literature available on men in nontraditional careers and careers in higher education and student affairs. The purpose of this basic interpretive qualitative study was to explore the perceptions of men working as health promotion professionals in the higher education setting regarding their entry into and persistence in their chosen profession.

# Discussion of Findings

Through this study, I sought to answer the following research questions:

- 1. What influenced participants in their decision to enter health promotion in the setting of higher education?
- 2. What influences the participants to remain in health promotion in the setting of higher education?
  - 2a. What benefits do men perceive working in health promotion in higher education?
  - 2b. What challenges do men perceive working in health promotion in higher education?

Based on the data collected, I found many factors that impacted the career entry and career persistence of participants. In the following sections, I explore the findings of my study based

on the literature available on men entering and persisting in nontraditional careers and professionals entering and persisting in careers in student affairs in higher education.

## Career Entry

After completing the data analysis, I found six major themes of career entry that resonated with participants including (a) an attraction to or interest in teaching, helping others, and health; (b) being one of very few men; (c) it is more than just a public health or student affairs degree; (d) influence from others; (e) experiences growing up; and (f) experience as a student leader (see Table 2). Some positively impacted career entry while others could negatively impacted career entry. Additionally, not all factors related directly to the participants' gender.

# Attraction to or Interest in Teaching, Helping Others, or Health

All participants discussed an interest in teaching, helping others, or health prior to their career entry. While this did not directly relate to participants' gender the literature indicates that men entering nontraditional careers had an interest in the topics of their career prior to entering (Carmichael, 1992; Sargent, 2004; Werhan, 2010). Similar to the participants in this study, those in Werhan's (2010) study were not aware of the career option in FCS until college. However, their interests led them to pursue a FCS degree program. In the current study, participant interests included medical practice, substance-abuse prevention, K-12 education, and wellness education. While no participants indicated knowledge of health promotion in higher education initially, all had an interest that eventually led them into the career.



# One of Very Few Men

A factor directly related to participants' gender was experiencing a female-dominated environment in two distinct areas; in graduate school and in their professional career.

#### A Lot of Women in Graduate School

The two participants who earned public health degrees and the one who earned a degree in education said that they became aware of the female-dominance of their career choice while completing their degree. Research indicates that more women than men enter programs and earn degrees in student affairs and public health (National Center for Education Statistics, 2012b; Renn & Hodges, 2007; Taub & McEwen, 2006). The participants discussed the female-dominated nature of their programs in regard to the student body; however, none spoke about the gender of the faculty. The literature I reviewed did not discuss gender of faculty either. The female-dominated nature of their degree programs did not negatively impact the participants' decision to enter health promotion in higher education. Previous findings indicate that men may not believe that just because a career is female-dominated, it is nontraditional (Williams, 1995). Though participants in this study perceived their career path as female-dominated, they did not state that this was a barrier to entry for them. However, two discussed how a female-dominated degree program could discourage men from entering:

People are drawn towards groups, affinity groups or otherwise, with like-minded people often who not only share a lot of characteristics but often who look like them. And so, unless you're super comfortable in your own skin or willing to take a risk, I think it can be a little intimidating to consider entering a profession where literally you could always be the only person of that identified sex or gender, you know, in that space and in that room. And I think some people are just inherently more comfortable and open to that, and for others, I do wonder if that would be something that would discourage pursuit of that career option. (Harry)



At that point in time, you know, 20-24 years ago, those folks may have had a lot more traditional views. There's a part of me that questions perhaps when faced with that transformation; the issues of social justice, equality, egalitarianism, feminism. It may be those topics are scary to folks, to males because then they have to question their identity. They have to question who they are, they have to question their sense of right and wrong. And those are hard questions to answer. Maybe the process is too hard or the answers are things that they're not comfortable with. And as a result, that effects their selection of a career in health promotion or something else. (Mark)

Gottfredson's (1996) theory of career entry suggests that if a career is stereotyped as a career for women, men may be less likely to enter. Based on the statements of the participants, earning a degree that is stereotyped as for females could impact a man's interest in pursuing that career. However, for the participants in this study, a female-dominated degree program did not impact their desire to enter the degree program or career they chose.

The ACHA's *Guidelines for Hiring Health Promotion Professionals in Higher Education* (2014a) states that preferred degrees for professionals include Public Health, Health Promotion, and Health Education, and more women than men earn degrees in these fields (National Center for Education Statistics, 2012b). Mark noted that female applicants comprised the majority of his applicant pool for previous graduate and professional positions. Therefore, the field itself may be restricting who applies and is considered qualified for these positions.

#### Female-Dominated

Similarly, the participants all perceived health promotion in higher education as a female-dominated career; however, this perception seemed to develop only after entering the field. Participants noted that the female-dominant aspect of the field was especially apparent at conferences and meetings with many health promotion professionals. A lack of knowledge of the career, a finding I discuss later in this chapter, may have impacted the participants'



perception. I found that none of the participants knew about this field until during or after graduate school, so they may not have known the field was female-dominated. However, based on previous research, even if they had known, that demographic may not have impacted their career entry (Williams, 1995).

# It is More Than Just a Public Health or Student Affairs Degree

Participants discussed two challenges to career entry in health promotion in higher education related to their degree programs including people do not know what it is and there is not a standardized training for the career.

# What is College Health?

Similar to the literature on student affairs, participants indicated that they did not learn about health promotion in higher education as a career option until college or after (Blimling, 2002). Three participants said they "stumbled" into it. Mark stated that while completing his Ph.D. degree in public health, the faculty in his program did not know how to effectively explain or categorize this field:

One of my biggest frustrations when I was working on my doctorate is that my faculty never really knew where to put college health promotion. It wasn't community health. They would kind of lump it into school health, but then they would have to retract that and say, "Well, when we're talking school health and that's not really college health." My faculty didn't really know where to fit it, and many of them had not really had much interaction with the college health promotion office. So they weren't very likely to promote it as a career possibility.

This factor did not directly relate to the participants' gender and none of the literature on men in nontraditional degrees discussed lack of knowledge of the field as a barrier to entry.

However, all of the participants discussed the lack of knowledge and understanding of health



promotion in higher education as a potential barrier to entry. I discuss this further in the next section. Another factor to consider is that the participants entered the field 10 or more years ago. I do not know whether knowledge or understanding of the field has increased in the past 10 years. However, the findings from McNeil's (2012) study indicate the knowledge of the field has not increased. Further research with newer professionals may be needed to learn whether knowledge of the field has increased.

## Coursework Did Not Prepare Me For This

Four participants stated that the training they received in their degree programs did not cover everything needed for entering this field. James shared,

I think that one of the, my biggest critique with our field/industry is that it doesn't do a whole lot in regards to actually true job and career preparation. There are a lot of people in our field who, you know, they are driven by their passion to help people and serve. And that's great. But I think it's important to make sure that they have an understanding that this work is grounded in science, and it should be approached in a logical, you know, methodical way... You know, so I feel like it shouldn't be so heavily driven by leaders who stumble upon it or accidently run into it. (James)

I found this consistent with a dissertation focused on determining whether there is a need for a graduate-level preparation program for health promotion in higher education. Though the study did not discuss gender, the findings of McNeil's study indicate a need for "skill-based development and an understanding of the unique elements of higher education as a setting of practice" (2012, p. 66). Along with the need to build the skills necessary for the job, connecting with faculty in an academic department may also benefit students. Faculty and administrators in academic programs can act as mentors and role models to male students (Werhan, 2010). With no degree program focused on this field, there may not be a way to connect students with



faculty or administrators who understand the field. This implication could impact both men and women interested in the field of health promotion in higher education.

Mark described a program specifically for health promotion in higher education as not only a need but as something he saw as a possible step for himself career-wise:

My dream vision would be to have a dual appointment with a college health service and an academic health education/health promotion program or college student affairs program where there would at least be a certificate, if not a master's [degree] in college health promotion. And then I could combine supervision and practical experience in the applied setting but also be a faculty member and mentor in the academic instructional setting. So yes, I think we need those kind of things. But I think if there were some institutions that were to develop some programs, I think within a few years, they would find it very lucrative and, you know, I think they would literally have an overabundance of applications. I think there are a lot of undergrad students, I think there are a lot of peer ed students who are saying, "I want to do this, I want to go into it." And if it were made available, you know, the field of dreams; if you building it, they will come. I think it would be a great opportunity for the field. And you know if no academic institutions are willing to do it, then it would be really nice to see whether it's ACHA, whether it's NASPA, whether it's BACCHUS. It would be nice to see some kind of health and wellness in higher education certificate program. Particularly for those that don't have health education, health promotion backgrounds.

These findings present a potential barrier to career entry. Participants could not enter a degree program specifically for health promotion in higher education because it did not exist. Based on the research on men entering nontraditional careers, I found this concerning. If there were such a program, faculty and academic advisors could act as recruiters, encouraging men to consider the field (Werhan, 2010). The literature also calls for male professionals to connect with men in training to encourage them to enter the field and to act as mentors and educators (Galbraith, 1992; Gillingham, 2006; Rice & Goessling, 2005). Similarly, literature on entry into student affairs indicates that connecting with someone currently working in student affairs may encourage entry into the field (Blimling, 2002; Calhoun & Taub, 2014; Hunter, 1992; Taub &



McEwen, 2006). When this study's participants entered college, there was no specific degree program, which made facilitating these types of connections more difficult.

In order to overcome this barrier, current health promotion practitioners could work with male students who have an interest in health, wellness, or education, to encourage them to consider entering into the field. Practitioners and faculty could develop a degree or certificate program aimed at those interested in working in health promotion in higher education and assist with career entry and knowledge of the field. Faculty could cover content specific to this field, and connections between faculty, practitioners, and those wishing to enter the field could be made more easily. A degree or certificate program could benefit both men and women interested in this field. Additionally, several participants discussed an interest in teaching. They could explore teaching classes on health promotion in higher education in a public health or student affairs degree program.

### Influence From Others

Participants discussed how influencers impacted their career entry. "In order for entry-level men in student affairs to be strong role models, they will need to have support themselves (Calhoun & Taub, 2014, p. 47). Three participants in this study said that people influenced their decision to pursue a degree in medicine, education, and substance-abuse prevention. The three talked about very different influencers: family, teachers in high school, and a mentor. Werhan (2010) discusses family and teachers as a support for men pursuing a female-dominated career. Participants discussed a spouse and teachers in college, respectively. David and Harry talked about parents and grandparents; and David discussed teachers in high school. Mark connected with a mentor who worked in substance-abuse prevention who positively impacted his decision



to enter the field of health promotion in higher education. Though this factor does not directly relate to participants' gender, several studies of men indicate that mentor relationships and connecting with a person working in the field positively impacted men's decision to enter a nontraditional career (Betz & O'Connell, 1992; Calhoun & Taub, 2014; Gillingham, 2006; Koberg et al., 1998; Williams, 1992).

# **Experiences Growing Up**

Three participants indicated that experiences growing up impacted their career interest. Two said that they grew up in households where alcoholism was present, and both spent many years educating on the topic of alcohol and substance-abuse prevention. A third participant moved to an area as a young man where he witnessed teenage pregnancy, and he went on to spend a lot of his early years in the field doing sexual health work. Chusmir (1990) found that external influences such as a man's family and society can impact his decision to enter a nontraditional career. However, Chusmir focused on adult male attitudes toward men's role in a family and men's attitude regarding children, not on how family impacted the men growing up. Therefore, while the literature includes information on family in regard to career entry, it does not discuss it in the same way as participants in this study.

#### Experience as a Student Leader

Three participants spoke about how being a student leader positively influenced their career entry. Literature on career entry in student affairs indicates that involvement as a student leader in a student affairs department was a positive factor for career entry (Blimling, 2002; Hunter, 1992; Taub & McEwen, 2006). Participants in the current study discussed involvement



in peer education and Greek life. Two studies include Greek life as a student leadership opportunity (Blimling, 2002; Hunter, 1992). I did not locate literature on peer education related to eventual career entry into student affairs. However, David discussed peer education as a potential entry point for the field, based on his experiences with other health promotion professionals:

A lot of my health educators over the past few years, they were peer health educators in college. They might have been an intern in the health promotion office, or they might have been an undergraduate student in the health promotion office, and/or they might have had a graduate assistantship in a health promotion office or within their public health or health and human services department or whatever it was structured as. So I would encourage someone to get involved in any or all of those ways to see if it's a good fit. To see if it's something they enjoy doing. I would encourage people to get involved doing college health in whatever ways they can. Again, through peer health education, unpaid or paid internships, or paid/unpaid undergrad or grad positions.

Health promotion practitioners may want to encourage those involved as student leaders, including male peer health educators, to consider a career in health promotion in higher education. However this could prove difficult because participants mentioned and the literature suggests that peer education is female-dominated (Beshers, 2008).

The literature reviewed focuses on undergraduate student leadership; however, David described how a graduate student working in his office decided to enter the field:

A wonderful young lady, she was in the college student personnel administration program. She didn't know what career track she was going to have in student affairs, but in her mind it was probably going to be judicial affairs or residence life or, you know, something more traditional. One of the great things about the program, the CSPA program, was that they require students to do two different 1-year graduate assistantships in two different departments. Two very different departments. So, you're exposed to, very intimately exposed to other departments in student affairs. That's what happened with this young lady. She got placed in our department and she just fell in love with health promotion. She was like, "Wow, I knew you guys did this work but I didn't realize how cool it was." So I had some individual meetings with her and just kind of, you know, answered her questions and kind of told my story, kind of outlined potential options of things she might want to consider. I mean, that's what she's going to do. She's graduating this May, and she's probably going to either enroll in yet another



master's program for public health or otherwise just try and move into a college health, health promotion professional field.

Similarly, Mark stated that his involvement as a graduate assistant led to his decision to enter the field. Participation as a leader may not be relegated to undergraduates. Researchers may want to focus on graduate student involvement as a possible factor of career entry into health promotion in higher education. Practitioners should encourage those with an interest in health or student affairs to consider a graduate assistantship or internship with their department. This factor did not directly relate to participants' gender; however, the majority of the participants indicated that student leadership may be an important factor for career entry.

While not all related to their gender, the participants discussed factors that impacted their decision to enter health promotion in higher education. Additionally, they talked about potential barriers that could negatively impact entry. Health promotion practitioners may want to reach out to undergraduate and graduate students in leadership positions and those with an interest in health, wellness, and education. Practitioners could also work with faculty to educate them on the career and to assist faculty in promoting the field. Practitioners and faculty may want to explore the possibility of establishing a degree or certificate program for health promotion in higher education to assist with the preparation of future practitioners. Finally, practitioners interested in teaching might consider developing a class on the field for a public health or student affairs degree program.



### Persistence in Health Promotion in Higher Education

The analysis of the interviews led to a number of factors that benefitted or challenged participants (see Table 3). I found that some factors related to gender, while others did not directly connect to gender.

#### **Beneficial Factors**

Participants discussed multiple factors that they found beneficial about working in health promotion in higher education. The five major themes included (a) participants finding the work fulfilling and enjoyable, (b) mentorship, (c) professional relationships, (d) support from friends and family, and (e) the privilege that comes from being a man.

#### Fulfilling and Enjoyable

Positions in student affairs attract some new professionals because they find the work fulfilling (Taub & McEwen, 2006). Participants discussed positive aspects of their work which included interaction with students, developing initiatives, and having a positive impact on the campus environment. Two participants described specific examples of challenging initiatives that stood out for them as positive experiences:

I think probably my proudest moment was when I worked at my second institution and we were successful not only in implementing a 100% smoke-free policy on the campus, but we did it in conjunction with a smoke-free municipal policy for all restaurants and bars. It's probably one of the most rewarding and stressful situations in my career. It's one of my proudest moments because despite all this work and the adversity, with one fell swoop, this policy created a much healthier campus environment that really impacted a lot of people. So it was very rewarding, it was very exciting, and what made it even sweeter and one of my favorite moments was it was against a lot of adversity. Initially, there was a lot of support for this policy, and I had a lot of people behind me, and I was definitely in charge on behalf of the campus. I was this point person. I spoke at more city council meetings, I was on the radio more, I was on TV more than I'd ever



been in my entire life about anything. This was big news anywhere but essentially for such a small community and small campus, but at a certain point when things kind of got weird and kind of hot under the collar, all of a sudden, I was kind of leading the charge with the proverbial saber and looked back over my shoulder and all of a sudden, there weren't so many people behind me. So overcoming that adversity and creating such a great effect was just amazing. So that was probably my favorite moment when that finally happened. (David)

Being charged with overhauling our alcohol policy for our annual rivalry football tailgate, but in order to do that, we really needed a pretty multidisciplinary group. And one was convened. And I had a key part in that process. But really, everybody from medical and mental health services, to the dean's office and athletics, to the city and town government, our relations people, our PR folks, students, alumni, facilities, you know, some of our maintenance people, campus police, traffic, safety. I mean you name it, everybody was around the table really strategizing and working on this together, and we came up with what I think was a really terrific plan...We had a significant reduction in citations [and] virtually no hospitalizations or admissions to our infirmary. (Harry)

Similarly, participants discussed the importance of connecting to, mentoring, and working with students and how much they enjoyed those experiences. Previous research indicates that men in student affairs find mentorship and role-modeling to be important aspects of their job (Calhoun & Taub, 2014). Participants described their experiences working with students:

To know that you are in a position to influence their lives, to create success in their part in the collegiate environment, you know? I talked about seeing myself as a student affairs, higher education professional and yeah, sure, I surely want to help our students, empower them to make smarter, safer, healthier decisions that's going to impact their health and their quality of life, but it also helps them create success both in the classroom and out of the classroom. I see that as well. And just on a more individual level, it's just really fun to work with students. I was just having a great conversation with one of my students, one of my undergraduate students whose intention is to go to medical school, and checking in with him and, you know, he's just picking my brain about things. Just these informal conversations with students that aren't really part of the job. If you can impact their life in some small way or just support them and validate what they're experiencing. Just reminding them that they will be successful and they can get through this, and just those kinds of interactions are really great. So I think that's what keeps me coming back day-to-day. (David)

I'm starting to meet with students that are going through break-ups and other life challenge, trying to figure out how to use the tools that I have to help them to go through



difficult situations. So this one student that went through a break-up with somebody that was really not good for her, not really good for anybody at that point. Borderline abusive in terms of emotional, not violently abusive but sort of putting her down, the way she looks and what have you. But she was having the urge to contact him or pick up his texts or when he would call. So we just worked out a few theory pieces that I still use today that could help people to think about how they can manage situations like that where intellectually, you know that you shouldn't be with this person but emotionally, you're still attached to them because, well, you were dating for 2 years. That was really a nice moment where I was able to not try to fix the problem per se, but help them to reframe it in a way that they can actually hold it better. So moments like that really, really powerful, really, feel like I'm actually, like I'm helping somebody. (Matt)

Most participants indicated positive feedback regarding their profession. Rosser and Javinar (2003) focused their research on career satisfaction, morale, and intent to leave student affairs. The researchers found that those who felt unhappy in the profession and at their institution had low job satisfaction and morale. I found no literature on men in nontraditional careers that studied enjoyment as a factor of career persistence; therefore, this could be an area of study for future research.

#### Mentorship

Several participants stated that as seasoned professionals in the field, mentoring students and newer professionals was an important aspect of their career. Secondly, participants shared experiences with their mentors.

Students and Newer Professionals. Previous research indicates that entry-level men in student affairs desire to mentor, role-model, and teach other men. Also, male mentors positively influenced entry-level men (Calhoun & Taub, 2014). Participants talked about the importance of mentoring and educating students. David discussed meeting with students to talk about their future goals:



I'm happy to meet with students about all sorts of things. Students who are interested in the Peace Corps; I invite them to come talk to me. I'll make a lot of time to talk because that's a big decision, a big commitment. So I'll block out a couple hours of my time to talk about that. I've also had a lot of students coming to me who were interested in a career in student affairs or specifically, health education in higher education. I will make myself available for 1, 2, or more hours. I'll go have lunch with students, you know, to talk about these things because it's really important and it really takes away from my desk time, which is nonexistent these days. But it's really important for me to interact with students.

Mentoring students and newer professionals may benefit both the mentor and mentee, as it seemed to bring certain participants satisfaction (Rosser & Javinar, 2003). Additionally, one of the factors of career entry I discussed previously was an interest in helping others. Mentoring students and newer professionals may be fulfilling participants' desire to help others. Research suggests that men in nontraditional careers should reach out to and mentor newer men in the field (Dohner et al., 1990; Gillingham, 2006). This study's findings therefore confirm that mentoring benefit men in nontraditional careers. Similarly, Renn and Jessup-Anger (2008) found that new student affairs professionals felt that their "mentors and supervisors played in their lives during their transition into the field" (p. 328).

None of the participants discussed mentor relationships with younger male professionals in the field. However, there may not be many such men for them to mentor. One participant said that women comprised the majority of applicants for positions. Mark talked about multiple mentor relationships; however, all with female professionals. Both male and female professionals in the field could reach out to newer male professionals. Even though some studies indicate that same-gender mentor relationships are preferable (Koberg et al., 1998), others found that being in the same field was more important than gender (Betz & O'Connell, 1992).

<u>Participants' Mentors.</u> Four participants discussed relationships with their mentors. They shared stories of how their mentors assisted them both personally and professionally, and I



found this consistent with previous research (Renn & Hodges, 2007). Harry described how one of his mentors assisted his career advancement:

When I was approached about the opportunity at my current school, I was really conflicted about it and had a lot of heart-to-heart discussions with my supervisor at the time, which I think can be very awkward and not something that a lot of people would necessarily pursue. And I think that I just elected to be very forthcoming and transparent about it, and in retrospect, I suspect that was appreciated. But what I really appreciated about my supervisor at the time is that, you know, he was such an understanding and compassionate person and, you know, really allowed me to sort of talk things through with him. I mean, clearly, he had a stake in the outcome, but he also, he also had a stake, I felt, like a genuine stake in what was best for me and for my family and for my future...And, you know, I feel like at some point, you know, he just sort of gave me a nudge and without saying as much just said, "You know what you need to do right?" And I did, and I sort of took a leap of faith, but I was really grateful for that compassion and understanding.

While participants discussed mentorship, none had had a mentor in health promotion in higher education. Studies of men in nontraditional careers indicate an importance of mentorship within the field (Betz & O'Connell, 1992; Dohner et al., 1990; Galbraith, 1992; Gillingham, 2006). Two participants discussed mentors within higher education student affairs. Harry stated,

The people with whom I've interacted with, that have been most influential in my training and, you know, in my personal and professional development have all been deans. You know, I don't feel like I ever had a health promotion person, specific, like, health promotion person that was, like, looking after me. Certainly lots of elders and wonderful colleagues along the way, but I think most of the people who know me most intimately and have offered, you know, sage advice and counsel over time have been in more dean of students-type role. And I just find that very interesting. I'm not sure what that means, if it's circumstantial, if it's who I'm drawn to, or I'm not sure. I don't know what it means.

Mark felt that when he was a newer professional, mentorship within the field was not available:

I think about my generation of health educators, and I think about those folks who were kind of the leaders before my generation, and they're really a generation ahead...I believe that I felt that lacking in the profession. And I think it is very important for us not to make that mistake.



I was surprised by this finding, as I felt I received mentorship by professionals in the field.

Because participants entered the field 10 or more years ago, I do not know whether there is still a lack of mentorship for newer professionals. Participants never stated that a lack of mentors in the field negatively impacted their persistence. The research on men in nontraditional careers did not discuss mentors for men outside of their nontraditional career in regard to career persistence. Researchers may want to explore if there are any other nontraditional careers where there is a lack of mentorship for men in the field.

Participants also discussed gender in regard to their mentors. The majority had male mentors. Koberg et al.'s (1998) study indicates that people are more likely to have mentors of the same gender. One of my participants discussed having female and homosexual male mentors in the field of wellness education. Though their research focused on career entry rather than career persistence, Betz and O'Connell (1992) found that a mentor in the same nontraditional profession, regardless of gender, might benefit a man in that career.

I found that mentorship played an important role in participants' professional and personal lives. Practitioners may want to make an effort to reach out to students and newer professionals to assist with career entry and persistence. Additionally, men in other nontraditional career may want to connect with newer male professionals.

# Professional Relationships

Participants discussed the importance of connecting with other professionals in the field. Harry stated that his colleagues are "fun. They are some of the smartest, most well-intentioned people that I know. But, you know, I hold them in such high regard. They're creative, they're talented, they have good senses of humor." Other participants said that they enjoyed spending



time with other professionals and that they looked forward to conferences and meetings where they would see other health promotion professionals. The student affairs literature states that involvement in professional organizations and conferences allows new and seasoned professionals to network with other professionals (Chernow et al., 2003). The participants utilized involvement in professional organizations such as the NASPA and the ACHA to connect with other professionals. However, several participants noted that it was most apparent that they were the minority in the profession when attending conferences with other health promotion professionals.

It may be beneficial to create a time and place for male practitioners to connect with each other. Practitioners volunteering with regional and national organizations could facilitate mentoring and conversations for male professionals. Research about men in nontraditional careers indicates they benefit from connecting with other male colleagues (Gillingham, 2006; Koberg et al., 1998). Therefore, connecting with other men in the field may aid in career persistence for men in health promotion in higher education and men in other nontraditional careers.

### Support From Friends and Family

The participants all mentioned that they received support from friends, family, colleagues in the field, and/or mentors. David said,

My spouse has a good understanding of what I do. Understands the importance of it. She's an accountant, and she does finance in higher education. So, she understands the university tempo. It's not necessarily 8-5 banking hour's kind of business.

Similar to David's experience, the participants in Werhan's (2010) study stated that support from a spouse is important.



These findings contrast with some of the literature, as men in nontraditional careers were sometimes seen as lacking ambition or as taking a "step down" (Carmichael, 1992; Sargent, 2004; Williams, 1992). Simpson (2005) discusses how some men choose not to talk about their work in a nontraditional career to friends and family because they would not be supported. None of the participants discussed doing this; however, choosing not to remain silent could be attributed to the fact that those outside of the field may not understand health promotion in higher education or that it is female-dominated. If those outside of the field do not know about the demographics, they may not view health promotion as nontraditional for men. I discuss this further in the challenges section. However, based on participant feedback, experiencing support from friends and family may benefit career persistence for men in nontraditional fields.

# The Privilege That Comes From Being a Man

Participants stated that they believed they may benefit in some way because of their gender; specifically in terms of a better salary and connecting with male administrators. I found this to be consistent with the literature in which both men in a nontraditional career and men in student affairs believed they benefitted because of their gender. Male social work students believed they had more interesting placements and were given more challenging jobs because of their gender. They also felt that they had the unique position of acting as mentors to male clients because they could connect with them (Giesler, 2013). Entry-level men in student affairs stated that it may be easier for them to assume leadership roles because they are men (Calhoun & Taub, 2014).

Participants did not talk about male privilege in terms of career advancement, which is inconsistent with some of the literature. Research indicates that being male in a nontraditional



career may lead to easier advancement to upper-level administrative positions (Calhoun & Taub, 2014; Lupton, 2006; Sargent, 2004; Williams, 1992). One possible explanation for my participants' silence could be the lack of advancement opportunities in the health promotion in higher education field above the director role. Additionally, while participants felt privileged, several also indicated that they felt health promotion may be considered "less than" other areas of student affairs. Some spoke about how other administrators sometimes have to be reminded of the role health promotion plays on campus. Therefore, the power and control that some men look for when they advance professionally may not be available in this field (Calhoun & Taub, 2014). I discuss these challenges further in the next section.

While participants said that they experienced male privilege, it was difficult for some of them to recall exact moments or times they benefitted. Instead, they stated that because of their gender, they may make more money and they may connect better with male administrators. The literature on gender in student affairs indicates that the majority of professionals entering a career in student affairs are female (Renn & Hodges, 2007; Taub & McEwen, 2006; Turrentine & Conley, 2001). However, males comprise the majority of those in senior-level positions (Engstrom et al., 2006; Hamrick & Carlisle, 1990; Turrentine & Conley, 2001). The participants may therefore be correct about their assumption that men in this field are privileged.

Practitioners in health promotion in higher education and in other nontraditional careers may want to develop ways to talk to male students and practitioners about privilege (Loschiavo et al., 2007).

There may be other factors that counteract male privilege. "Men have multiple aspects of identity...that cause them to understand power and privilege in a variety of ways" (Loschiavo et al., 2007, p. 198). James said that his race may mitigate the benefits of being male:



I would consider my current place of employment, it's very, in its not-so-great moments, is still a very patriarchal organization. So do people listen to me a little bit more than some of my colleagues? Probably. And is it a function of my gender? Probably. But it doesn't also hurt that I'm the leader of the department, so it's any one of those things. So yeah, in that way, I think it has something to do with all that. I think that if anything is sometimes more of an immediate, you know, before you consider my maleness, I think people tend to consider race into the equation. Being one of very few Black persons in public health, then my maleness comes into it. When you said benefits, I was, like, well, yeah I see benefits, but then sometimes are they negated or equalized by other, by my race.

Research suggests that fewer non-Whites enter student affairs, and few reach the level of senior student affairs officer (Engstrom et al., 2006; Turrentine & Conley, 2001). However, Engstrom et al. (2006) found that "though women and non-Whites continue to be underrepresented as SSAOs, once they do achieve these positions, and once individual and institutional factors are controlled for, they appear to be paid equitably" (p. 258). Future researchers may want to explore race and male privilege as they relate to working in nontraditional careers.

The five factors that participants discussed benefitted them and may have positively impacted their career persistence. The information I presented in this section could be used by professionals working in other nontraditional fields for men to aid in their persistence. I discuss this further in the implications section.

#### **Challenging Factors**

Each of the participants described challenges that they faced, and one participant discussed his intention to leave the field because of some of the challenges. Some of these factors relate directly to gender, while others were more general. The seven major themes were (a) feeling unsupported, (b) salary, (c) others not understanding what they do, (d) career



aspirations, (e) moving up into more administrative roles, (f) balancing being a man and a lack of men in the profession, and (g) sexual identity.

# Feeling Unsupported

I found that not feeling valued or supported at one time or another by their institution, their department, or their supervisor challenged participants. Harry stated that "much of what I do in my role, it's not the sexiest topics, it's not what's at the forefront of people's minds. They always have to be reminded of these issues and how they're [sic] impacting students." These experiences impacted not just participants but their entire department. James described an experience that impacted his department:

I know it was a big deal for my team particularly. It wasn't just a move, it was a...I think it represented a lot of things. Here we are, in what was probably one of the nicer suites of the building. You know, all of my staff had very nice-sized offices with windows, and then, we get moved into a spot that is basically opposite of that. Small offices, no windows, and part of the reasoning or rationale behind the move was that, you know, we were, another department was essentially needing to expand and to have more providers, so we were asked...we were not asked, we were told that we'd be moved. So I just think the whole process of it felt very challenging for my staff in particular. They attached a lot of meaning to that, and it was hard to not take that personally. I've got to, you know, carry the flag and be positive, and be the leader. When, you know, at the same time, I had my own concerns and frustrations and maybe they weren't as difficult for me personally, just because my own demeanor and just how I approach challenges. But I still recognized and embraced all the feelings that my staff was feeling. And then I had to, you know, again, carry the flag but also still defend my team. It was just very, you know, difficult, and it just was not a fun time.

Participants also discussed the fact that feeling unsupported could be especially problematic for men in the field. David described that feeling unsupported could be detrimental to male persistence in health promotion in higher education:

I don't want to stereotype men but, if men are coming from that masculine, want to be in charge, want to be in the leadership kind of mentality, that might be how some men define masculinity, then they might identify with that notion or maybe that stereotype.



And they can't be a "real director" even if they can, you know, even if there is that opportunity for advancement, as the case with my department. You know, if they need something more substantial, something more real to satisfy that, they might not get it within the traditional historical context of how a health promotion department operates. They might have to, or feel the need to move into another area. Or move into another discipline to satisfy that.

Calhoun and Taub (2014) found that some newer male professionals in student affairs desired to advance within the field to roles with more power and control. If men in health promotion in higher education perceive that they are not seen as "real directors," then they may perceive a lack of power and control. This could negatively impact persistence for men in health promotion in higher education and for men in other nontraditional careers. Additionally, Rosser and Javinar (2003) found that feeling unhappy in the profession of student affairs and going unrecognized by the institution can lead to decreased satisfaction and morale. Eventually, these feelings could lead to professionals leaving their positions. None of the participants stated an intention to leave the profession because they felt unsupported.

#### Salary

Participants discussed not feeling valued monetarily. Mark stated that "there's part of me that's, like, with my skill set and the skills that we have as health educators, I could probably go and do a million other things and probably find some things that are more financially lucrative." Two participants wished they had understood the low salary prior to entering the field. Others talked about how professionals may want to consider consulting, educating, or presenting outside of their institution in order to earn extra money. I found no research on the salary of professionals working in health promotion in higher education; therefore, I cannot say how it



compared to others working in health promotion outside of higher education or to professionals working in other areas of higher education.

Salary could be especially problematic for men in the field. A study of entry-level men in student affairs indicates that the desire or need for a higher salary could influence participants to leave the profession (Calhoun & Taub, 2014). Participants stated that low salary may impact male entry and persistence in health promotion in higher education:

I don't know that my, any of my female colleagues feel the same way. I don't know that they are sitting there thinking about, worrying about how much money they are going to make in a year so that they can take care of their families. And, you know, not...feeling a sense that they're the sole providers and don't know that they have this. I don't know what it's like to be a woman first and foremost, obviously, so I don't know exactly how they feel. But I know what I'm missing is that conversation, like, I don't feel like I can have that in a professional setting, you know, where it's, like, me and whoever else, we're on the same page when it comes to those types of thoughts and feelings and ambitions and aspirations. I mean, when it comes to salaries and things like that, it's certainly going to help attract more men if it's perceived as an opportunity to earn a good income...I wouldn't say that, we certainly don't have horrible incomes by any stretch, but certainly, in comparison to other fields...I think we do ourselves a disservice when saying, "Oh, you're just not going to make a lot of money if you come into public health." Well, money motivates a lot of people to do a lot of things, so that's probably not the best PR message to say. (James)

I think, to put it bluntly, I think men, often in the role that has been sold to them and offered to them, look for satisfaction in regards to money and prestige and power and, you know, hierarchy and some of these other things. (Matt)

Calhoun and Taub (2014) found that men "who were projecting family commitments in the future" indicated a greater interest in leaving the field for a higher salary (p. 42). Therefore, men entering health promotion in higher education may consider leaving the field for a higher salary elsewhere. Researchers may want to explore if men in other nontraditional career leave to find employment in more lucrative fields. Practitioners in the field of health promotion in higher education may want to be aware that salary may be a reason why men are choosing not to enter and remain in the field.



# What Do You Actually Do?

A general lack of understanding of the field challenged participants. When I asked participants to describe how they explained their profession to others, three participants laughed. James stated, "People don't know what we do, and that's just a function of public health. I think most people just don't have much of a clue what I do on a day-to-day basis." Matt's response was "Which, by the way, is the joke of our field. You know, 'What is it you do anyway?" Participants stated that even when they tried explaining their career, many still did not understand. Some participants seemed frustrated by this fact.

This factor is not directly connected to participants' gender and the literature on men in nontraditional careers does not discuss this challenge. However, one study indicates that student affairs is not considered as a career option because it is not discussed prior to entry into college (Blimling, 2002). If people do not know about or understand a career, how can practitioners recruit new professionals into the field? For the participants, this did not seem to impact their persistence; however, it did seem to cause some frustration. I do not know if this impacts persistence in other nontraditional careers for men as none of the literature discussed a lack of awareness of the field.

There is some documentation available about this career: the ACHA published the Standards of Practice for Health Promotion in Higher Education (2012b) and the Guidelines for Hiring Health Promotion Professionals in Higher Education (2014a) that outline expectations of health promotion professionals working in higher education. Practitioners could utilize these tools to help explain the work.



### Career Aspirations

The lack of advancement opportunity challenged some participants. Four participants discussed not knowing their next steps professionally. James said,

I believe that I've reached a ceiling. So growth and career growth specifically. I think, and I think Number 2 right along with that is just, you know, this level of needing to feel challenged. Needing to feel that you're, you're professionally growing and maturing in sophistication of work, you know. So I think that kind of, the competitive spirit or whatever you want to call it, I think, is where I don't feel, I don't feel I'm getting fulfilled career-wise that way.

Three other participants planning to continue in the field discussed possibly moving into a faculty role or advancing to a SSAO position. Literature on student affairs indicates that professionals may advance into a SSAO role because of the need for a new challenge (Blimling, 2002). In a study of entry-level men in student affairs, the majority of participants discussed their motivation to advance in the field in order to make a difference, though some wanted to advance in order to gain more power and control (Calhoun & Taub, 2014). I found that none of the participants wanted to advance because of a desire for more power or control. Some of the literature on men in nontraditional career indicates that some men feel like they must advance into more administrative positions because those roles are more gender appropriate (Carmichael, 1992; Sargent, 2004; Simpson, 2005; Williams, 1992). The participants in this study did not discuss this as a reason for wanting to advance.

I found that participants did not discuss male privilege in terms of advancement in the field. This may be because participants did not feel there was a position within health promotion in higher education into which they could advance. The literature on nontraditional careers for men did not discuss a lack of advancement opportunity as a barrier to persistence; therefore, it may be something for researchers to explore in the future. Previous literature indicates that more

men have advanced into SSAO positions than women; however, this was not discussed by participants (Engstrom et al., 2006; Hamrick & Carlisle, 1990; Turrentine & Conley, 2001). I did not include any participants who held SSAO positions. Researchers may want to study male health promotion professionals in SSAO positions or student affairs leadership roles in regard to male privilege.

I found that participants perceived a lack of advancement opportunity; however, they also seemed hesitant to advance because they enjoyed their work. I discuss this contradiction further in the next section. Depending on the institution, there may be opportunities for health promotion professionals to advance. While writing this chapter, I learned that one participant accepted a position as a SSAO and he will still oversee wellness initiatives. Therefore, there may be some opportunities for men to advance while still overseeing health promotion. The ACHA's *Guidelines for Hiring Health Promotion Professionals in Higher Education* (2014a) includes two levels of health promotion professional: staff and director. Practitioners may want to consider adding a SSAO-level position to this document because these positions may exist at some institutions.

#### Moving Up Means More Administration

The majority of participants stated their desire to advance; however, advancement may remove them from some of the work they enjoyed. David, James, and Mark all discussed the fact that as directors, they presented less often than they did while in a staff role. Matt had no intention to advance because he did not want to give up his work as a health educator. The literature on men in nontraditional careers indicates that men sometimes feel pushed or forced into more administrative roles (Carmichael, 1992; Sargent, 2004; Simpson, 2005; Williams,



1992). Studies also indicated that men are expected to move into more administrative roles because they are more legitimate or appropriate for men (Carmichael, 1992; Sargent, 2004; Williams, 1992). I did not find that participants felt pressured to advance or that staff-level positions were considered inappropriate for men. Instead, most of the participants had a desire to advance within higher education but to maintain some of their role as health promotion professionals. Participants suggested that they would want to teach or maintain administrative oversight over the health promotion department if they advanced. Making these types of positions or opportunities available may assist with persistence.

I was surprised when analyzing the findings because the two previous sections contradict one another. While participants indicated a desire to advance, they also wanted to continue at least some of the staff-level duties such as presenting and working directly with students. None of the literature on men working in nontraditional careers or professionals in student affairs discussed this type of contradiction. Some men, like Matt, may choose to remain in a staff-level position while others may advance to director-level and beyond. For some men, working in the staff-level position throughout the course of their career may be fine. However, it may be beneficial for practitioners in ACHA or NASPA to determine if there is a need for a health promotion SSAO position at some institutions of higher education.

#### Balancing Being a Man and a Lack of Men in the Profession

Participants discussed balancing their masculinity and their work as a health promotion professional as a challenge. I asked all participants to define masculinity. Harry shared the following:

I think a lot of that translates to our work in health promotion because often, you know, there are perceptions or messages that in order to be a real man, you need to drink a lot,



you need to be aggressive, you have a right to your sexual conquests, um, you shouldn't cry, you shouldn't express too much emotion at all because that's a sign of weakness. You can get physical if and when you need to be physical. Even sort of images of, you know, what men, I guess, should look like, and so I think a lot of that, you know, has influenced thought processes and gender norms and stereotypes and sort of how people relate to, to one another.

I asked participants and their gender and their role as health promotion professionals.

Participants discussed gender in terms of their work, being viewed as different from other men, and about the need for more men in the profession.

Gender and the Work. The participants discussed several ways that their being male impacted their work with other professionals and with students. David, James, and Mark all described challenges interacting with female staff members:

I know in our office, there are times when I'm like, "Ok, we have a young woman whose apparel and dress may be questionable, may not be as professional." I.E. you know, a little too much cleavage showing, and I'll go to my female colleagues and ask, "Can you be the one having this conversation with them?" 'Cause I just think it's kind of creepy, you know? "Oh, you know, the 42-year-old boss is noticing that I'm showing too much cleavage and is coming in to have this conversation with me about it." I think it is far easier to ask my female staff to have that conversation with that female. (Mark)

Sargent (2004) states that there is sometimes suspicion of a man's intent when working with young children. While not exactly the same issue, in Mark's example, he seemed concerned that others could feel uncomfortable if he were to discuss inappropriate attire with a young female employee.

David and James discussed instances in which their gender impacted interactions with students. They felt that female students may not feel comfortable talking with them about certain "women's" issues. The literature indicates that having men in certain roles benefitted nontraditional professions because male professionals could then connect with male students and clients. However, these studies do not discuss issues of connection between male

professionals and female students and clients (Giesler, 2013; Sargent, 2004; Werhan, 2010).

Researchers on men working I nontraditional careers may want to focus on this in the future.

A "Different Kind of Man." Three participants felt that they were viewed as being different from other men. Mark said,

I'm kind of like the anti male. A lot of it has to do more with my personal life than maybe my professional life. My approach to parenting, my approach to family is really kind of egalitarian, equality than perhaps what a lot of my friends view. You know, those conversations and those viewpoints come into everyday conversations at work...I will make a statement, and a lot of times, the ladies in the room will be like, "Oh wow, you know, my husband would never think that way, or, my husband would never make a comment or my husband would never allow that because they typically have more traditional views, more conservative views." I like to think that being a health educator has helped me stay away from hegemonic thoughts and tendencies. I think being a health educator has made me more aware of when I do fall into hegemonic tendencies and viewpoints.

Calhoun and Taub (2014) found that some men in student affairs felt that they had a responsibility to "counter traditional gender stereotypes and combat the negativity that exists about men, power, and privilege" (p. 45). While my participants noticed this perception about themselves, they did not indicate that it negatively impacted their persistence in the field.

More Men. Participants spoke about the lack of men in the profession. James said that missing male energy was a reason he considered leaving the field. "There are times where I do envision being somewhere else. Some other type of area where I'm able to just be a little bit more free and natural." He described how he connected with other men and got exposed to male energy by spending time with his friends outside of work. Some men in nontraditional careers reported feeling a conflict between their work and maintaining a masculine identity (Simpson, 2005). Another study discusses how men in nontraditional careers developed coping mechanisms to handle the role strain (Cross & Bagilhole, 2002). I do not know why the majority



of men in this study did not experience role strain but men in other nontraditional careers do.

This may be an area for future researchers to explore.

While the majority of participants did not discuss role strain or the need to develop coping mechanisms, some participants talked about a desire to connect more with men in the field. Four of them spoke about the benefits of having more men in the field. David said,

I think it would bring a different perspective to the discipline. I think it's, you know, being an underrepresented group, I think there is perhaps a difference in the male perspective in our discipline, not only in terms of what we bring to the table but just maybe also how we do the work that we do. I think it could add an element of diversity that isn't quite there, and I think that's always a good thing to have a lot of different perspectives, a lot of different ideas, approaching work through different lenses, and so in a general sense, I think it's a great thing...If you had, as it pertains to college health programs, if you had more men helping to create these programs, helping to put on these programs, that might have some sort of impact in helping to reach and engage men. I'm not sure that's accurate or if that would have a significant impact, but I think that might be, I might be on to something there.

All participants shared an interest in the findings of this study because they wanted to know more about other men in the profession. Future researchers and practitioners in the field may want to explore ways in which men in the field could connect with one another.

None of the participants shared that they ever felt discriminated against because of their gender. Mark stated,

And it really kind of falls into stereotypes that, well, "Ok, you're the male so you need to set up the gazebo," or you know, "Come help lift things." Actions that would be attributed to, "You're male; you have physical strength, therefore would you please do this, or do this." But I wouldn't say those are everyday situations. I don't feel like I've really, I mean outside of that, I don't feel like I've been discriminated against as a male in the health promotion profession.

This finding contrasts with the literature that states that men in some nontraditional careers may be discriminated against and are not allowed to work in certain areas or are viewed negatively because of their career choice (Lupton, 2006; Sargent, 2004; Williams, 1992). Another study



found that male librarians were made to feel uncomfortable by the female "in-group" (Hickey, 2006). Calhoun and Taub (2014) state that men in student affairs reported being asked to take on more male roles like doing physical labor and making difficult leadership decisions. The participants did not discuss experiencing discrimination; however, several talked about being asked, because of their gender, to meet with or present to male students. Practitioners in health promotion in higher education and other nontraditional career for men may want to be aware of any gender-based bias or discrimination that could potentially occur because the literature reveals it impacts persistence (Johnson, 2010)

## **Sexual Identity**

Three participants stated that the majority of men in the field may be homosexual. I found no research on the sexual identity of those in this field; therefore, I cannot say if this perception is, in fact, a reality. Two of the participants said that others in the field have made them feel uncomfortable, at times, by making assumptions about their sexual identity:

The men that are in public health, many of them are homosexual and I am not. So I think that it also changes gender dynamics in that regard as well. You know, so it's, I'd be lying if I said that I didn't think, "Do people think I'm gay?" Because, you know, I'm talking about sex, and I'm at this conference, or there are other men who are more likely gay because of, they just look, the way that they are presenting themselves. You know? So does that mean by association that I'm also gay, even though I'm not? So, you know, it's not a big worry by any means, but certainly it's something that I think about, and there are times that where, you know, you kind of get used to...So in one regard, I know I shouldn't be bothered by people asking or wondering or thinking and all that. In certain professional conferences...you know, conference sessions are over, now it's time to be casual and social where that conversation comes up about, you know, why are you doing this? And/or, you know, questions about sexual orientation and things like that. (James)

Several studies suggest that there is an assumption that men in nontraditional careers are homosexual (Carmichael, 1992; Sargent, 2004; Simpson, 2005; Werhan, 2010). While two



participants discussed uncomfortable experiences, they did not state that it impacted their career persistence. Similarly, entry-level men in student affairs indicated that homophobia was a factor in gender role conflict but something that they were comfortable dealing with (Calhoun & Taub, 2014).

David described a different challenge in regard to sexual identity. Rather than be perceived as homosexual, his heterosexuality sometimes made LGBTQ students wary of him:

I did a lot of work at my third institution with the LGBTQ community. My position, along with my assistant director, we co-coordinated the LGBTA education program. So I was very, very, and I still am in fact and always have been, but probably because of the nature of my position was a very active support, advocate, and ally for the LGBTQ community. And one of the things that, I think ultimately I built trust with the students that identified as LGBTQ, but there was always this, for the most part, I wouldn't say always, but more often than not, whenever there's a new student within the program or as new students were getting to know me, I think there was an inherent caution. I don't want to say distrust, but an inherent caution that I was a guy because whether it's a stereotype or there's some truth to it, I think I was perceived as being more homophobic than perhaps a woman might have been. I think that once the students got to know me, that's certainly 100% not me. But I kind of sensed standoffishness with both men and women who identified as, you know, LGBT. And so that was kind of an interesting reflection that I've had about that. And again, I don't think it really impacted my work ultimately, but I did have to do a little extra relationship building perhaps or making a point to let students know that I was definitely cool with them being 100% who they are.

All participants identified as heterosexual; however, several stated that they had either worked with, been friends with, or were accepting of those who identify as homosexual. Jome et al. (2005) found that men employed in nontraditional careers are more accepting of homosexual men than those who worked in more traditional careers.

I did not include gender identity as one of my selection criteria for participants nor did I ask about sexual identity. However, all participants mentioned their sexual identity during the interviews. Giesler (2013) had a similar experience when interviewing male social work students, as they made a point to share their heterosexual orientation. I did reach out to two



individuals whom I knew were homosexual; one was unable to commit the time to the interviews and the other never responded to my e-mail about participating. I do not know the sexual identity of others who did not respond to my recruitment e-mails. For three of the current participants, I did not know their sexual identity when I asked them to participate. Therefore, while there may be a perception of homosexuality in the field, I do not know if this is a reality. Researchers may want to focus on the experiences of homosexual men working in health promotion in higher education in order to gain their perspective, as I did not cover it in this study.

These seven factors challenged participants and may have an impact on career persistence for men working in health promotion in higher education. While not all related to participants' gender, the information gathered could be utilized by men working in other nontraditional careers to aid in their persistence.

#### Gender Not Considered

While not a major theme, I believe it is important to point out that some participants said that they did not consider their gender regularly, or at all, prior to the interviews. I asked questions regarding how gender impacted their work with students or with colleagues and what it was like to be a man in this field. Some made comments like, "I actually don't, I don't know. And it may just be a blind spot that I have, but I'm not sure what the benefits of being a male in health promotion," and "What does it mean to be a man working in health education? Um...you know, I don't think about it."

Based on this feedback, until I began asking these questions, gender was not something some participants considered regularly in terms of their work. I believe this is important to note



because men in the field may not think about their maleness or how their gender impacts their work on a regular basis. This may be different than men in other nontraditional careers. None of the literature I reviewed indicated that men in other nontraditional careers did not think about their work in regard to their gender. Participants in this study discussed the benefits of having more men in the field:

I would love to hear more from other men in the field. I think that, you know, I don't have a lot of interaction, and I don't have these types of conversations with other men... I think that was beneficial to hear because there are times I feel very isolated and feel like, you know, I'm alone in this thinking. And it, and that adds a certain level of frustration, like, you know, it's, I can't, and I think that's where some of the career side of things, it's like, well, I don't have many men who can relate specifically. (James)

It hasn't changed the quality of my friendships, but I know that there are times like, one of the things that I feel is that there are not a lot of people I can go talk to about all the different responsibilities and roles that I have as a professional, as a health educator, as a husband, as a father, those types of things. There are just not a lot of people to connect with to kind of talk about the challenges of holding all those different roles. (Mark)

Therefore, while being male may not be something that the participants consider regularly, these findings suggest that recruiting more men into the profession would benefit the men currently working in the field.

### Implications for Practice

Based on the data collected, I found practice implications for two specific groups: health promotion and student affairs professionals working in higher education and graduate-level preparation programs at institutions of higher education. Additionally, professionals in other nontraditional fields for men may be able to utilize the findings of this study in regard to career entry and persistence. Each of these groups could utilize the findings in a different way.



The findings suggest that more men in health promotion in higher education could be beneficial for both students and practitioners. In order to increase awareness of the profession and encourage men to enter, current health promotion and student affairs professionals could reach out to men who might be interested in this field including current male peer educators and male student leaders. Both participants and previous studies suggest that involvement as a student leader positively impacted entry into professions in student affairs (Blimling, 2002; Hunter, 1992; Taub & McEwen, 2006). Additionally, practitioners could reach out to graduate assistants and interns in student affairs and men in student affairs, public health, psychology, and education graduate programs, as the findings of this study suggest that men in these programs may be interested in the profession if they knew it existed. Practitioners could also encourage graduate students to consider completing internships or assistantships in their offices. The findings of this study and of previously completed studies of men in nontraditional careers and student affairs professionals suggest that connecting with current practitioners in a field can positively impact career entry for men considering a nontraditional career or a career in student affairs (Betz & O'Connell, 1992; Calhoun & Taub, 2014; Gillingham, 2006; Koberg et al., 1998; Williams, 1992). This finding suggests that professionals in other nontraditional fields for men could also male students in gender nontraditional degree programs to encourage career entry.

Health promotion and student affairs practitioners can reach out to faculty in student affairs and public health programs to describe the type of work done in health promotion offices and to share the ACHA's *Guidelines for Hiring Health Promotion Professionals in Higher Education* (2014a) and the *ACHA Standards of Practice for Health Promotion in Higher Education* (2012b). As participants suggested, faculty in public health and student affairs



graduate programs may not know or understand health promotion in higher education. By learning this information, faculty may be more willing and able to discuss this as a career option. Previous research suggests that faculty could positively influence a man's decision to enter a nontraditional career (Werhan, 2010). Finally, several participants indicated that they had an interest in teaching. Male health promotion professionals could consider joining the faculty in public health or student affairs degree programs. This would benefit both career entry and persistence. Participants indicated that they enjoyed teaching but had a desire to continue in their administrative role. This role would allow practitioners an opportunity to teach and could facilitate conversations with students about the field. Additionally, they could then also serve as faculty mentors to students interested in learning more about the field. Current practitioners could also provide mentorship and guidance to newer professionals at their institution and throughout the field. Some participants indicated that this was missing for them, and previous research on student affairs suggests that practitioners reach out to newer professionals (Blimling, 2002; Calhoun & Taub, 2014; Hunter, 1992; Taub & McEwen, 2006). None of the literature on men working in nontraditional fields discussed a lack of awareness of their field; however, they professionals in these fields may want to determine how men in their field learned about it. This could aid in recruiting more men into nontraditional careers.

Male and female practitioners involved in regional and national associations such as the ACHA and the NASPA could set up networking opportunities for male practitioners at conferences or work to create mentorship opportunities to connect more seasoned with newer male professionals. Participants suggested that connecting with other men would be beneficial. Research of men in nontraditional careers suggests that men currently in nontraditional careers should reach out to and mentor newer men in the field (Dohner at al., 1990; Gillingham, 2006).



The findings of this study suggest that mentoring and connecting with others in the field benefitted participants. Men working in other nontraditional fields could also look into setting up mentorship opportunities for newer male colleagues because research suggests this may aid in career persistence. Additionally, those involved with professional organizations could utilize conferences or webinars to educate SSAOs and other higher education administrators on health promotion in higher education. Some participants indicated that their administration did not understand or value their work; therefore, they could utilize these conferences and meetings to educate administrators.

Faculty and staff working in graduate-level programs at institutions of higher education could explore the possibility of adding a health promotion in higher education graduate or certificate program to their department. While not specific to gender, the participants indicated that this in a need that could benefit both male and female health promotion professionals. Previous research indicates that a program for health promotion in higher education is needed and that a certificate program was under development at one university (McNeil, 2012). If a degree or certificate program is not a possibility at an institution, faculty and staff in student affairs and public health programs could work with the health promotion professionals on their campus to learn more about the profession. They could also review both that ACHA's *Guidelines for Hiring Health Promotion Professionals in Higher Education* (2014a) and *Standards of Practice for Health Promotion in Higher Education* (2012b). By doing this, they would be better prepared to discuss health promotion in higher education as a career option with students. They could also facilitate connections between interested students, both male and female, and the health promotion professionals on campus. If there were male health promotion



professionals on campus, faculty could connect interested male students with these professionals for potential mentoring.

### Implications for the Literature

The findings of this study add to the body of literature on men in nontraditional careers and to the literature on professionals in student affairs in higher education. In terms of career entry, the findings are consistent with much of the literature including an interest in the field (Carmichael, 1992; Sargent, 2004; Werhan, 2010), influence from others (Betz & O'Connell, 1992; Blimling, 2002; Calhoun & Taub, 2014; Gillingham, 2006; Koberg et al., 1998; Taub & McEwen, 2006; Williams, 1992), and having a positive experience as a student leader (Blimling, 2002; Hunter, 1992; Taub & McEwen, 2006). Therefore, the findings of this study add to the literature on men in nontraditional careers and professionals working in student affairs.

However, I found other entry factors not discussed in the literature. Participants discussed a lack of awareness of health promotion in higher education as a possible career and a lack of a preparation program as two barriers to entry. None of the literature on men in nontraditional careers discussed these as potential barriers to entry. Several of the nontraditional careers discussed in the literature including nursing and education do have degree programs; therefore that would not be a barrier to entry for those careers. I do not know if other nontraditional careers are more well-known; therefore, I do not know whether a lack of awareness could be a barrier to entry. Additionally, participants described experiences as children that shaped their career choice, which did not come across in any of the literature



reviewed. These factors add to the body of literature and may aid in further exploration by other researchers.

In terms of career persistence, many of this study's findings are consistent with the literature including beneficial factors such as mentorship (Betz & O'Connell, 1992; Calhoun & Taub, 2014; Dohner et al., 1990; Galbraith, 1992; Gillingham, 2006; Koberg et al., 1998; Renn & Hodges, 2007), connecting with male professionals in the field (Calhoun & Taub, 2014; Gillingham, 2006; Koberg, et al., 1998), support from friends and family (Werhan, 2010), and aspects of male privilege (Giesler, 2013; Loschiavo et al., 2007; Lupton, 2006; Sargent, 2004; Williams, 1992). However, some of the findings are not consistent with the previous literature.

I found that participants experienced a lack of mentorship in the field and previous literature indicates mentorship in the field is important for men in nontraditional careers (Betz & O'Connell, 1992; Dohner et al., 1990; Galbraith, 1992; Gillingham, 2006). This presents a potential challenge to career persistence for men in health promotion and potentially for men in other nontraditional careers. None of the literature on nontraditional careers discussed how a lack of mentorship could challenge persistence; therefore this finding provides future researchers a potential issue to study. I also found that participants received support from friends and family while the literature indicates a lack of support for men in nontraditional careers (Carmichael, 1992; Sargent, 2004; Williams, 1992). I do not know why men in health promotion in higher education receive support from family and friends and some men in other nontraditional careers feel unsupported. If men in other nontraditional careers felt supported it may aid in their career persistence. This discrepancy could be studied in future. Finally, little of the reviewed literature on men in nontraditional careers discussed men's enjoyment of their work. The participants in this study all discussed how they found the work fulfilling. Only one



study of student affairs professionals related to this; it found that a high job satisfaction and morale benefit career persistence (Rosser & Javinar, 2003). It may be important to learn from men in other nontraditional careers what aspects of their work they find enjoyable and fulfilling because this information could be used to aid in their persistence.

Similarly, I found some factors including low salary (Calhoun & Taub, 2014), feeling conflict between the male identity and the work (Calhoun & Taub, 2014; Cross & Bagilhole, 2002; Giesler, 2013; Sargent, 2004; Simpson, 2005; Werhan, 2010), lack of advancement opportunity (Blimling, 2002; Calhoun & Taub, 2014), moving into more administrative roles (Carmichael, 1992; Sargent, 2004; Simpson, 2005; Williams, 1992), and sexual identity (Carmichael, 1992; Hickey, 2006; Sargent, 2004; Simpson, 2005; Werhan, 2010) that are well researched that challenged participants. However, one factor discussed in the literature, feeling discriminated against (Lupton, 2006; Sargent, 2004; Williams, 1992), did not seem to challenge participants. I do not know why men in health promotion in higher education do not experience discrimination from coworkers or those outside of the field, but future studies could explore this further. Additionally, several participants indicated that while they wanted to advance in their career, they felt no pressure to advance. This contrasts with some of the previous literature which indicates that some men in nontraditional careers are pushed into more male-appropriate administrative roles (Carmichael, 1992; Sargent, 2004; Simpson, 2005; Williams, 1992). None of the participants considered the health educator role as less appropriate for men and the director role as more appropriate. I do not know why there is this difference between health promotion in higher education and other nontraditional careers; however, future research could study this further.



The findings of this study add to the information available on men in nontraditional careers and student affairs professionals. While some of what I found is consistent with the literature, I found additional factors of career entry and persistence that researchers may want to explore further.

#### Recommendations for Future Research

The findings of the current study indicate that more research needs to be completed on men in nontraditional careers, student affairs professionals, and on health promotion in higher education. In regard to career entry, some participants indicated peer education as a positive factor. While not directly related to gender, future research could focus on peer education as a potential motivator for career entry into student affairs and health promotion in higher education. Though research on peer education could focus on both genders, researchers could pay researchers could explore male peer health educators in higher education, as previous research focused on male peer educators in community settings (Beshers, 2008; Cupples et al., 2010). However, research indicates that men may be underrepresented in peer education groups. Therefore, future research could examine male entry into peer education groups.

Similarly, some participants indicated that experiences as graduate students impacted career entry; therefore, researchers could explore graduate students working in health promotion in higher education and other areas of student affairs on their experiences. While this finding was not gender specific, future research could focus on male graduate students in health promotion and student affairs. Finally, there seems to be little information available regarding career entry as a result of the impact of experiences in childhood. This may be a factor of career



entry that future researchers on both men in nontraditional careers and professionals in student affairs should focus on.

In terms of career persistence, participants indicated that salary could be a factor that negatively impacts persistence, especially for men. While I found some information on salary for professionals working in student affairs I found no information regarding the salary of health promotion professionals. The ACHA Health Promotion Section could add a question regarding salary ranges at the staff and director levels to its next needs assessment. Similarly, participants mentioned there was a lack of understanding about health promotion in higher education outside of the profession. Research on men in nontraditional careers and on professionals in student affairs could explore how a lack of understanding might impact career entry and persistence. In terms of gender, participants indicated that their gender impacted their work with students. I did not locate any literature on how the gender of a professional impacted working with students in any student affairs departments. Therefore, more research on how the gender of the professional impacts working with students may be needed.

As I discussed previously, the participants in this study did not include all demographic possibilities. Future research could focus on the demographic areas this study does not include. Additionally, when I asked participants about challenges, two of them mentioned demographics that impacted their work. Harry discussed age as a challenge:

Just huge struggles for me, early on, feeling like I constantly had to prove myself and battle against almost a sort of, like, reverse ageism. Like, I was discriminated against and not given the sort of credit that I maybe deserved or could work to earn because I was perceived as being so young in the role and just not really taken seriously. And so that was tough, for quite a few years, to feel I was constantly, you know, battling uphill and with that.

Similarly, James said that for him, both race and gender were a challenge:



I guess ultimately what I'm saying is sometimes I feel this tug-of-war between how race-neutral, gender-neutral I kind of need to be and to maintain a certain expectation, a professional expectation, while also feeling a sense of, am I doing my maleness, or am I doing my gender and race a disservice by not being vocal and not speaking up on certain things because of that, you know?

These may not be factors that challenged the persistence of all participants; however, these two examples are possible future areas of exploration.

#### Limitations

A basic interpretive qualitative approach requires the researcher to analyze the data and present the findings based on his/her interpretation and for the researcher to reflect on and interpret the data collected. (Creswell, 2003). This could result in my having interpreted the data incorrectly. However, in order to ensure trustworthiness, I conducted member checks, that required me to share my analysis and interpretation with participants (Creswell, 1998). I received feedback from all participants regarding the data collected. I shared my analysis with participants on two occasions, and I received feedback from two participants regarding my interpretation of the findings. Additionally, I utilized thick descriptions and peer debriefing which both aid in establishing trustworthiness (Creswell, 2003; Lincoln & Guba, 1985).

Therefore, my interpretation of the data should be considered credible and trustworthy (Lincoln & Guba, 1985).

The findings of this qualitative study are not generalizable to the entire population of men who worked in health promotion in higher education; however, they may be transferrable (Merriam, 2009). I focused specifically on men currently working in health promotion in higher education, and all participants had at least 10 years of experience. I did not include retirees or those no longer working in health promotion in higher education. Men may have left to pursue

advancement in higher education to work in health promotion outside of higher education or to take on a new type of career. I hoped that by including men with several years of professional experience, they would have had more experiences to draw from and may have had a better understanding of why they had persisted. Finally, I sampled participants working at 4-year institutions located in the United States. The majority of participants identified as White and all were at least 35 years old and self-identified as heterosexual. While not generalizable to all men working in the field, the findings may be transferable and provide important insight about men working in health promotion in higher education in other countries, men with fewer years of experience in the field, men who left the profession, men working in the field at other types of institutions, and men working in other nontraditional careers.

Another possible limitation is that I am female-identified while my participants were male-identified. Several studies indicate that participants might be more at ease when speaking with an interviewer who is their gender (Cross & Bagilhole, 2002; Lupton, 2006). However, I have a similar professional background as the participants, as I am a health promotion professional in the higher education setting and, I hoped that this commonality would allow the participants to feel more comfortable discussing information with me.

A final limitation to this study is that I worked in the same profession as the participants. I knew and had interacted with three of the participants previously at conferences and through involvement in national health promotion organizations. These previous interactions could have made the participants more comfortable because I understood and knew the profession. However, it could have made the participants less likely to discuss certain pieces of information if they were concerned about confidentiality. I expressed the confidential nature of this research as well as the use of pseudonyms regarding the findings. By taking these steps to ensure



confidentially I believe I put the participants at ease so that they were willing to discuss their career entry and persistence openly.

#### Conclusion

This study explored career entry and persistence of men in health promotion in higher education. I found multiple factors that impacted career entry and persistence for men in this field. Many of the findings are consistent with the literature; however, other findings require future research. The findings can be used by researchers to develop new studies on men in nontraditional careers and men in student affairs and health promotion. Practitioners in health promotion and faculty in graduate preparation programs can utilize the findings to reach out to and work with men interested in or working in the field. Finally, those working in graduate preparation programs can utilize the findings to develop courses or programs focused on health promotion in higher education.



#### REFERENCES

- 1998 Amendments to the Higher Education Act of 1965 § 20 U.S.C. 1001 (1998). Retrieved from http://www2.ed.gov/policy/highered/leg/hea98/sec101.html
- American College Health Association. (2012a). *ACHA health promotion needs assessment 2012*. Baltimore, MD: Author.
- American College Health Association. (2012b). *Standards of practice for health promotion in higher education*. Baltimore, MD: Author.
- American College Health Association. (2014a). Guidelines for hiring health promotion professionals in higher education. Baltimore, MD: Author.
- American College Health Association. (2014b). *National college health assessment II: Reference group executive summary fall 2014.* Baltimore, MD: Author.
- American College Personnel Association. (1996). *The student learning imperative: Implications for student affairs.* Washington, DC: Author.
- American College Personnel Association & National Association of Student Personnel Administrators. (2010). *Professional competency areas for student affairs practitioners*. Washington, DC: Author. Retrieved from http://www.naspa.org/images/uploads/main/Professional\_Competencies.pdf
- Berdahl, R. O., Altbach, P. G., & Gumport, P. J. (1999). The contexts of American higher education. In P. J. Gumport, P. G. Altbach, & R. O. Berdahl (Eds.), *American higher education in the 21<sup>st</sup> century.* (pp. 1-37). Baltimore, MD: Johns Hopkins University Press.
- Beshers, S. C. (2008). Where are the guys in peer education? A survey of peer education programs related to adolescent sexual health in New York State. *American Journal of Sexuality Education*, *3*(3), 277-294. doi:10.1080/15546120802148010
- Betz, M., & O'Connell, L. (1992). The role of inside and same-sex influencers in the choice of nontraditional occupations. *Sociological Inquiry*, 62(1), 98-106. doi: 10.1111/j.1475-682X.1992.tb00185.x



- Biddix, J. P. (2011). "Stepping stones": Career paths to the SSAO for men and women at four-year institutions. *Journal of Student Affairs Research and Practice*, 48(4), 443-461.
- Blimling, G. S. (2002). Reflections on career development among student affairs leaders. *New Directions for Student Services*, 98, 27-36.
- Bogdan, R. C., & Biklen, S. K. (2007). *Qualitative research for education: An introduction to theories and methodologies* (5<sup>th</sup> ed.). Boston, MA: Allyn and Bacon.
- Calhoun, D. W., & Taub, D. J. (2014). Exploring the gender identity roles of entry-level men in student affairs. *College Student Affairs Journal*, 32(1), 35-51.
- Carmichael, J. V. (1992). The male librarian and the feminine image: A survey of stereotype, status, and gender perceptions. *Library & Information Science Research*, 14, 411-446.
- Chernow, E. K., Cooper, D. L., & Winston, R. B. (2003). Professional association involvement of student affairs professionals. *NASPA Journal*, 40(2), 43-58.
- Chusmir, L. H. (1990). Men who make nontraditional career choices. *Journal of Counseling & Development*, 69, 11-16. doi: 10.1002/j.1556-6676.1990.tb01446.x
- Council for the Advancement of Standards in Higher Education. (2012). *CAS professional standards for higher education* (8th ed.). Washington, DC: Author.
- Corbin, J., & Strauss, A. (2007). *Basics of qualitative research: Techniques and procedures for developing grounded theory* (3<sup>rd</sup> ed.). Thousand Oaks, CA: Sage.
- Creswell, J. W. (1998). *Qualitative inquiry and research design: Choosing among five traditions.* Thousand Oaks, CA: Sage.
- Creswell, J. W. (2003). Research design: Qualitative, quantitative, and mixed methods approaches (2nd ed.). Thousand Oaks, CA: Sage.
- Cross, S., & Bagilhole, B. (2002). Girls' jobs for the boys? Men, masculinity and non-traditional occupations. *Gender, Work and Organization*, 9(2), 204-226. doi: 10.1111/1468-0432.00156
- Cupples, J. B., Zukoski, A. P., & Dierwechter, T. (2010). Reaching young men: Lessons learned in the recruitment, training, and utilization of male peer sexual health educators. *Health Promotion Practice*, 11(1), 19-25. doi: 10.1177/1524839909358847
- Daddona, M. F., Cooper, D., & Dunn, M. S. (2006). Career paths and expectations of recent doctoral graduates in student affairs. *NASPA Journal*, 43(2), 203-215. doi: 10.2202/1949-6605.1636



- Davidson, E. S. (2008). Perceived continuing education needs and job relevance of health education competencies among health education and promotion practitioners in college health settings. *Journal of American College Health*, *57*(2), 197-210. doi: 10.3200/JACH.57.2.197-210
- Dodson, T. A., & Borders, L. D. (2006). Men in traditional and nontraditional careers: Gender role attitudes, gender role conflict, and job satisfaction. *The Career Development Quarterly*, *54*(4), 283-296. doi: 10.1002/j.2161-0045.2006.tb00194.x
- Dohner, R. E., Loyd, M., & Stenberg, L. (1990). Men: The other professionals in home economics education. *The Journal of Home Economics*, 82(4), 32-36.
- Engstrom, C. M., McIntosh, J. G., Ridzi, F. M., & Kruger, K. (2006). Salary determinants for senior student affairs officers: Revisiting gender and ethnicity in light of institutional characteristics. *NASPA Journal*, *43*(2), 243-263. doi: 10.2202/1949-6605.1645
- Galbraith, M. (1992). Understanding career choices of men in elementary education. *Journal of Educational Research*, 85(4), 246-253. doi: 10.1080/00220671.1992.9941123
- Giesler, M. A. (2013). The problem of privilege: Male social work students' preservice experiences. *Journal of Ethnographic & Qualitative Research*, 7(4), 204-214.
- Gillingham, P. (2006). Male social workers in child and family welfare: New directions for research. *Social Work*, *51*(1), 83-85.
- Glaser, B. (1978). Theoretical sensitivity. Mill Valley, CA: Sociology Press.
- Glaser B., & Strauss, A. L. (1967). *The discovery of grounded theory: Strategies for qualitative research.* Chicago, IL: Aldine.
- Glesne, C. (2006). *Becoming qualitative researchers: An introduction*. Boston, MA: Pearson Education.
- Gottfredson, L. S. (1981). Circumscription and compromise: A developmental theory of occupational aspirations. *Journal of Counseling Psychology*, 28(6), 545-579. doi: 10.1037/0022-0167.28.6.545
- Gottfredson, L. S. (1996). Gottfredson's theory of circumspection and compromise. In D. Brown & L. Brooks (Eds.), *Career choice and development* (3<sup>rd</sup> ed.; pp.179-232). San Francisco, CA: Jossey-Bass.
- Green, L., & Kreuter, M. (1990). Health promotion as a public health strategy for the 1990s. *Annual Review of Public Health*, 11, 313-334.
- Hamrick, F. A., & Carlisle, L. W. (1990). Gender diversity in student affairs: Administrative perceptions and recommendations. *NASPA Journal*, 27(4), 306-311.



- Harris, F., III, & Edwards, K. E. (2010). College men's experiences as men: Findings and implications for two grounded theory studies. *Journal of Student Affairs Research and Practice*, 47(1), 43-62.
- Heppner, M. J., & Heppner, P. P. (2009). On men and work: Taking the road less traveled. *Journal of Career Development*, 36(1), 49-67. doi: 10.1177/0894845309340789
- Hickey, A. (2006). Cataloguing men: Charting the male librarian's experience through the perceptions and positions of men in libraries. *The Journal of Academic Librarianship*, 32(3), 286-295. doi.org/10.1016/j.acalib.2006.02.009
- Hogue, M., DuBois, C. L., & Fox-Cardamone, L. (2010). Gender differences in pay expectations: The roles of job intention and self-view. *Psychology of Women Quarterly*, 34(4), 215-227.
- Holland, J. L. (1962). Some explorations of a theory of vocational choice: One- and two-year longitudinal studies. *Psychological Monographs: General and Applied*, 76(26), 1-49.
- Holland, J. L. (1985). *Making vocational choices: A theory of vocational personalities & workplace environments* (2<sup>nd</sup> ed.). Englewood Cliffs, NJ: Prentice-Hall Inc.
- Hunter, D. E. (1992). How student affairs professionals choose their careers. *NASPA Journal*, 29(3), 181-216. Retrieved from http://ifst680.tripod.com/Hunter.pdf
- Jackson, Z. V., Wright, S. L., & Perrone-McGovern, K. M. (2010). Work-family interface for men in nontraditional careers. *Journal of Employment Counseling*, 47, 157-166. doi: 10.1002/j.2161-1920.2010.tb00100.x
- Johnson, S. P. (2010). Still so few male teachers: Now what? Young Children, 65(3), 18-23.
- Jome, L. M., Surething, N. A., & Taylor, K. K. (2005). Relationally oriented masculinity, gender nontraditional interests, and occupational traditionality of employed men. *Journal of Career Development*, 32, 183-197. doi: 10.1177/0894845305279167
- Kanter, R.M. (1977). Men and women of the corporation. New York, NY: Basic Books.
- Keeling, R. P., & Dungy, G. J. (2004). *Learning reconsidered: A campus-wide focus on the student experience*. Washington, DC: National Association of Student Personnel Administrators, American College Personnel Association.
- Koberg, C. S., Boss, R.W., & Goodman, E. (1998). Factors and outcomes associated with mentoring among health-care professionals. *Journal of Vocational Behavior*, *53*, 58-72.
- Lease, S. H. (2003). Testing a model of men's nontraditional occupational choices. *The Career Development Quarterly*, *51*, 244-258. doi: 10.1002/j.2161-0045.2003.tb00605.x

الغ الاستشارات كالاستشارات

- Lemkau, J. P. (1984). Men in female dominated professions: Distinguishing personality and background features. *Journal of Vocational Behavior*, 24, 110-122.
- Lincoln, Y. S., & Guba, E. G. (1985). *Naturalistic inquiry*. Newbury Park, CA: Sage.
- Loschiavo, C., Miller, D. S., & Davis, J. (2007). Engaging men in difficult dialogues about privilege. *College Student Affairs Journal*, 26(2), 193-200.
- Lupton, B. (2006). Explaining men's entry into female-concentrated occupations: Issues of masculinity and social class. *Gender, Work, and Organization*, 13(2), 103-128. doi: 10.1111/j.1468-0432.2006.00299.x
- McNeil, M. P. (2012). Alignment of academic preparation and job skills: A health promotion case study (Unpublished doctoral dissertation). Nova Southeastern University, Fort Lauderdale, FL.
- Merriam, S. B. (2002). Introduction to qualitative research. In S. B. Merriam (Ed.), *Qualitative research in practice: Examples for discussion and analysis* (pp. 1-33). San Francisco, CA: Jossey-Bass.
- Merriam, S. B. (2009). *Qualitative research: A guide to design and implementation*. San Francisco, CA: Jossey-Bass.
- Merriam, S. B., & Brockett, R. G. (2007). *The profession and practice of adult education. An introduction*. San Francisco, CA, Jossey-Bass.
- Merriam, S. B., & Simpson, E. L. (2000). *A guide to research for educators and trainers of adults* (2<sup>nd</sup> ed.). Malabar, FL: Krieger Publishing.
- National Association of Student Personnel Administrators. (2011). *Wellness and health promotion benchmark survey 2011*. Washington, DC: Author.
- National Center for Education Statistics. (2012a). *Digest of education statistics: 2011*. Washington, DC: U.S. Department of Education.
- National Center for Education Statistics. (2012b). *The condition of education 2012*. Washington, DC: U.S. Department of Education.
- Patton, M. Q. (1987). *How to use qualitative methods in evaluation*. Newbury Park, CA: Sage Publications.
- Patton, M. Q. (1990). *Qualitative research and evaluation methods* (2<sup>nd</sup> ed.). Newbury Park, CA: Sage Publications.



- Patton, M. Q. (2002). *Qualitative research and evaluation methods* (3<sup>rd</sup> ed.). Thousand Oaks, CA: Sage.
- Renn, K. A., & Hodges, J. (2007). The first year on the job: Experiences of new professionals in student affairs. *Journal of Student Affairs Research and Practice*, 44(2), 604-628.
- Renn, K. A., & Jessup-Anger, E. R. (2008). Preparing new professionals: Lessons for graduate preparation programs from the national study of new professionals in student affairs. *Journal of College Student Development, 49*(4), 319-335.
- Rhodes, F. H. (2001). *The creation of the future: The role of the American university*. Ithaca, NY: Cornell University Press.
- Rice, C. J., & Goessling, D. P. (2005). Recruiting and retaining male special education teachers. *Remedial and Special Education*, 25(6), 347-356. doi: 10.1177/07419325050260060501
- Robertson, F. O. (1950). *Health services in state institutions of higher learning in Mississippi*. Washington, DC: American Council on Education.
- Rosser, V. J., & Javinar, J. M. (2003). Midlevel student affairs leaders' intentions to leave: Examining the quality of their professional and institutional work life. *Journal of College Student Development*, 44(6), 813-830.
- Sargent, P. (2004). Between a rock and a hard place: Men caught in the gender bind of early childhood education. *The Journal of Men's Studies*, *12*(3), 173-192. doi: 10.3149/jms.1203.173
- Sargent, P. (2005). The gendering of men in early childhood education. *Sex Roles*, *52*(3/4), 251-159. doi: 10.1007/s11199-005-1300-x
- Seidman, I. (2006). *Interviewing as qualitative research: A guide for researchers in education and the social sciences*. New York, NY: Teachers College Press.
- Simpson, R. (2005). Men in non-traditional occupations: Career entry, career orientation and experience of role strain. *Gender, Work and Organization*, 12(4), 363-380. doi: 10.1111/j.1468-0432.2005.00278.x
- Sloane, B. C., & Zimmer, C. H. (1992). Health education and health promotion on campus. In K. Patrick (Ed.), *Principles and practices of student health: Volume 3: College health* (pp. 539-557). Oakland, CA: Third Party.
- Taub, D. J., & McEwen, M. K. (2006). Decision to enter the profession of student affairs. *Journal of College Student Development*, 47(2), 206-216. doi: 10.1353/csd.2006.0027



- Tokar, D. M., & Jome, L. M., (1998). Masculinity, vocational interests, and career choice traditionality: Evidence for a fully mediated model. *Journal of Counseling Psychology*, 45(4), 424-435. doi: 10.1037/0022-0167.45.4.424
- Turrentine C. G., & Conley, V. M. (2001). Two measures of the diversity of the labor pool for entry-level student affairs positions. *NASPA Journal*, 39(1), 84-102.
- U.S. Department of Education. (1998). 1998 Amendments to the Higher Education Act of 1965. Retrieved from http://www2.ed.gov/policy/highered/leg/hea98/sec101.html
- Walsh, B. (1988). Traditional female occupations and Holland's theory for employed men and women. *The Journal of Vocational Behavior*, 32(3), 358-365. doi.org/10.1016/0001-8791(88)90026-7
- Werhan, C. R. (2010). Men as FCS teachers: An exploration of a gendered profession. *Journal of Family and Consumer Sciences*, 102(1), 27-32.
- Williams, C. L. (1992). The glass escalator: Hidden advantages for men in the "female" professions. *Social Problems*, *39*(3), 253-267. doi: 10.2307/3096961
- Williams, C. L. (1993). Introduction. In C. L. Williams (Eds.), *Doing "women's work": Men in nontraditional occupations* (pp. 1-10). Newbury Park, CA: Sage.
- Williams, C. L. (1995). *Still a man's world: Men who do "women's work"*. Berkley: University of California Press.
- World Health Organization. (1991). *Sundsvall statement on supportive environments for health*. Sundsvall, Sweden: Author.
- World Health Organization. (2013, February 10). What do we mean by "sex" and "gender"?. Retrieved from http://apps.who.int/gender/whatisgender/en/index.html
- Zimmer, C. G. (2002). Health promotion in higher education. In H. S. Turner & J. L. Hurley (Eds.), *The history and practice of college health* (pp. 311-327). Lexington: University Press of Kentucky.
- Zimmer, C. G., Hill, M. H., & Sonnad, S. R. (2003). A scope-of-practice survey leading to the development of standards of practice for health promotion in higher education. *Journal of American College Health*, *51*(6), 247-254. DOI:10.1080/07448480309596357



**APPENDICES** 



# APPENDIX A

RECRUITMENT E-MAIL



Subject: Invitation to participate in dissertation data collection

NAME:

My name is Alicia Czachowski. I am in the process of completing my Ed.D. in Adult & Higher Education at Northern Illinois University and am currently working on my dissertation on men who work in health promotion in higher education. The title of my study is *Male Health Promotion Professionals: Why They Enter and Remain*.

The purpose of my study is to explore the perceptions of men working as health promotion professionals in the higher education setting regarding their entry into and persistence in their chosen profession. I will be conducting interviews with men currently working in health promotion in higher education at public and/or private 4-year institutions of higher education on why and how they entered their career, as well as what has kept them working in this career. More specifically, I am looking for participants who have been employed in this field for 5 or more years.

I am contacting you to invite you to be a participant in this study. The study will consist of participating in three 60-90-minute phone interviews over the course of 3-4 weeks. Additionally, I intend to audio record each interview. If you have any questions about the study, please contact me using the phone number or e-mail address provided below.

If you are willing to participate in this study, please confirm with me via e-mail. Once I receive the confirmation, I will e-mail you an informed consent form to complete. More information about the study is available on the informed consent form. Please review this information. Once you sign, please return a scanned copy to me at the e-mail address below. Once I receive the signed form, I will work with you to set up a time for the first interview.

The data obtained will be used by me only. Participant's names and the names of any departments/institutions where the participants have been employed will not be included in the final report and any publications resulting from the study. I can share the final report with you.

Thank you, in advance, for helping with this project. If there is someone else you believe I should be speaking to about this project, please let me know.

Thank you,

Alicia K. Czachowski, MPH, CHES Counseling, Adult, & Higher Education Northern Illinois University DeKalb, IL 60115

Phone: 315-430-8987

Email: aczachowski1@niu.edu



# APPENDIX B

INFORMED CONSENT FORM



# Male Health Promotion Professionals: Why They Enter and Remain Counseling, Adult, & Higher Education Northern Illinois University

I agree to participate in the research project titled *Male Health Promotion Professionals: Why They Enter and Remain* being conducted by Alicia Czachowski, a graduate student at Northern Illinois University. I have been informed that the purpose of the study is to explore the perceptions of men working as health promotion professionals in the higher education setting regarding why and how they entered their career and what has kept them working in this career.

I understand that if I agree to participate in this study, I will be asked to do the following:

- Participate in 3 interviews with the researcher via telephone. Each interview will last approximately 60-90 minutes over 3-4 weeks. Interviews will be audio recorded.
  - During the first interview, participants will be asked to answer questions about why they chose to become a health promotion professional, why they chose the setting of higher education, experiences of previous and current employment, and how they entered this career.
  - o The second interview will allow the researcher and the participant to follow-up on what was discussed in the first interview and will go into more detail on the experiences and perceptions of working in a female-dominated profession.
  - o During the third interview the researcher and the participant will review previous findings and clarify the meaning of the data collected in the first two interviews.
- The researcher may also contact participants via phone after the final interview if any additional information is needed.

I am aware that my participation is voluntary and may be withdrawn at any time without penalty or prejudice, and that if I have any additional questions concerning this study, I may contact Alicia Czachowski at 315-430-8987 or by e-mail at aczachowski1@niu.edu or the dissertation chair, Dr. D. Eric Archer, by e-mail at darcher@niu.edu. I understand that if I wish further information regarding my rights as a research subject, I may contact the Office of Research Compliance at Northern Illinois University at (815) 753-8588.

I understand that the intended benefits of this study include helping researchers and practitioners to better understand why men choose to enter and remain in health promotion in higher education. The findings of this study will add to the understanding of men working in nontraditional careers and in health promotion at institutions of higher education. The findings could be used to aid in encouraging more men to enter and remain in this career.

I have been informed that potential risks and/or discomforts I could experience during this study include the possibility of discomfort from sharing both positive and negative personal experiences while working as a health promotion professional and the possibility of feeling discomfort from sharing information about other professionals in higher education, and departments and institutions in higher education where I have been employed. Finally, I may feel discomfort sharing this information with the researcher as she also works in health promotion in higher education. If I feel uncomfortable at any time I can choose not to answer a



question without penalty. I understand that there is no risk of physical injury from participation in this study.

I understand that all information gathered during this study will be kept confidential. The consent form and the transcripts of the interviews will be kept confidential. My name as well as the names of any of my colleagues and departments/institutions where I have been employed will not be included in the interview transcripts and the final report. All records will be kept locked in a filing cabinet and all electronic files will be password-protected.

I understand that my consent to participate in this project does not constitute a waiver of any legal rights or redress I might have as a result of my participation, and I acknowledge that I have received a copy of this consent form.

Name
Signature
Date
I understand that the researcher would like to audiotape the interviews and I agree to allow the researcher to audiotape any interviews in which I participate. I understand that I may decide to disallow the researcher from audiotaping at any time.
Signature Date

Alicia K. Czachowski, MPH, CHES Northern Illinois University DeKalb, IL 60115

Phone: 315-430-8987

Email: aczachowski1@niu.edu



APPENDIX C
INTERVIEW 1 GUIDE



- 1. Describe what you wanted your career to be when you were growing up. How did this change over time?
- 2. How did you become interested in health promotion?
  - a. Experiences?
  - b. A person/people?
  - c. Interaction with students?
  - d. Compliments your personality?
  - e. Your environment/society?
- 3. Describe your education/training.
  - a. Degree program(s)? Majors/minors?
  - b. Internship?
  - c. Certifications?
  - c. Interactions with professionals already in the field?
- 4. Describe your career progression.
  - a. How many years have you worked in health promotion in higher education?
  - b. What health promotion in higher education positions have you held? What have your job title been?
  - c. At what type(s) of schools have you worked?
  - d. Have you worked outside of health promotion in higher education? If so, what type of career?
- 4. Describe your current job/career.
  - a. At what type of school do you work?
  - b. How would you describe your health promotion program?
  - c. How has your work changed over time?
- 5. What influenced you to enter a career in health promotion in higher education?
  - a. An event/experience?
  - b. A person?
  - c. A mentor?
  - d. Compliments your personality?
  - e. Your environment/society?
- 6. What has influenced you to continue being employed in this field?
  - a. Experiences?
  - b. A person/people?
  - c. A mentor?
  - d. Interaction with students?
  - e. Compliments your personality?
  - f. Your environment/society?



- 7. How would you describe what you do to someone interested in becoming a health promotion professional in the setting of higher education? How would you prepare them to become involved?
  - a. What important things should someone know?
  - b. What is expected of a health promotion professional in the setting of higher education?
  - c. What are the other people involved in the profession like?
  - d. What would you have wanted to know before you decided to pursue this career?
- 8. Demographic questions:
  - a. Age in years:
  - b. Ethnicity/Race:



APPENDIX D

**INTERVIEW 2 GUIDE** 

المنارة الاستشارات

- 1. Describe a typical day at work.
  - a. What experiences do you have?
  - b. What positive events occur?
  - c. What negative events occur?
- 2. Describe your favorite moment working as a health promotion professional in higher education. Describe your least favorite moment.
  - a. A moment presenting/educating?
  - b. An interaction with a student or colleague?
  - c. A special event?
  - d. Why was this moment your favorite/least favorite moment?
- 3. How do students respond to you as a health promotion professional working in higher education? Do you perceive any difference in how they respond to male versus the female professionals?
  - a. Positive reactions?
  - b. Negative reactions?
  - c. Any changes throughout your career?
- 4. How have other health promotion professionals responded to you as a health promotion professional working in higher education? Do you perceive any difference in how they respond to male versus the female colleagues?
  - a. Positive reactions?
  - b. Negative reactions?
  - c. Any changes throughout your career?
- 5. How have the people in your life responded to your being a health promotion professional working in higher education?
  - a. Friends?
  - b. Family?
  - c. Colleagues?
  - d. Others who work at your institution?
  - e. Positive reactions?
  - f. Negative reactions?
- 6. What impact do you perceive your gender has on your experiences in your career?
  - a. Perceived benefits?
  - b. Perceived consequences?
  - c. Any discrimination?
  - d. Any impact on career persistence?
- 7. What advice or information could you share with a new male professional working in health promotion in higher education? How might this change if the new professional was female?



APPENDIX E

**INTERVIEW 3 GUIDE** 



- 1. What is your personal definition of a health promotion professional working in higher education?
  - a. What qualities or personality traits should they possess?
  - b. Do they need any special skills or talents?
- 2. What does it mean to you to be a man working in health promotion in higher education?
- 3. I will follow-up and explore the meaning given to some of the specific experiences from the first two interviews.
- 4. Where do you see yourself in 5 years? 10 years?
  - a. If in health promotion, why do you believe you may remain?
  - b. If in another career or role, why do you believe you may leave?

